ORDINARY AND MODULAR REPRESENTATION THEORY OF TRUNCATED CURRENT LIE ALGEBRAS

by

MATTHEW CHAFFE

A thesis submitted to The University of Birmingham for the degree of DOCTOR OF PHILOSOPHY

> School of Mathematics College of Engineering and Physical Sciences The University of Birmingham March 2024

UNIVERSITY^{OF} BIRMINGHAM

University of Birmingham Research Archive

e-theses repository

This unpublished thesis/dissertation is copyright of the author and/or third parties. The intellectual property rights of the author or third parties in respect of this work are as defined by The Copyright Designs and Patents Act 1988 or as modified by any successor legislation.

Any use made of information contained in this thesis/dissertation must be in accordance with that legislation and must be properly acknowledged. Further distribution or reproduction in any format is prohibited without the permission of the copyright holder.

Abstract

In this thesis, we study the representation theory of truncated current Lie algebras associated to Lie algebras of reductive groups. After giving the necessary preliminaries, we begin by considering the representation theory of these Lie algebras in characteristic 0 by defining a generalisation of the Bernstein–Gelfand–Gelfand category \mathcal{O} for reductive Lie algebras and using this to study the problem of computing composition multiplicities of Verma modules. In particular, we give an inductive procedure to compute these multiplicities in terms of the composition multiplicities of Verma modules for reductive Lie algebras, which are famously given by the Kazhdan–Lusztig polynomials. These results have been published in [7] and [8].

We then move on to consider the representation theory of truncated current Lie algebras in prime characteristic. Here, after proving some elementary structural results such as the classification of semisimple and nilpotent elements, we tackle three main problems. The first is on upper and lower bounds for the dimensions of simple modules; we give an upper bound on the dimensions of simple modules for all p-characters and a lower bound for certain p-characters. Then we classify simple modules for certain p-characters. Finally, we finish by computing the composition multiplicities for projective modules for the restricted enveloping algebras of truncated current Lie algebras, and show they can be given in terms of the composition multiplicities of baby Verma modules for the corresponding reductive Lie algebra. These results are the subject of the preprint [9].

ACKNOWLEDGEMENTS

Foremost and first, I would like to thank my supervisors Simon Goodwin and Lewis Topley for their help and support throughout my PhD. Their advice and knowledge has been invaluable for me in my development as a mathematician. In addition, I would also like to thank Chris Parker for his guidance in his role as co-supervisor. I thank the EPSRC and the University of Birmingham for financial support.

I would also like to thank all my friends and colleagues in the School of Mathematics, of whom there are too many to list here, and without whom my time at Birmingham would not have been as enjoyable as it was.

Finally, I would like to thank my parents for all their love and support; without you I would not be where I am today.

CONTENTS

1	Intr	roduction 1					
	1.1	Main	results	2			
2	Pre	eliminaries					
	2.1	Notation and conventions					
	2.2	Reduc	etive groups and their Lie algebras	10			
	2.3	Category $\mathcal O$ for reductive Lie algebras					
		2.3.1	Definition	13			
		2.3.2	The Kazhdan–Lusztig conjecture	15			
		2.3.3	Twisting functors on category \mathcal{O}	16			
	2.4	Restri	cted Lie algebras	18			
		2.4.1	Definition	18			
		2.4.2	The standard hypotheses for reductive groups	20			
		2.4.3	Representation theory and the Kac–Weisfeiler conjectures	21			
	2.5	Trunc	ated current groups and Lie algebras	23			
		2.5.1	Symmetric invariants and the centre of the enveloping algebra	26			
		2.5.2	Indices of truncated current Lie algebras	28			
		2.5.3	Support varieties in the truncated current case	30			
3	Ord	linary	Representation Theory of Truncated Current Lie Algebras	32			

	3.1	Catego	ory \mathcal{O} for truncated current Lie algebras	
		3.1.1	Definition and first properties of \mathcal{O}	
		3.1.2	Highest weight modules	
		3.1.3	Decomposition of category \mathcal{O}	
	3.2	Parab	olic induction	
		3.2.1	Central characters	
		3.2.2	The parabolic induction functor	
	3.3	Twisti	ing Functors	
		3.3.1	Definition of twisting functors	
		3.3.2	Twisting functors between blocks of category \mathcal{O} 60	
	3.4	Comp	osition multiplicities of Verma modules	
		3.4.1	Definition of composition multiplicities	
		3.4.2	Computation of composition multiplicities of Verma modules 85	
4	Mod	dular l	Representation Theory of Truncated Current Lie Algebras 89	
	4.1	Struct	ture theory	
	4.2	Some	Morita equivalences	
	4.3	The K	Cac-Weisfeiler conjectures in the truncated current case 95	
		4.3.1	The first Kac–Weisfeiler conjecture	
		4.3.2	The second Kac–Weisfeiler conjecture	
	4.4	Classi	fication of simple modules	
		4.4.1	Baby Verma modules and their simple quotients	
		4.4.2	Simple modules	
		4.4.3	The general linear algebra	
	4.5	Carta	n invariants for the restricted enveloping algebras	
		4.5.1	Graded $U_0(\mathfrak{g}_m)$ -modules	

4.5.2	Composition multiplicities of restricted baby Verma modules 107			
4.5.3	Cartan invariants for $U_0(\mathfrak{g}_m)$			
List of References 112				

CHAPTER 1

INTRODUCTION

In this thesis, we study the representation theory of a class of Lie algebras called truncated current Lie algebras, sometimes also known as (generalised) Takiff Lie algebras. These Lie algebras are of the form $\mathfrak{g}_m := \mathfrak{g} \otimes \mathbb{k}[t]/(t^{m+1})$ where \mathfrak{g} is the Lie algebra of a reductive group over an algebraically closed field \mathbb{k} . They first appeared in work of Takiff [51] who showed that in the m=1 case they have polynomial rings of symmetric invariants; these results were subsequently generalised to all m by Raïs-Tauvel in [45] and then more recently to the case of truncated multicurrent algebras by Macedo-Savage in [31].

The representation theory of truncated current Lie algebras in characteristic 0 was first studied in detail by Wilson in [56], in which the theory of highest weight modules for \mathfrak{g}_m was developed. Subsequently, in the $\mathfrak{g} = \mathfrak{sl}_2$, m = 1 case several generalisations of the BGG category \mathcal{O} were investigated by Mazorchuk-Söderberg in [34]; one of the main objects of study in this thesis is a generalisation of one of their definitions to all choices of \mathfrak{g} and m. The $\mathfrak{g} = \mathfrak{sl}_2$, m = 1 case was further explored by Zhu [58] and [59] in which several families of simple modules were described.

On the other hand, the representation theory of truncated current Lie algebras in prime characteristic, and indeed of non-reductive Lie algebras in general, is less explored. There are many results in the reductive case that it is widely believed should hold more generally, for example the Kac–Weisfeiler conjectures made in [55] and subsequently generalised in [28]. Truncated current Lie algebras provide an excellent proving ground for such

conjectures since they are sufficiently closely related to reductive Lie algebras that one can often exploit the reductive case to obtain information about the truncated current case.

Truncated current Lie algebras, and in particular the results on symmetric invariants, also have applications to the theory of vertex algebras. More specifically, in the problem of determining the centre of the critical level affine vertex algebra associated to \mathfrak{g} , the semiclassical limit of the centre decomposes as a direct limit of the algebras $S(\mathfrak{g}_m)^{\mathfrak{g}_m}$ discussed above (see [15, §3.4] and [37, §6]). There are also relationships to the theory of W-algebras; for example this result on the centre of the affine vertex algebra together with techniques involving affine W-algebras were used by Arakawa–Premet [2] to solve Vinberg's problem for centralisers. Finally, we also mention work of Kamgarpour [29] which discusses connections between truncated current Lie algebras and the geometric Langlands program.

1.1 Main results

We now discuss the main results of the thesis. Some of the notation used here will not be introduced formally until Chapter 2, although we will keep the use of such notation to a minimum. The characteristic 0 results were first published in a special case in a paper of the author [7] and then subsequently in full generality in joint work of the author and Lewis Topley [8]. The prime characteristic results can be found in the preprint [9], also joint work of the author and Lewis Topley.

In characteristic 0, our main focus is on generalising several results on the representation theory of Lie algebras of reductive groups to the truncated current case. In particular, we give a generalisation of the definition of Verma modules to truncated current Lie algebras, and aim to answer the question of their composition multiplicities. Fix \mathfrak{g} the Lie algebra of a reductive algebraic group, $\mathfrak{h} \subseteq \mathfrak{g}$ some maximal torus, and $\mathfrak{b} \subseteq \mathfrak{g}$ a Borel subalgebra containing \mathfrak{h} . We consider the representation theory of the truncated

current Lie algebra \mathfrak{g}_m defined earlier. We first introduce in §3.1 a generalisation of the BGG category \mathcal{O} to truncated current Lie algebras. We then decompose this category into direct summands denoted $\mathcal{O}^{(\mu)}$, where $\mu = (\mu_1, \dots, \mu_m) \in (\mathfrak{h}^*)^m$. By proving several equivalences between these $\mathcal{O}^{(\mu)}$ we are able to reduce to the problem of computing composition multiplicities of Verma modules to the $\mu_m = 0$ case. The first of these is a parabolic induction style result:

Theorem 1.1.1. Let $\mu = (\mu_1, \dots, \mu_m) \in (\mathfrak{h}^*)^m$ be such that the coadjoint stabiliser $\mathfrak{g}^{\mu_m} = \mathfrak{l}$ is in standard Levi form. Then the categories $\mathcal{O}^{(\mu)}(\mathfrak{l}_m)$ and $\mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ are equivalent.

This is proved in §3.2. The second equivalence is a generalisation of an equivalence via twisting functors for reductive Lie algebras described in §2.3.3, which will be the main focus of §3.3.

Theorem 1.1.2. Let α be a simple root and $\mu \in \mathfrak{h}^*$ not orthogonal to α . The categories $\mathcal{O}^{(\mu)}$ and $\mathcal{O}^{(s_{\alpha}(\mu))}$ are equivalent.

See Theorem 3.3.11 for a precise description of the functors inducing this equivalence of categories.

The equivalences given in Theorems 1.1.1 and 1.1.2 allow us to reduce the problem of computing composition multiplicities of Verma modules to the subcategory $\mathcal{O}^{(0)}$. Since Verma modules for truncated current Lie algebras are in general not finite length modules, we first make a short digression in §3.4.1 to give a satisfactory definition of composition multiplicities for infinite length modules in \mathcal{O} . We then compute the composition multiplicities of Verma modules in $\mathcal{O}^{(0)}$ via a direct calculation, which yields the following result:

Theorem 1.1.3. The composition multiplicities of a given Verma module for \mathfrak{g}_m can be given in terms of the composition multiplicities of certain Verma modules for \mathfrak{l}_{m-1} , where $\mathfrak{l} \subseteq \mathfrak{g}$ is some Levi factor of \mathfrak{g} .

A precise formula can be found in Corollary 3.4.8. In particular, applying this result inductively the composition multiplicities of Verma modules for \mathfrak{g}_m can be given in terms

of those of Verma modules for some reductive Lie algebra \mathfrak{l} , which as we discuss in §2.4.3 are given by the Kazhdan–Lusztig polynomials.

In the prime characteristic chapter of the thesis, we begin in §4.1 by proving some basic results on semisimple and nilpotent elements in truncated current Lie algebras. We then prove various Morita equivalences between certain reduced enveloping algebras in §4.2. These results are vital to proving the results in the remainder of Chapter 4. In §4.3, we investigate the Kac–Weisfeiler conjectures described in §2.4.3 for truncated current Lie algebras. The two main results in this section are as follows:

Theorem 1.1.4. The first Kac-Weisfeiler conjecture holds for \mathfrak{g}_m where \mathfrak{g} is the Lie algebra of a reductive algebraic group under the standard hypotheses.

Proposition 1.1.5. Let \mathfrak{g} be the Lie algebra of a reductive algebraic group under the standard hypotheses. Then for \mathfrak{g}_m , we have:

- (i) The second Kac-Weisfeiler conjecture holds for all homogeneous p-characters.
- (ii) The second Kac-Weisfeiler conjecture holds for all semisimple p-characters.

We also give a family of examples of Lie algebras which are not the Lie algebras of reductive algebraic groups where the second Kac–Weisfeiler conjecture holds for all *p*-characters.

Proposition 1.1.6. The second Kac-Weisfeiler conjecture holds for $(\mathfrak{sl}_2)_m$ in characteristic greater than 2.

In §4.4, we give a partial classification of the simple modules for truncated current Lie algebras. We deal in particular with the case where the p-character χ is homogeneous and its nilpotent part is in standard Levi form; see §4.4.1 for more detail on these assumptions and why they are necessary. Under these assumptions we obtain a complete classification of simple $U_{\chi}(\mathfrak{g}_m)$ -modules.

Finally, in §4.5 we compute the Cartan invariants for the restricted enveloping algebra $U_0(\mathfrak{g}_m)$; that is, the composition multiplicities of the projective $U_0(\mathfrak{g}_m)$ -modules. We obtain the following result (see Theorem 4.5.9 for a precise formula):

Theorem 1.1.7. The Cartan invariants for $U_0(\mathfrak{g}_m)$ are given in terms of the composition multiplicities of certain $U_0(\mathfrak{g})$ -modules, known as (restricted) baby Verma modules.

CHAPTER 2

PRELIMINARIES

2.1 Notation and conventions

In this section, we introduce some basic definitions and notation that we will use throughout the thesis. Throughout, we work over an algebraically closed field k. In §2.3 and Chapter 3 our field will have characteristic 0, while in §2.4 and Chapter 4 our field will have characteristic p > 0; elsewhere we make no assumptions on the characteristic unless explicitly stated.

We assume familiarity with basic notions from the theory of k-algebras and modules such as the definition of Noetherian and Artinian modules, see for example [20]. Unless otherwise specified, all algebras in this thesis are unital but not necessarily commutative. In addition, we assume elementary definitions from category theory such as adjoint functors, left and right exact functors, and full subcategories; again these can be found in for example [20]. We will in particular often use the language of abelian categories and definitions such as projective and injective objects in abelian categories. We also assume basic notions from the theory of algebraic groups such as the definition of reductive and semisimple groups, see e.g. [49], and some very basic algebraic geometry.

We now define some module theoretic notation we will use throughout the thesis. Let M be a finite length module over an algebra A, and let $0 = M_0 \subseteq M_1 \subseteq \cdots \subseteq M_k = M$ be

a composition series for M. Then for a simple A-module L, the composition multiplicity of L in M, denoted [M:L] is the number of quotients M_i/M_{i-1} in the composition series that are isomorphic to L. Note that by the Jordan-Hölder Theorem [M:L] is independent of the choice of composition series.

Let A be an algebra over a field \mathbb{k} , and let M, N be respectively right and left Amodules. Then the *tensor product* of M and N over A, denoted $M \otimes_A N$, is the vector
space spanned by elements of the form $m \otimes n$ for $m \in M$ and $n \in N$, subject to the
relations:

$$(m_1+m_2)\otimes n=m_1\otimes n+m_2\otimes n$$

$$m \otimes (n_1 + n_2) = m \otimes n_1 + m \otimes n_2$$

$$(m \cdot r) \otimes n = m \otimes (r \cdot n).$$

Note that this is just a vector space over \mathbbm{k} with no inherent action of A. However for any \mathbbm{k} -algebra B, a left B-action on M induces a left B-action on $M \otimes_A N$ via $b \cdot (m \otimes n) = (b \cdot m) \otimes n$. Similarly a right B-action on N induces a right B-action on $M \otimes_A N$ via $(m \otimes n) \cdot b = m \otimes (n \cdot b)$. Unadorned tensor products are always considered to be over \mathbbm{k} .

We now define *Lie algebras*, certain examples of which will be the main objects of study in this thesis.

Definition 2.1.1. A Lie algebra \mathfrak{g} over \mathbb{k} is a \mathbb{k} -vector space equipped with a bilinear map

$$[-,-]:\mathfrak{g}\times\mathfrak{g}\to\mathfrak{g}$$

called the Lie bracket on \mathfrak{g} , such that for all $x, y, z \in \mathfrak{g}$ we have:

$$[x, y] = -[y, x]$$

and

$$[x, [y, z]] + [y, [z, x]] + [z, [x, y]] = 0.$$

Example 2.1.2. The main classes of Lie algebras we consider in this thesis are all examples of Lie algebras of linear algebraic groups; these are often referred to as algebraic Lie algebras. Let G be a linear algebraic group over \mathbb{R} . Then $\mathfrak{g} = \operatorname{Lie}(G)$, the Lie algebra of G, can be described in the following way. Let $\operatorname{Der}(\mathbb{R}[G])$ be the set of derivations on the ring of functions of G. A derivation $D \in \operatorname{Der}(\mathbb{R}[G])$ is said to be left-invariant if $D(g \cdot f) = g \cdot D(f)$ for all $g \in G$ and $f \in \mathbb{R}[G]$. Here G acts on $\mathbb{R}[G]$ via left translation, i.e. $(g \cdot f)(y) = f(g^{-1}y)$. If D_1, D_2 are two left-invariant derivations, then the commutator $[D_1, D_2] = D_1 \circ D_2 - D_2 \circ D_1$ is another left-invariant derivation, and the set $\mathfrak{g} = \operatorname{Lie}(G)$ of such derivations carries the structure of a Lie algebra. For example, if $G = \operatorname{GL}_n$ then $\operatorname{Lie}(G) = \mathfrak{gl}_n$, which can be viewed (under a suitable identification) as the Lie algebra consisting of all $n \times n$ matrices with Lie bracket given by [A, B] = AB - BA.

The two main classes of Lie algebras that will appear in this thesis are Lie algebras of reductive groups (see §2.2) and their corresponding truncated current Lie algebras (see §2.5). We will be particularly interested in studying modules over these Lie algebras, which are defined in the following way:

Definition 2.1.3. Let \mathfrak{g} be a Lie algebra over \mathbb{k} . Then a \mathfrak{g} -module is a \mathbb{k} -vector space M equipped with an action of \mathfrak{g} such that for all $x, y \in \mathfrak{g}$ and $m \in M$ we have:

$$x \cdot (y \cdot m) - y \cdot (x \cdot m) = [x, y] \cdot m$$

Just as one can define the group algebra of a group, one can define an associative algebra $U(\mathfrak{g})$ called the *universal enveloping algebra* of \mathfrak{g} . This is an associative algebra such that modules for the Lie algebra \mathfrak{g} are equivalent to modules for the associative algebra $U(\mathfrak{g})$. It is defined in the following way:

Definition 2.1.4. Let $T(\mathfrak{g})$ be the free tensor algebra on \mathfrak{g} . Then

$$U(\mathfrak{g}) := T(\mathfrak{g})/(x \otimes y - y \otimes x - [x, y] : x, y \in \mathfrak{g})$$

Finally, we note that if \mathfrak{g} is a finite-dimensional Lie algebra (as all Lie algebras considered in this thesis will be) then $U(\mathfrak{g})$ is a Noetherian algebra. In particular, for a finite-dimensional Lie algebra \mathfrak{g} , all finitely generated $U(\mathfrak{g})$ -modules are Noetherian.

Two important representations of \mathfrak{g} are the *adjoint* and *coadjoint* representations. These are defined as follows:

- The adjoint module has underlying vector space \mathfrak{g} , with action given by $x \cdot y = [x, y]$ for all $x, y \in \mathfrak{g}$.
- The coadjoint module has underlying vector space \mathfrak{g}^* , the linear dual of \mathfrak{g} , with action given by $(x \cdot \chi)(y) = -\chi([x,y])$ for all $x,y \in \mathfrak{g}$ and $\chi \in \mathfrak{g}^*$.

For most classes of the Lie algebras considered in this thesis these modules are isomorphic, although this need not be the case in general. We define the *centraliser* \mathfrak{g}^x of an element $x \in \mathfrak{g}$ to be the Lie subalgebra $\mathfrak{g}^x := \{y \in \mathfrak{g} : y \cdot x = 0\} \subseteq \mathfrak{g}$. Analogously, the *coadjoint stabiliser* \mathfrak{g}^{χ} of $\chi \in \mathfrak{g}^*$ is the Lie subalgebra $\mathfrak{g}^{\chi} = \{y \in \mathfrak{g} : y \cdot \chi = 0\}$. We then define the *index* of \mathfrak{g} to be $\operatorname{ind}(\mathfrak{g}) := \min_{\chi \in \mathfrak{g}^*} \dim \mathfrak{g}^{\chi}$.

Similarly, if G is an algebraic group and $\mathfrak{g} = \text{Lie}(G)$, then G acts on \mathfrak{g} and \mathfrak{g}^* via respectively the adjoint and coadjoint actions as follows, where we view \mathfrak{g} as the set of left-invariant derivations of $\mathbb{k}[G]$:

- The adjoint module has underlying vector space \mathfrak{g} , with action given by $(g \cdot D)(f) = D(g \cdot f)$ for all $g \in G$, $D \in \mathfrak{g} \subseteq Der(\mathbb{k}[G])$, and $f \in \mathbb{k}[G]$.
- The coadjoint module has underlying vector space \mathfrak{g}^* , the linear dual of \mathfrak{g} , with action given by $(g \cdot \chi)(D) = \chi(g^{-1} \cdot D)$ for all $g \in G$, $y \in \mathfrak{g} \subseteq \operatorname{Der}(\mathbb{k}[G])$, and $\chi \in \mathfrak{g}^*$; here G acts on \mathfrak{g} via the adjoint action.

The adjoint actions of both G and $\mathfrak{g} = \operatorname{Lie}(G)$ on \mathfrak{g} then extend naturally to actions on both $U(\mathfrak{g})$ and $S(\mathfrak{g})$. We will sometimes refer $U(\mathfrak{g})$ equipped with the adjoint action of \mathfrak{g} as giving $U(\mathfrak{g})$ the structure of an 'ad(\mathfrak{g})-module' to distinguish the adjoint action from the action of \mathfrak{g} on $U(\mathfrak{g})$ by left or right multiplication.

2.2 Reductive groups and their Lie algebras

In this thesis, two classes of Lie algebras we will be concerned with are Lie algebras of semisimple groups and Lie algebras of reductive groups; see e.g. [26, Chapter II.1] for a slightly more detailed and general introduction to much of the material of this section. An algebraic group G is said to be *semisimple* if its solvable radical R(G) is trivial and reductive if its unipotent radical $R_u(G)$ is trivial. Since $R_u(G) \subseteq R(G)$ all semisimple groups are automatically reductive.

Example 2.2.1. Some examples of reductive groups are as follows:

- Any simple algebraic group is semisimple; this includes for example $SL_n(\mathbb{k})$, $SO_n(\mathbb{k})$, and $Sp_{2n}(\mathbb{k})$ (see [32, Table 9.2] for a full classification of the simple algebraic groups).
- The group GL_n is reductive, but not semisimple.
- The direct product of copies of \mathbb{G}_m , the multiplicative group of the field \mathbb{k} , is reductive. If H is an algebraic group isomorphic to such a direct product of copies of \mathbb{G}_m we call H a *torus*.

We now fix some notation which we shall use throughout the thesis. Fix a reductive group G, let $\mathfrak{g} = \operatorname{Lie}(\mathfrak{g})$, and choose some maximal (with respect to inclusion) torus H inside G. Let $\mathfrak{h} = \operatorname{Lie}(H) \subseteq \mathfrak{g}$. Then \mathfrak{h} is an abelian subalgebra of \mathfrak{g} whose adjoint action on \mathfrak{g} is semisimple; we call such a subalgebra a toral subalgebra or just a torus of \mathfrak{g} (in positive characteristic we also insist that a torus is closed under the map $(\bullet)^{[p]}$ that we will discuss in §2.4.1). In fact, since H is a maximal torus of G, it follows that \mathfrak{h} is a maximal torus of G. Under the adjoint action of G all maximal tori in G are conjugate, as are all maximal tori of G.

We write $X^*(H)$ for the character group of H, the group of homomorphisms of algebraic groups $H \to \mathbb{G}_m$. The Lie algebra \mathfrak{g} is then graded by the action of H as $\mathfrak{g} = \sum_{\alpha \in X^*(H)} \mathfrak{g}_{\alpha}$, where $\mathfrak{g}_{\alpha} = \{x \in \mathfrak{g} : h \cdot x = \alpha(h)x \text{ for all } h \in \mathfrak{h}\}$; here H acts on \mathfrak{g} via

the adjoint action. We write $\Phi = \{\alpha \in X^*(T) : \mathfrak{g}_{\alpha} \neq 0\} \setminus \{0\}$. This root decomposition of \mathfrak{g} has the properties that $\mathfrak{g}_0 = \mathfrak{h}$ and $\dim(\mathfrak{g}_{\alpha}) = 1$ for all $\alpha \in \Phi$.

Remark 2.2.2. Given an element $\alpha \in X^*(H)$, one can differentiate to obtain an element $d\alpha \in \mathfrak{h}^*$. If $x \in \mathfrak{g}_{\alpha}$, then one can easily show that x also lies in the $d\alpha$ eigenspace for the adjoint action of \mathfrak{h} on \mathfrak{g} .

In characteristic 0, the homomorphism of abelian groups $d: X^*(H) \to \mathfrak{h}^*$ is injective, so in Chapter 3 we will usually abuse notation and write α instead of $d\alpha$. We then view Φ as a subset of \mathfrak{h}^* rather than of $X^*(H)$. On the other hand, in prime characteristic the map d is no longer injective, so in Chapter 4 we will always distinguish between elements of $X^*(H)$ and elements of \mathfrak{h}^* .

Now we make a choice of positive roots $\Phi^+ \subseteq \Phi$ which gives a corresponding choice of simple roots $\Delta \subseteq \Phi$. In everything that follows we assume that we have made some fixed choice of maximal torus \mathfrak{h} , positive roots Φ^+ , and simple roots Δ . This choice of positive roots gives rise to a partial order on \mathfrak{h}^* given by $\lambda \leq \mu$ if $\mu - \lambda \in \mathbb{Z}_{\geq 0}\Phi^+$. We now fix a basis for \mathfrak{g} as follows. For each $\alpha \in \Phi^+$, we choose some $e_\alpha \in \mathfrak{g}_\alpha \setminus \{0\}$ and some $f_\alpha \in \mathfrak{g}_{-\alpha} \setminus \{0\}$; we also sometimes adopt the convention that $e_{-\alpha} = f_\alpha$. We then set $h_\alpha = [e_\alpha, f_\alpha]$ for all $\alpha \in \Phi^+$, and by rescaling e_α and f_α if necessary we may assume that $[h_\alpha, e_\alpha] = 2e_\alpha$ and $[h_\alpha, f_\alpha] = -2f_\alpha$. The set $\{e_\alpha : \alpha \in \Phi^+\} \cup \{h_\alpha : \alpha \in \Delta\} \cup \{f_\alpha : \alpha \in \Phi^+\}$ is then a basis for \mathfrak{g} provided G is semisimple. If G is reductive then we extend this to a basis for \mathfrak{g} by adding additional basis elements to the set $\{h_\alpha : \alpha \in \Delta\}$ to extend it to a basis of \mathfrak{h} . In characteristic 0 these additional elements can always be chosen to lie in $\mathfrak{z}(\mathfrak{g})$, the centre of \mathfrak{g} .

The choice of positive roots yields a triangular decomposition of \mathfrak{g} in the following way: let $\mathfrak{n}^+ = \bigoplus_{\alpha \in \Phi^+} \mathfrak{g}_{\alpha}$ and let $\mathfrak{n}^- = \bigoplus_{\alpha \in \Phi \setminus \Phi^+} \mathfrak{g}_{\alpha}$. Then by the previous paragraph $\mathfrak{g} = \mathfrak{n}^+ \oplus \mathfrak{h} \oplus \mathfrak{n}^-$. We also write $\mathfrak{b} = \mathfrak{n}^+ \oplus \mathfrak{h}$ for the Borel subalgebra of \mathfrak{g} corresponding to this choice of torus and positive roots.

The Weyl group W of G with respect to a choice of torus H is defined to be W :=

N(H)/H where N(H) denotes the normaliser of H in G. Since H is self-centralising, $w \in W$ acts on H by choosing some lift $\overline{w} \in N(H)$ of w and acting by conjugation. This then induces actions on $X^*(H)$, $\mathfrak{h} = \text{Lie}(H)$, and \mathfrak{h}^* in a natural way.

For many purposes, a more useful choice of action of w on \mathfrak{h}^* is actually a shifted version of this action, denoted \bullet , defined as follows. Let $w \in W$ and $\lambda \in \mathfrak{h}^*$. Then $w \bullet \lambda := w(\lambda + \rho) - \rho$, where $\rho = \frac{1}{2} \sum_{\alpha \in \Phi^+} \alpha$. In the characteristic 0 case, there is also a W-invariant inner product on \mathfrak{h}^* which we denote (\cdot, \cdot) ; see for example [23, §0.2] or [22, §9]. For $\lambda \in \mathfrak{h}^*$ and $\alpha \in \Phi$ we then define $\langle \lambda, \alpha \rangle = (\lambda, \alpha^\vee)$, where $\alpha^\vee = 2\alpha/(\alpha, \alpha)$. This bracket $\langle \cdot, \cdot \rangle$ plays an important role in the representation theory of \mathfrak{g} ; see for example [23, Theorem 2.5] and §2.3.3 below.

Example 2.2.3. Let $G = \operatorname{SL}_n$ and assume the characteristic of \mathbb{R} does not divide n (we will discuss why we make this assumption in more detail in §2.4.2). Then $\mathfrak{g} = \mathfrak{sl}_n$, the Lie algebra of traceless $n \times n$ matrices. The natural choice of torus $H \subseteq \operatorname{SL}_n$ is $H = \{\text{determinant 1 diagonal } n \times n \text{ matrices}\}$, and the corresponding maximal torus of \mathfrak{g} is $\mathfrak{h} = \{\text{traceless diagonal } n \times n \text{ matrices}\}$. The Weyl group W is isomorphic to the symmetric group on n elements. With respect to this choice of torus the root system Φ of \mathfrak{g} , when viewed as lying in \mathfrak{h}^* , is given by $\Phi = \{\epsilon_i - \epsilon_j : 1 \leq i, j \leq n, i \neq j\}$, where $\epsilon_i \in \mathfrak{h}^*$ is given by

$$\epsilon_i(\operatorname{diag}(x_1, x_2, \dots, x_n)) = x_i$$

For $\epsilon_i - \epsilon_j \in \Phi$, the corresponding root space $\mathfrak{g}_{\epsilon_i - \epsilon_j}$ is spanned by the elementary matrix E_{ij} . There is a natural choice of positive roots $\Phi^+ = \{\epsilon_i - \epsilon_j : 1 \le i < j \le n\}$ and simple roots $\Delta = \{\epsilon_i - \epsilon_{i+1} : 1 \le i \le n-1\}$. The triangular decomposition of \mathfrak{sl}_n in this case is given by:

$$\mathfrak{n}^+ = \begin{pmatrix} 0 & * & \cdots & * & * \\ 0 & 0 & & * & * \\ \vdots & & \ddots & & \vdots \\ 0 & & 0 & * \\ 0 & & \cdots & 0 & 0 \end{pmatrix}, \qquad \mathfrak{h} = \begin{pmatrix} * & 0 & \cdots & 0 & 0 \\ 0 & * & & & 0 \\ \vdots & & \ddots & & \vdots \\ 0 & & & * & 0 \\ 0 & 0 & \cdots & 0 & * \end{pmatrix}, \qquad \mathfrak{n}^- = \begin{pmatrix} 0 & 0 & \cdots & 0 \\ * & 0 & & & 0 \\ \vdots & & \ddots & & \vdots \\ * & & & 0 & 0 \\ * & * & \cdots & * & 0 \end{pmatrix}$$

2.3 Category \mathcal{O} for reductive Lie algebras

2.3.1 Definition

We now proceed to give a brief overview of the theory of category \mathcal{O} for Lie algebras of reductive groups, following [23]. Let G be a reductive group over \mathbb{C} , let $\mathfrak{g}=\mathrm{Lie}\,G$, and retain all notation from the previous section. The category $U(\mathfrak{g})$ -mod of all $U(\mathfrak{g})$ -modules is difficult to study since it does not display certain nice homological properties. On the other hand, if we only consider finite-dimensional representations of \mathfrak{g} the problem becomes much easier. For example, if G is semisimple, then the finite-dimensional representations of \mathfrak{g} are always semisimple and the simple modules are easy to classify (see for example [23, Theorem 1.6]); the general reductive case is not much more difficult. It is therefore natural to ask if there is some category of $U(\mathfrak{g})$ -modules which includes some of the infinite-dimensional modules but is still well-behaved. Such a category was introduced by Bernstein-Gelfand-Gelfand in the 1970s. This category, known as the BGG category \mathcal{O} , is defined as follows:

Definition 2.3.1. The category \mathcal{O} is the full subcategory of $U(\mathfrak{g})$ -mod whose objects are the modules M satisfying the following conditions:

- $(\mathcal{O}1)$ M is finitely generated.
- $(\mathcal{O}2)$ The Lie subalgebra \mathfrak{h} acts semisimply on M.

(O3) The Lie subalgebra \mathfrak{n}^+ acts locally nilpotently on M.

We now briefly list some basic properties of this category, all of which can be found for example in [23, §1-3].

- The category \mathcal{O} is abelian.
- The category \mathcal{O} is Noetherian, and Artinian, i.e. every object in \mathcal{O} is both Noetherian and Artinian. In particular, all modules in \mathcal{O} have finite length.
- The category \mathcal{O} is closed under taking submodules, quotients, and finite direct sums.
- Any indecomposable module in \mathcal{O} admits a generalised central character χ . Hence there is a direct sum decomposition $\mathcal{O} = \bigoplus_{\chi} \mathcal{O}^{\chi}$, where \mathcal{O}^{χ} is the full subcategory of \mathcal{O} whose objects are the modules with generalised central character χ .
- Each category \mathcal{O}^{χ} contains only finitely many simple modules.
- Each category \mathcal{O}^{χ} contains enough projectives and injectives, i.e. every $M \in \mathcal{O}^{\chi}$ is (isomorphic to) a quotient of a projective object in \mathcal{O}^{χ} and a submodule of an injective object in \mathcal{O}^{χ} .

The simple modules in the BGG category \mathcal{O} can be constructed as follows. First, we define a class of modules called the *Verma modules*. Let $\lambda \in \mathfrak{h}^*$. Then the Verma module of weight λ , denoted $M(\lambda)$, is defined by:

$$M(\lambda) := U(\mathfrak{g}) \otimes_{U(\mathfrak{b})} \mathbb{C}_{\lambda}$$

where \mathbb{C}_{λ} is the one-dimensional $U(\mathfrak{b})$ -module where $h \in \mathfrak{h}$ acts by $\lambda(h)$ and \mathfrak{n}^+ acts by 0. Furthermore, we have (see for example [23, Theorem 1.2(f) & Theorem 1.3]) the following theorem:

Theorem 2.3.2. Each $M(\lambda)$ has a unique maximal submodule and hence a unique simple quotient. Write $L(\lambda)$ for this unique simple quotient. Then the $L(\lambda)$ are pairwise non-isomorphic, and any simple module in \mathcal{O} is isomorphic to some $L(\lambda)$.

2.3.2 The Kazhdan–Lusztig conjecture

We now continue from the previous section to describe an important result in the theory of category \mathcal{O} , the famous Kazhdan–Lusztig conjecture. This was conjectured by Kazhdan–Lusztig in 1979 [30] and proved independently by Beilinson–Bernstein [5] and Brylinski–Kashiwara [6] in 1981.

To motivate this result, we first define the notion of the formal character of a module. This will use the fact that by $(\mathcal{O}2)$, any module $M \in \mathcal{O}$ is (as a vector space) a direct sum $M = \bigoplus_{\lambda \in \mathfrak{h}^*} M^{\lambda}$ where $M^{\lambda} = \{v \in M : h \cdot v = \lambda(h) \text{ for all } h \in \mathfrak{h}\}$, the λ weight space of M. These weight spaces have the following properties, see for example [23, §1.1]:

Lemma 2.3.3. Let $M \in \mathcal{O}$. Then

- (i) Each M^{λ} is finite-dimensional.
- (ii) If M is indecomposable then the set $\{\lambda \in \mathfrak{h}^* : M^{\lambda} \neq 0\}$ is bounded above with respect to the partial order on \mathfrak{h}^* described earlier.

We can now give the following definition:

Definition 2.3.4. The formal character of M is the function

$$\operatorname{ch} M: \mathfrak{h}^* \to \mathbb{Z}_{\geq 0}$$

$$\lambda \mapsto \dim(M^{\lambda})$$

These formal characters are related to composition multiplicities via the following lemma:

Lemma 2.3.5. Let $M \in \mathcal{O}$. Then there exist unique non-negative integers k_{μ} such that $\operatorname{ch} M = \sum_{\mu \in \mathfrak{h}^*} k_{\mu} \operatorname{ch} L(\mu)$. Furthermore, $k_{\mu} = [M : L(\mu)]$ for all $\mu \in \mathfrak{h}^*$.

Proof. To show existence, observe that for $N \subseteq M \in \mathcal{O}$ we have $\operatorname{ch} M = \operatorname{ch} N + \operatorname{ch}(M/N)$. Applying this to a composition series $0 = M_0 \subseteq M_1 \subseteq \cdots \subseteq M_k = M$, we see that

ch $M = \sum_{i=1}^k \operatorname{ch}(M_i/M_{i-1})$ so the character of M is indeed the sum of characters of simple modules. Uniqueness follows from the fact that $L(\mu)^{\lambda} = 0$ unless $\lambda \leq \mu$ and the fact that $\dim(L(\mu)^{\mu}) = 1$.

One motivation for the Kazhdan-Lusztig conjecture was the problem of computing the formal characters of simple modules in category \mathcal{O} . In particular, it is easy to show that $[M(\lambda):L(\mu)]=0$ unless $\lambda\geq\mu$ and that $[M(\lambda):L(\lambda)]=1$, so by inverting a suitable matrix one obtains an expression for $\operatorname{ch} L(\mu)$ in terms of the $[M(\lambda):L(\mu)]$ and the $\operatorname{ch} M(\lambda)$. Since the characters of Verma modules are easy to compute, to find the characters of simple modules it suffices to compute the composition multiplicities $[M(\lambda):L(\mu)]$ of the Verma modules. This motivated the following result, known as the Kazhdan-Lusztig conjecture and proved by Beilinson-Bernstein [5] and Brylinski-Kashiwara [6].

Theorem 2.3.6. The composition multiplicities $[M(\lambda) : L(\mu)]$ are given by the values of certain polynomials, called Kazhdan–Lusztig polynomials, at 1.

For a detailed constriction of the Kazhdan–Lusztig polynomials via the Hecke algebra we refer the reader to [23, §8.2]. We also remark that the precise statement only applies to composition multiplicities in the principal block \mathcal{O}^0 ; however there exist various category equivalences that allow us to reduce to this case. For our purposes, the precise definition of the Kazhdan–Lusztig polynomials is not so important; what is important is that these composition multiplicities $[M(\lambda):L(\mu)]$ are in principle computable and are given by these polynomials.

2.3.3 Twisting functors on category \mathcal{O}

There are several types of equivalences between direct summands of the BGG category \mathcal{O} . The most relevant such family of equivalences for the purposes of this thesis are the twisting functors. These were first defined by Arkipov in [4]; see also work of Andersen–Stroppel [1] and the overview given in [10, §2]. We begin by fixing a simple root α , and

define a $U(\mathfrak{g})$ - $U(\mathfrak{g})$ -bimodule in the following way. Consider the localisation U_{α} of $U(\mathfrak{g})$ with respect to the multiplicative subset generated by f_{α} . This is in particular a $U(\mathfrak{g})$ - $U(\mathfrak{g})$ -bimodule, and contains $U(\mathfrak{g})$ as a sub-bimodule in a natural way. We also define an automorphism ϕ_{α} of \mathfrak{g} by lifting $s_{\alpha} \in W \cong N(T)/T$ to an element of $N(T) \subseteq G$ and conjugating by this element. The bimodule S_{α} we are interested in is then given by $S_{\alpha} := \phi_{\alpha}(U_{\alpha}/U(\mathfrak{g}))$, where for a given $U(\mathfrak{g})$ - $U(\mathfrak{g})$ -bimodule M we let $\phi_{\alpha}(M)$ denote the bimodule obtained by twisting the left action (but not the right action) on M by ϕ_{α} . There is then a right exact functor:

$$T_{\alpha}: \ U(\mathfrak{g})\operatorname{-mod} \longrightarrow U(\mathfrak{g})\operatorname{-mod}$$

$$M \longmapsto S_{\alpha} \otimes_{U(\mathfrak{g})} M$$

In fact one can define functors T_w for each $w \in W$ by declaring that for $w = s_{\alpha_1} s_{\alpha_2} \dots s_{\alpha_k}$ a minimal length expression for w, the functor $T_w := T_{s_{\alpha_1}} \circ T_{s_{\alpha_2}} \circ \cdots \circ T_{s_{\alpha_k}}$. Alternatively, there is a more sophisticated (but equivalent) way to define T_w directly; see for example [1, §2].

Although these functors T_w are defined as functors $U(\mathfrak{g})$ -mod $\to U(\mathfrak{g})$ -mod, they in fact descend to equivalences between certain direct summands of \mathcal{O} . To make this precise, we introduce the notation of a block of category \mathcal{O} . These are subcategories of the categories \mathcal{O}^{χ} defined earlier. Define an equivalence relation \sim on \mathfrak{h}^* by setting $\lambda \sim \mu$ if there exists a non-split extension of $L(\lambda)$ by $L(\mu)$ and extending this to an equivalence relation. The block of \mathcal{O} corresponding to an equivalence class $[\lambda]$ under this relation is the full subcategory of \mathcal{O} containing precisely the modules whose composition factors are all of the form $L(\nu)$ for $\nu \in [\lambda]$.

Definition 2.3.7. Let $\rho = \frac{1}{2} \sum_{\alpha \in \Phi^+} \alpha$ and define

$$\mathfrak{h}_{\mathrm{dom}}^* := \{ \lambda \in \mathfrak{h}^* : \langle \lambda + \rho, \alpha^{\vee} \rangle \notin \mathbb{Z}^{<0} \text{ for all } \alpha \in \Phi^+ \},$$

the set of dominant weights in \mathfrak{h}^* . Then for $\lambda \in \mathfrak{h}^*_{\mathrm{dom}}$, we define \mathcal{O}_{λ} to be the block of \mathcal{O} corresponding to the equivalence class $[\lambda]$; as λ ranges over $\mathfrak{h}^*_{\mathrm{dom}}$ these \mathcal{O}_{λ} form a complete irredundant list of blocks of \mathcal{O} .

The category \mathcal{O} decomposes as a direct sum $\mathcal{O} = \bigoplus_{\lambda \in \mathfrak{h}_{\text{dom}}^*} \mathcal{O}_{\lambda}$; see for example [23, §1.13]. Twisting functors then yields equivalences between these blocks by the following result; a proof can be found in [10, Theorem 2.1].

Theorem 2.3.8. Let $\mu \in \mathfrak{h}_{dom}^*$ and $\alpha \in \Delta$ be such that $\langle \mu, \alpha^{\vee} \rangle \notin \mathbb{Z}$. Then $T_{s_{\alpha}} : \mathcal{O}_{\mu} \to \mathcal{O}_{s_{\alpha}\mu}$ is an equivalence of categories.

2.4 Restricted Lie algebras

2.4.1 Definition

We now move on to discuss some preliminaries for the chapter of the thesis dealing with the representation theory of truncated current Lie algebras over fields of prime characteristic. For $\S 2.4-\S 2.5$ and the entirety of Chapter 4, we fix \Bbbk an algebraically closed field of characteristic p>0; all Lie algebras, associative algebras, vector spaces etc. will be over \Bbbk . The first important concept we will need is that of a restricted Lie algebra. Let $\mathfrak g$ be a Lie algebra over \Bbbk . Then a p^{th} power map on $\mathfrak g$ is a map

$$(ullet)^{[p]}: \mathfrak{g}
ightarrow \mathfrak{g}$$

such that the map $\xi: \mathfrak{g} \to U(\mathfrak{g})$ given by $\xi(x) = x^p - x^{[p]}$ satisfies:

$$\xi(x) \in Z(\mathfrak{g})$$

$$\xi(x+y) = \xi(x) + \xi(y)$$

$$\xi(\alpha x) = \alpha^p \xi(x)$$

for any $x, y \in \mathfrak{g}$ and $\alpha \in \mathbb{k}$. A Lie algebra \mathfrak{g} equipped with such a map is called a restricted Lie algebra. In addition, we write $Z_p(\mathfrak{g}) \subset Z(\mathfrak{g})$ for the image of ξ and refer to it as the p-centre of $U(\mathfrak{g})$.

Example 2.4.1. Let \mathfrak{g} be the one-dimensional Lie algebra, and let $0 \neq x \in \mathfrak{g}$. Then for any $\lambda \in \mathbb{k}$, there is a p^{th} power map on \mathfrak{g} given by $(\alpha x)^{[p]} = \alpha^p \lambda x$.

Example 2.4.2. Let G be an algebraic group over \mathbb{R} . Then there is a natural restricted structure on $\mathfrak{g} = \operatorname{Lie}(G)$ given by viewing \mathfrak{g} as invariant derivations on G (as described in §2.1) and taking $x^{[p]}$ to be the p^{th} power of x as a derivation (which in characteristic p will again be an invariant derivation). In the case $G = \operatorname{GL}_n$, SL_n , SO_n or Sp_{2n} , the p^{th} power map on \mathfrak{g} obtained in this way is the same as the map given by taking the p^{th} matrix power, where we view $\mathfrak{g} = \mathfrak{gl}_n$, \mathfrak{sl}_n , \mathfrak{so}_n , \mathfrak{sp}_{2n} as subalgebras of \mathfrak{gl}_n or \mathfrak{gl}_{2n} in the usual way.

Any p^{th} power map $(\bullet)^{[p]}$ allows us to define a notion of semisimple and nilpotent elements for any restricted Lie algebra in the following way:

Definition 2.4.3. Let \mathfrak{g} be a restricted Lie algebra. Then an element $x \in \mathfrak{g}$ is:

- Semisimple if $x \in \text{span}\{x^{[p]^i} : i \ge 1\}$
- Nilpotent if $x^{[p]^i} = 0$ for sufficiently large i.

We can then give a notion of 'Jordan decomposition' in any restricted Lie algebra via the following result.

Lemma 2.4.4. Let \mathfrak{g} be a restricted Lie algebra and let $x \in \mathfrak{g}$. Then there exist unique $s, n \in \mathfrak{g}$ such that s is semisimple, n is nilpotent, x = s + n, and [s, n] = 0. Furthermore, $\mathfrak{g}^x = (\mathfrak{g}^s)^n$.

Proof. Such s and n exist by [50, Theorem 2.3.5]. To see that $\mathfrak{g}^x = (\mathfrak{g}^s)^n$, we use the general fact that there always exists a faithful restricted representation $\mathfrak{g} \hookrightarrow \mathfrak{gl}(V)$ and by the classical Jordan decomposition we have $\mathfrak{gl}(V)^x = (\mathfrak{gl}(V)^s)^n$. The result then follows by taking the intersection of both sides with \mathfrak{g} .

2.4.2 The standard hypotheses for reductive groups

In small characteristics, many families of Lie algebras of reductive groups exhibit very different behaviour from the characteristic 0 or large characteristic cases. For example, in general \mathfrak{sl}_n is a simple Lie algebra but if p divides n then it has a one-dimensional centre. To exclude these cases, in the prime characteristic portion of this thesis we impose the following conditions on our reductive groups G:

- (S1) The derived subgroup of G is simply connected.
- (S2) The characteristic p is 'good' for each irreducible component of the root system of G. This is defined in the following way: make a choice of simple roots $\{\alpha_1, \ldots, \alpha_n\}$, and write the highest root $\alpha^{\#}$ as a sum of simple roots $\alpha^{\#} = \sum k_i \alpha_i$. Then p is good for an irreducible root system if it does not divide any k_i .
- (S3) There is a non-degenerate G-invariant bilinear form κ on $\mathfrak{g} = \text{Lie}(G)$.

These conditions are called the *standard hypotheses* and we will refer to a reductive group G satisfying (S1) – (S3) as a *standard reductive group*; we briefly mention that this definition is slightly stronger than the one sometimes used elsewhere in the literature (see the remarks in [36, §4] for a more detailed discussion of the various notions of 'standard reductive groups').

Remark 2.4.5. The condition (S3) ensures the existence of an isomorphism $f: \mathfrak{g} \to \mathfrak{g}^*$ between the adjoint and coadjoint modules given by $(f(x))(y) = \kappa(x,y)$

Example 2.4.6. The 'bad' primes for each irreducible root system are as follows:

- Type A_n : None.
- Types B_n, C_n, D_n : p = 2.
- Types E_6, E_7, F_4, G_2 : p = 2, 3.
- Type E_8 : p = 2, 3, 5.

Furthermore, if G is almost simple and (S1) and (S2) hold, then in addition (S3) holds unless G is of type A_n for n such that p divides n + 1.

In this thesis we will primarily be concerned with the Lie algebras of standard reductive groups as opposed to the groups themselves. These Lie algebras can be classified in the following way (see for example [27, §2.9] or [44, §2.1]):

Proposition 2.4.7. Let G be a standard reductive group and let $\mathfrak{g} = \text{Lie}(G)$. Then \mathfrak{g} is a direct sum of Lie algebras of the following forms:

- Simple Lie algebras of types B, C, D, E, F, G in good characteristic.
- \mathfrak{sl}_n where p does not divide n.
- \mathfrak{gl}_n where p divides n.
- Restricted tori, i.e. abelian Lie algebras equipped with a bijective p-th power map.

2.4.3 Representation theory and the Kac–Weisfeiler conjectures

We now discuss a few relevant features of the representation theory of restricted Lie algebras which will use later in Chapter 4. The first of these is the following result, which first appeared in [13, Theorem 5.1] and illustrates some of the differences between the characteristic 0 and prime characteristic settings.

Proposition 2.4.8. Let \mathfrak{g} be a Lie algebra over an algebraically closed field of characteristic p > 0. Then the simple $U(\mathfrak{g})$ -modules are all finite-dimensional, and the dimensions of the simple modules are bounded above.

Remark 2.4.9. In the case \mathfrak{g} is restricted, this result will follow from the discussion below; however it is true in general even without the restricted hypothesis.

Remark 2.4.10. This result is in contrast to the characteristic 0 case, where any non-abelian Lie algebra has infinite-dimensional simple modules. For example, if $\mathfrak{g} = \mathfrak{sl}_2(\mathbb{C})$

then for most values of λ the Verma module $M(\lambda)$ described in §2.3.1 is an infinite-dimensional simple module. Even if we restrict our attention to finite-dimensional simple $\mathfrak{sl}_2(\mathbb{C})$ -modules, there is (up to isomorphism) exactly one simple module of any positive integer dimension and hence the dimensions of finite dimensional simple modules are not bounded.

Now, if \mathfrak{g} is a restricted Lie algebra and M is a simple $U(\mathfrak{g})$ -module, then for all $x \in \mathfrak{g}$ the element $x^p - x^{[p]}$ acts on M by scalars since it lies in $Z(\mathfrak{g})$. Using the axioms of a p^{th} power map, one can deduce that in fact there exists $\chi \in \mathfrak{g}^*$ such that $x^p - x^{[p]}$ acts on M by $\chi(x)^p$ for all $x \in \mathfrak{g}$; we call this χ the p-character of the simple module M. Hence M can also be viewed as a simple module for the algebra $U_{\chi}(\mathfrak{g}) := U(\mathfrak{g})/(x^p - x^{[p]} - \chi(x)^p)$, which is a finite-dimensional algebra of dimension $p^{\dim(\mathfrak{g})}$ referred to as a reduced enveloping algebra.

By Proposition 2.4.8 the simple modules for a restricted Lie algebra are all finite-dimensional of dimension less than some fixed upper bound. It is not hard to show (see e.g. [24, Remark 2.8]) that this upper bound is at most $p^{\frac{1}{2}\dim(\mathfrak{g})}$. On the other hand, finding a precise value for this upper bound is much more difficult. One can also ask more generally what the possible dimensions of simple modules are. These questions led Kac-Weisfeiler to make the follow conjectures in [55].

Conjecture 2.4.11 (KW1). Let \mathfrak{g} be a restricted Lie algebra. Then the maximal dimension of a simple $U(\mathfrak{g})$ -module is $p^{\frac{1}{2}(\dim(\mathfrak{g})-\inf(\mathfrak{g}))}$.

Theorem 2.4.12 ([40], KW2). Let G be a standard reductive group, and let $\mathfrak{g} = \text{Lie}(G)$. Then for any p-character χ and any simple $U_{\chi}(\mathfrak{g})$ -module M, we have that $p^{\frac{1}{2}(\dim(\mathfrak{g})-\dim(\mathfrak{g}^{\chi}))}$ divides $\dim M$.

Remark 2.4.13. The statement of (KW1) still makes sense even if \mathfrak{g} is not restricted; however there exists a counterexample in this case (see [54]).

Remark 2.4.14. Although (KW2) was proved by Premet in [40], Kac later conjectured

in a review of Premet's paper [28] that it holds for any algebraic group G under certain mild conditions on the characteristic.

Outside of the case where G is a standard reductive group, very little is known about classes of groups for which (KW2) holds. On the other hand, there are several classes of Lie algebras for which (KW1) is known to hold; these include:

- Any restricted completely solvable Lie algebra [55].
- Any restricted Lie algebra \mathfrak{g} such that there exists $\chi \in \mathfrak{g}^*$ with \mathfrak{g}^{χ} a torus [43]. In particular, this includes all Lie algebras of standard reductive groups.
- Any centraliser $(\mathfrak{gl}_n)^e$ of a nilpotent element e in \mathfrak{gl}_n [53, Theorem 4].

In addition, a result of Martin–Stewart–Topley [33, Theorem 1.1] states that for any fixed n, there exists p_0 such that (KW1) holds for any restricted subalgebra of \mathfrak{gl}_n over an algebraically closed field of characteristic greater than p_0 (although their methods do not yield an explicit bound on p_0).

2.5 Truncated current groups and Lie algebras

The main class of Lie algebras we study in this thesis are the *truncated current Lie algebras*, which are defined as follows:

Definition 2.5.1. Let \mathfrak{g} be a Lie algebra. Then the m^{th} truncated Lie algebra on \mathfrak{g} , denoted \mathfrak{g}_m , is:

$$\mathfrak{g}_m := \mathfrak{g} \otimes \mathbb{k}[t]/(t^{m+1})$$

where the Lie bracket on \mathfrak{g}_m is given by

$$[x \otimes t^i, y \otimes t^j] = [x, y] \otimes t^{i+j}.$$

If G is an algebraic group and $\mathfrak{g} = \operatorname{Lie}(G)$, then the following result (see for example [38, Appendix]) shows that \mathfrak{g}_m is also the Lie algebra of an algebraic group G_m called the group of m-jets on G. It also provides an alternative characterisation of truncated current Lie algebras via jet schemes; for our purposes we usually only need the existence of the group G_m , so we will not discuss the theory of jet schemes in detail here.

Theorem 2.5.2. Let G be an algebraic group and let $\mathfrak{g} = \text{Lie}(G)$. Write G_m for the group of m-jets on G. Then $\mathfrak{g}_m = \text{Lie}(G_m)$.

Remark 2.5.3. In the case m = 1, the group G_1 can be viewed as the semidirect product $G \ltimes \mathfrak{g}$, where G acts on \mathfrak{g} via the adjoint representation and we view \mathfrak{g} as a group under addition. Similarly, the Lie algebra \mathfrak{g}_1 can be viewed as the semidirect product $\mathfrak{g} \ltimes \mathfrak{g}_{ab}$ where \mathfrak{g}_{ab} is the Lie algebra with the same underlying vector space as \mathfrak{g} but the abelian Lie bracket. In the case m > 1, one can still view G_m as $G \ltimes \mathfrak{g}^m$ where G acts diagonally on \mathfrak{g}^m via the adjoint action, but here the group structure one endows \mathfrak{g}^m with is more complicated then just addition.

We will be interested in truncated current Lie algebras in the case \mathfrak{g} is the Lie algebra of a reductive group (under the standard hypotheses when we work in prime characteristic). Fix a reductive group G with Lie algebra \mathfrak{g} and recall the notation of §2.2. We now define some notation used throughout the thesis.

The associative algebra $\mathbb{k}[t]/(t^{m+1})$ has a natural grading given by letting t^i lie in degree i, and this grading induces a grading on the Lie algebra \mathfrak{g}_m by letting $x \otimes t^i$ lie in degree i. We write $\mathfrak{g}_m^{(i)}$ for the i^{th} graded piece of \mathfrak{g}_m , and similarly write $\mathfrak{g}_m^{(\geq i)}$ and $\mathfrak{g}_m^{(<i)}$ for the sums of the graded pieces of degrees $\geq i$ and < i respectively.

When writing elements of \mathfrak{g}_m of the form $x \otimes t^i$, we will often omit the tensor product sign and simply write xt^i . Additionally, for $\alpha \in \Phi^+$ and $0 \le i \le m$, we set:

$$e_{\alpha,i} := e_{\alpha}t^i$$

$$h_{\alpha,i} := h_{\alpha}t^i$$

$$f_{\alpha,i} := f_{\alpha}t^{i}$$

The set $\{e_{\alpha,i}: \alpha \in \Phi^+, 0 \leq i \leq m\} \cup \{h_{\alpha,i}: \alpha \in \Delta, 0 \leq i \leq m\} \cup \{f_{\alpha,i}: \alpha \in \Phi^+, 0 \leq i \leq m\}$ is then a basis for \mathfrak{g}_m provided G is semisimple; again if G is reductive then we extend this to a basis by adding additional elements of \mathfrak{h} .

At several points we consider linear functions on \mathfrak{g}_m and \mathfrak{h}_m . Given $\lambda \in \mathfrak{h}_m^*$, we set:

$$\lambda_i = \lambda|_{\mathfrak{h}_m^{(i)}}$$

$$\lambda_{\geq i} = \lambda|_{\mathfrak{h}_m^{(i)}}$$

We may view λ_i as an element of \mathfrak{h}^* by identifying $\mathfrak{h}_m^{(i)}$ with \mathfrak{h} , and we often identify \mathfrak{h}_m^* with $(\mathfrak{h}^*)^{m+1}$ by identifying λ with $(\lambda_0, \ldots, \lambda_m)$. We will also sometimes write elements of \mathfrak{h}_m^* in the form (λ, μ) for $\lambda \in \mathfrak{h}^*$ and $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$, where $\mathfrak{h}^* \oplus (\mathfrak{h}_m^{(\geq 1)})^*$ is identified with \mathfrak{h}_m^* in the natural way. Furthermore, we will also sometimes view $\lambda \in \mathfrak{h}^*$ as an element of \mathfrak{g}^* by declaring that $\lambda(\mathfrak{n}) = \lambda(\mathfrak{n}^-) = 0$, for instance when we wish to consider the coadjoint stabiliser \mathfrak{g}^{λ} of an element $\lambda \in \mathfrak{h}^*$.

If $\kappa : \mathfrak{g} \times \mathfrak{g} \to \mathbb{k}$ is a non-degenerate G-invariant bilinear form on \mathfrak{g} then there is a non-degenerate G_m -invariant bilinear form κ_m on \mathfrak{g} given by:

$$\kappa_m(xt^i, yt^j) = \delta_{i+j,m}\kappa(x, y).$$

Hence if \mathfrak{g} is the Lie algebra of a reductive group in either characteristic 0 or prime characteristic under the standard hypotheses, we have an isomorphism of \mathfrak{g}_m -modules $\mathfrak{g}_m \cong \mathfrak{g}_m^*$ by the same argument as in Remark 2.4.5.

2.5.1 Symmetric invariants and the centre of the enveloping algebra

We now describe the structure of the symmetric invariants $S(\mathfrak{g}_m)^{G_m}$ and of the centre $Z(\mathfrak{g}_m)$ of the enveloping algebra $U(\mathfrak{g}_m)$ of a truncated current Lie algebra. We begin by defining a collection of divided power operators $\partial^{(k)}$ on $S(\mathfrak{g}_m)$ as follows. Set $\partial^{(0)} = \mathrm{id}_{S(\mathfrak{g}_m)}$, and then for $1 \leq k \leq m$ we define inductively:

$$\partial^{(k)}(xt^{j}) = \begin{cases} \binom{j}{k}xt^{j-k} & \text{if } j \ge k, \\ 0 & \text{otherwise,} \end{cases}$$
$$\partial^{(k)}(fg) = \sum_{i+j=k} \partial^{(i)}(f)\partial^{(j)}(g).$$

for any $0 \le j \le m$, $x \in \mathfrak{g}$, and f, g homogeneous elements of $S(\mathfrak{g}_m)$.

Now, by the Chevalley Restriction Theorem, $S(\mathfrak{g})^G$ is a graded polynomial algebra generated by rank(\mathfrak{g}) algebraically independent homogeneous elements. Choose such elements p_1, \ldots, p_r . There is an inclusion map $S(\mathfrak{g}) \hookrightarrow S(\mathfrak{g}_m^{(m)})$ given by extending the map $\mathfrak{g} \hookrightarrow \mathfrak{g}_m^{(m)}$ which sends $x \mapsto x \otimes t^m$. By considering the action of G_m on $\mathfrak{g}_m^{(m)}$, this map then descends to an inclusion $S(\mathfrak{g})^G \to S(\mathfrak{g}_m^{(m)})^{G_m}$. Abusing notation, view the p_i as elements of $S(\mathfrak{g}_m)^{G_m}$ via this inclusion.

The following result was proved in characteristic 0 by Raïs–Tauvel [45, §3], and in a slightly different form in prime characteristic by Arakawa–Topley–Villareal [3, Theorem 4.4].

Theorem 2.5.4. The symmetric invariants $S(\mathfrak{g}_m)^{G_m}$ form a polynomial algebra in (m+1) rank (\mathfrak{g}) variables, generated by the algebraically independent elements $\{\partial^{(k)}(p_j): 1 \leq j \leq r, 0 \leq k \leq m\}$.

Now, for $0 \le k \le m$, we define another map $d^{(k)}: S(\mathfrak{g}_m^{(m)}) \to S(\mathfrak{g}_m)$ by setting

$$d^{(k)}(f) = \sum_{x} {m \choose k} x_{m-k} \frac{df}{dx_m}$$

where the sum is taken over the basis of \mathfrak{g} fixed in §2.2 and we write x_i as shorthand for $x \otimes t^i$. Here by df/dx_m we mean that we view f as a polynomial in our chosen basis elements of $\mathfrak{g} \otimes t^m$ and then differentiate with respect to x_m . For example, if x, y are two basis elements of \mathfrak{g} and $f = x_m^2 y_m$, then $df/dx_m = 2x_m y_m$ and $df/dy_m = x_m^2$, so $d^{(k)}(f) = \binom{m}{k}(2x_{m-k}x_m y_m + y_{m-k}x_m^2)$.

The following corollary with be useful at several points. In characteristic 0 it was proved in [45, Lemma 3.2(ii)], but the same argument is still valid in prime characteristic; in any case we shall only need it in characteristic 0.

Lemma 2.5.5. For any $1 \le j \le r, 0 \le k \le m$, there exists $q_j^{(k)} \in S(\mathfrak{g}_m^{(m-k+1)})$ such that $\partial^{(k)} p_j = d^{(k)} p_j + q_j^{(k)}$.

Although the map $d^{(k)}$ does not preserve G_m invariants, we do have the following result which we shall use later:

Lemma 2.5.6. The map $d^{(k)}$ preserves $\operatorname{ad}(\mathfrak{h})$ invariants, where \mathfrak{h} is viewed as a subalgebra of \mathfrak{g}_m via the natural inclusion $\mathfrak{g} \hookrightarrow \mathfrak{g}_m^{(0)}$.

Proof. The $\mathrm{ad}(\mathfrak{h})$ -invariants in $S(\mathfrak{g}_m)$ are just the zero weight space for the action of $\mathrm{ad}(\mathfrak{h})$, but $d^{(k)}$ preserves $\mathrm{ad}(\mathfrak{h})$ weight spaces by definition.

We now seek to use this description of $S(\mathfrak{g}_m)^{G_m}$ to describe $Z(\mathfrak{g}_m)$, the centre $U(\mathfrak{g}_m)$. For now we assume that we are in characteristic 0, the only case that we require in this thesis, although we will later make some brief remarks on the situation in prime characteristic. By [14, §2.4], there is an isomorphism $\omega: S(\mathfrak{g}_m) \to U(\mathfrak{g}_m)$ of $\mathrm{ad}(\mathfrak{g}_m)$ -modules, called the symmetrisation map. Explicitly, this map is given by setting

$$\omega(x_1 \dots x_k) = \frac{1}{k!} \sum_{\sigma \in S_k} x_{\sigma(1)} \dots x_{\sigma(k)}$$

for any elements $x_1, \ldots, x_k \in \mathfrak{g}_m$ and extending linearly. We have $Z(\mathfrak{g}_m) = U(\mathfrak{g}_m)^{\mathfrak{g}_m}$ and since we are in characteristic 0 by exponentiating the action of \mathfrak{g}_m to an action of G_m this

is equal to $U(\mathfrak{g}_m)^{G_m}$. Hence ω descends to an isomorphism $S(\mathfrak{g}_m)^{G_m} \to Z(\mathfrak{g}_m)$. We then obtain as an immediate corollary of Theorem 2.5.4 the following:

Corollary 2.5.7. In characteristic 0, $Z(\mathfrak{g}_m)$ is a polynomial algebra generated by the $(m+1)\operatorname{rank}(\mathfrak{g})$ elements $\{\omega(\partial^{(k)}(p_j)): 1 \leq j \leq r, 0 \leq k \leq m\}$.

Remark 2.5.8. In prime characteristic the definition of the map ω given above is no longer valid. However, by [16, Theorem 1.2] the existence of an isomorphism of \mathfrak{g}_m -modules $U(\mathfrak{g}_m) \xrightarrow{\sim} S(\mathfrak{g}_m)$ is equivalent to the splitting of the inclusion $\mathfrak{g}_m \hookrightarrow U(\mathfrak{g}_m)$ of \mathfrak{g}_m -modules. Under the assumptions on \mathfrak{g} given in §2.4.2, this inclusion does indeed split; we will show this later in §2.5.3. A further problem is that it is no longer the case that $Z(\mathfrak{g}_m) = U(\mathfrak{g}_m)^{G_m}$, since the action of the Lie algebra \mathfrak{g}_m can no longer be exponentiated to an action of the group G_m . This can also be seen by observing that $Z(\mathfrak{g}_m)$ contains the p-centre $Z_p(\mathfrak{g}_m)$ described in §2.4.1, which in general is not contained in $U(\mathfrak{g}_m)^{G_m}$. In fact, one can show that $Z(\mathfrak{g}_m)$ is isomorphic to the tensor product of $U(\mathfrak{g}_m)^{G_m}$ and $Z_p(\mathfrak{g}_m)$ over their intersection.

2.5.2 Indices of truncated current Lie algebras

Recall that in §2.1 we defined the index of a Lie algebra \mathfrak{g} to be the minimal dimension of a coadjoint stabiliser in \mathfrak{g} . We now discuss these quantities for truncated current Lie algebras. The following result is [45, Theorem 2.8(i)].

Theorem 2.5.9. Let \mathfrak{g} be any Lie algebra over a field of characteristic 0. Then $\operatorname{ind}(\mathfrak{g}_m) = (m+1)\operatorname{ind}(\mathfrak{g})$.

Most of the arguments from [45] are still valid in prime characteristic, but a crucial step fails if $m \geq p$. We give a proof in this case when \mathfrak{g} is the Lie algebra of an algebraic group satisfying the hypotheses of §2.4.2.

Theorem 2.5.10. Let \mathfrak{g} be the Lie algebra of a reductive group over an algebraically closed field of characteristic p > 0 satisfying the standard hypotheses. Then $\operatorname{ind}(\mathfrak{g}_m) =$

(m+1) ind (\mathfrak{g}) .

Proof. Let $\mathfrak{h}^{\text{reg}}$ denote the set of elements of \mathfrak{h} which are regular as elements of \mathfrak{g} (i.e. that have minimal dimensional centraliser), and define $\mathfrak{h}_m^{\text{reg}} := \{\sum x_i t^i \in \mathfrak{h}_m \mid x_0 \in \mathfrak{h}^{\text{reg}} \}$. Observe that if $x_0 \in \mathfrak{h}^{\text{reg}}$ then $\mathfrak{g}^{x_0} = \mathfrak{h}$. For $x \in \mathfrak{h}_m^{\text{reg}}$ it is easy to check that $\mathfrak{g}_m^x = \mathfrak{h}_m$; the inclusion \subseteq follows from the fact \mathfrak{h} is commutative, while the inclusion \supseteq can be shown by considering an arbitrary element $\sum y_i t^i \in \mathfrak{g}_m^x$ and using the fact $x_0 \in \mathfrak{h}^{\text{reg}}$ to inductively show that each $y_i \in \mathfrak{h}$. Furthermore since $[\mathfrak{h},\mathfrak{g}] \subseteq \mathfrak{n}^- \oplus \mathfrak{n}^+$ we have that $[\mathfrak{h}_m,\mathfrak{g}_m] \subseteq \mathfrak{n}_m^- \oplus \mathfrak{n}_m^+$. By considering dimensions, we deduce that $\mathrm{ad}(x)\mathfrak{g}_m = \mathfrak{n}_m^- \oplus \mathfrak{n}_m^+$ for $x \in \mathfrak{h}_m^{\text{reg}}$.

Now consider the morphism

$$\varphi: G_m \times \mathfrak{h}_m \to \mathfrak{g}_m$$

given the restriction of the adjoint action map $G_m \times \mathfrak{g}_m \to \mathfrak{g}_m$. By [52, Lemma 1.6] the differential $d_{(1,x)}\varphi : \mathfrak{g}_m \times \mathfrak{h}_m \to \mathfrak{g}_m$ is given by

$$d_{(1,x)}\varphi(u,v) = [x,u] + v.$$

For $x \in \mathfrak{h}_m^{\text{reg}}$ we see that the image of $d_{(1,x)}\varphi$ is $\mathfrak{n}_m^- \oplus \mathfrak{n}_m^+ \oplus \mathfrak{h}_m = \mathfrak{g}_m$, i.e. $d_{(1,x)}$ is surjective. It follows that $G_m \times \mathfrak{h}_m^{\text{reg}} \to \mathfrak{g}_m$ is a dominant morphism. Hence the conjugates of $\mathfrak{h}_m^{\text{reg}}$ are dense in \mathfrak{g}_m . The set of regular elements of \mathfrak{g}_m is open dense and so intersects the conjugates of $\mathfrak{h}_m^{\text{reg}}$ non-trivially. Hence there is a regular element whose centraliser is conjugate to \mathfrak{h}_m , which has dimension $(m+1)\operatorname{rank}(\mathfrak{g}) = (m+1)\operatorname{ind}(\mathfrak{g})$. The existence of the isomorphism $\mathfrak{g}_m \cong \mathfrak{g}_m^*$ of \mathfrak{g}_m -modules then implies that $\operatorname{ind}(\mathfrak{g}_m) = (m+1)\operatorname{ind}(\mathfrak{g})$, completing the proof.

2.5.3 Support varieties in the truncated current case

The final preliminary we shall need for the prime characteristic chapter of the thesis is the notion of the support variety of a module. Let M be a $U_{\chi}(\mathfrak{g})$ -module for some restricted Lie algebra \mathfrak{g} and $\chi \in \mathfrak{g}^*$. Then the support variety of M, denoted $\mathcal{V}_{\mathfrak{g}}(M)$, is an algebraic variety which reflects in certain ways the properties of the module. We will not give a definition of $\mathcal{V}_{\mathfrak{g}}(M)$ here; instead we refer the reader to [17, Definition 6.1] and only mention the properties we require. Before moving on to the main results we shall use, we mention two other interesting properties of support varieties. Firstly, the dimension of $\mathcal{V}_{\mathfrak{g}}(M)$ is equal to the growth rate of a minimal projective resolution of M in the category $U_{\chi}(\mathfrak{g})$ -mod [17, Remark 6.3], and secondly there is a close relationship between support varieties and rank varieties [17, Theorem 6.4]. Their main use to us however will come from the following result, which is [17, Proposition 6.2 & Proposition 7.1]. This result gives a condition for certain modules to be projective, which will be useful later.

Theorem 2.5.11. Let M be a $U_{\chi}(\mathfrak{g})$ -module. Then:

- (i) M is projective as a $U_{\chi}(\mathfrak{g})$ -module if and only if $\mathcal{V}_{\mathfrak{g}}(M) = 0$.
- (ii) If $\mathfrak{l} \subseteq \mathfrak{g}$ is a restricted subalgebra, then $\mathcal{V}_{\mathfrak{l}}(M) = \mathcal{V}_{\mathfrak{g}}(M) \cap \mathfrak{l}$.

One can also define the support variety of the category of $U_{\chi}(\mathfrak{g})$ -modules, denoted $\mathcal{V}_{\mathfrak{g}}(\chi)$, by:

$$\mathcal{V}_{\mathfrak{q}}(\chi) = \mathcal{V}_{\mathfrak{q}}(M_1 \oplus \cdots \oplus M_k)$$

where M_1, \ldots, M_k form a complete set of irreducible $U_{\chi}(\mathfrak{g})$ -modules. By [41, Theorem 2.1(v), (vii)], we have that $\mathcal{V}_{\mathfrak{g}}(M) \subseteq \mathcal{V}_{\mathfrak{g}}(\chi)$ for any $U_{\chi}(\mathfrak{g})$ -module M.

We now state a result on support varieties in the truncated current case which we will need later. To prove this result we first need the following lemma:

Lemma 2.5.12. Let \mathfrak{g} be the Lie algebra of a standard reductive group in characteristic p > 0. Then the inclusion of $\operatorname{ad}(\mathfrak{g}_m)$ -modules $\mathfrak{g}_m \hookrightarrow U(\mathfrak{g}_m)$ splits.

Proof. First consider the case $\mathfrak{g} = \mathfrak{gl}_n$. In this case $(\mathfrak{gl}_n)_m$ can be identified with the Lie algebra of $n \times n$ matrices with coefficients in $\mathbb{k}[t]/(t^{m+1})$. We then obtain a map $U((\mathfrak{gl}_n)_m) \to (\mathfrak{gl}_n)_m$ which sends the PBW monomial $x_1 \otimes x_2 \otimes \cdots \otimes x_k$ to the product of matrices $x_1 x_2 \ldots x_k$ which gives the required splitting.

Now let \mathfrak{g} be the Lie algebra of a standard reductive group. By [18, Theorem A], \mathfrak{g} satisfies Richardson's property, i.e. there exists a faithful representation $\rho: \mathfrak{g} \to \mathfrak{gl}_n$ and an $\mathrm{ad}(\mathfrak{g})$ -submodule $M \subseteq \mathfrak{gl}_n$ such that $\mathfrak{gl}_n = \mathfrak{g} \oplus M$ as $\mathrm{ad}(\mathfrak{g})$ -modules. This then extends to a faithful representation $\rho_m: \mathfrak{g}_m \to (\mathfrak{gl}_n)_m$ such that $(\mathfrak{gl}_n)_m = \mathfrak{g}_m \oplus (M \otimes k[t]/(t^{m+1}))$ as $\mathrm{ad}(\mathfrak{g}_m)$ -modules. The desired splitting is then given by the composition $U(\mathfrak{g}_m) \to U((\mathfrak{gl}_n)_m) \to (\mathfrak{gl}_n)_m \to \mathfrak{g}_m$, where the first map is induced by ρ_m , the second is the the map given in the previous paragraph, and the third is projection along the decomposition $(\mathfrak{gl}_n)_m = \mathfrak{g}_m \oplus (M \otimes k[t]/(t^{m+1}))$.

By [16, Theorem 1.2] this is equivalent to the following corollary:

Corollary 2.5.13. As $ad(\mathfrak{g}_m)$ -modules $U(\mathfrak{g}_m)$ and $S(\mathfrak{g}_m)$ are isomorphic.

This enables us to give the following description of the support variety of the category of $U_{\chi}(\mathfrak{g}_m)$ -modules:

Proposition 2.5.14. Let \mathfrak{g} be the Lie algebra of a standard reductive group in characteristic p > 0. Then for any $\chi \in \mathfrak{g}_m^*$, we have $\mathcal{V}_{\mathfrak{g}_m}(\chi) = \mathcal{N}^{[p]}(\mathfrak{g}_m^{\chi})$, where $\mathcal{N}^{[p]}(\mathfrak{g}_m^{\chi})$ denotes the set of elements in \mathfrak{g}_m^{χ} annihilated by the p^{th} -power map on \mathfrak{g}_m^{χ} .

Proof. The argument from [43, Proposition 6.3] applies for any restricted Lie algebra \mathfrak{l} admitting an isomorphism $U(\mathfrak{l}) \cong S(\mathfrak{l})$ of ad(\mathfrak{l})-modules, so the desired result follows from Corollary 2.5.13.

CHAPTER 3

ORDINARY REPRESENTATION THEORY OF TRUNCATED CURRENT LIE ALGEBRAS

3.1 Category \mathcal{O} for truncated current Lie algebras

3.1.1 Definition and first properties of \mathcal{O}

Throughout this chapter, we work over \mathbb{C} (although all our results will be valid over any algebraically closed field of characteristic 0). Recall the notation for reductive groups and their Lie algebras introduced in §2.2 and for truncated current groups and Lie algebras introduced in §2.5. Our overall aim in this chapter is to generalise some of the results in the reductive case described in §2.3 and to state and prove a precise version of Theorem 1.1.3.

We begin by stating our definition of category \mathcal{O} for \mathfrak{g}_m , which is a generalisation of the $\mathfrak{g} = \mathfrak{sl}_2$, m = 1 case treated in [34]. We note that in the case m = 0, the category $\mathcal{O}(\mathfrak{g}_0)$ we obtain is precisely BGG category \mathcal{O} for $\mathfrak{g}_0 = \mathfrak{g}$.

Definition 3.1.1. The category $\mathcal{O}(\mathfrak{g}_m)$ is the full subcategory of $U(\mathfrak{g}_m)$ -mod with objects M satisfying the following:

- $(\mathcal{O}1)$ M is finitely generated.
- $(\mathcal{O}2)$ $\mathfrak{h}_m^{(0)}$ acts semisimply on M.

 $(\mathcal{O}3)$ $\mathfrak{h}_m^{(\geq 1)} \oplus \mathfrak{n}_m^+$ acts locally finitely on M.

Since $U(\mathfrak{g}_m)$ is a Noetherian algebra, by $(\mathcal{O}1)$ every $M \in \mathcal{O}(\mathfrak{g}_m)$ is Noetherian. Furthermore, it is not hard to see that $\mathcal{O}(\mathfrak{g}_m)$ is closed under submodules, quotients, and finite direct sums.

Remark 3.1.2. It is perhaps more natural to insist that all of \mathfrak{h}_m acts semisimply on M rather than just $\mathfrak{h}_m^{(0)}$. However, in the category defined by this stronger condition every module can be obtained by taking a module in the BGG category \mathcal{O} for \mathfrak{g} and letting $\mathfrak{g}_m^{(\geq 1)}$ acting trivially; see for example the case $\mathfrak{g} = \mathfrak{sl}_2$, m = 1 dealt with in [34, Corollary 5].

Let $M \in \mathcal{O}(\mathfrak{g}_m)$ and $\lambda \in \mathfrak{h}^*$, which we identify with $(\mathfrak{h}_m^{(0)})^*$ in the natural way. Define the λ weight space of M to be

$$M^{\lambda} := \{ v \in M \mid h \cdot v = \lambda(h)v \text{ for all } h \in \mathfrak{h}_m^{(0)} \}.$$

By $(\mathcal{O}2)$ we have $M = \bigoplus_{\lambda \in \mathfrak{h}^*} M^{\lambda}$ as vector spaces, and this is in fact a module grading of M where $U(\mathfrak{g}_m)$ is equipped with the natural grading of the root lattice. The elements of M^{λ} are called weight vectors of weight $\lambda \in \mathfrak{h}^*$. If $v \in M^{\lambda}$ is a weight vector satisfying $\mathfrak{n}_m^+ \cdot v = 0$ then we say that v is a maximal vector of weight λ .

Now let $\mu \in \mathfrak{h}_m^*$ and recall that $\mu_i := \mu|_{\mathfrak{h}_m^{(i)}}$. We say that $v \in M^{\mu_0}$ is a highest weight vector of weight μ if v is maximal of weight μ and

$$h \cdot v = \mu(h)v$$
 for all $h \in \mathfrak{h}_m$.

The following basic properties of weight spaces of $M \in \mathcal{O}(\mathfrak{g}_m)$ can be shown using the same argument as in BGG category \mathcal{O} , see for example [23, §1.1]:

Lemma 3.1.3. For any $M \in \mathcal{O}(\mathfrak{g}_m)$, we have $\dim(M^{\lambda}) < \infty$ for all $\lambda \in \mathfrak{h}^*$. In addition, $\{\lambda \in \mathfrak{h}^* : M^{\lambda} \neq 0\} \subseteq \bigcup_{\lambda \in I} (\lambda - \mathbb{Z}_{\geq 0} \Phi^+)$ for some finite subset $I \subseteq \mathfrak{h}^*$.

We also have the following useful corollary, which allows us obtain highest weight vectors from maximal vectors.

Corollary 3.1.4. Suppose that $M \in \mathcal{O}$ admits a nonzero maximal vector of weight $\lambda \in \mathfrak{h}^*$ in M. Then M admits a nonzero highest weight vector of weight μ for some $\mu \in \mathfrak{h}_m^*$ satisfying $\mu_0 = \lambda$.

Proof. Let V be the space of maximal vectors of weight λ . This is finite-dimensional since it is a subspace of M^{λ} , which is finite-dimensional by Lemma 3.1.3. Now, the action of $\mathfrak{h}_m^{(\geq 1)}$ preserves weight spaces (since $\mathfrak{h}_m^{(\geq 1)}$ commutes with $\mathfrak{h}_m^{(0)}$) and maximal vectors (by considering the commutation relations in $U(\mathfrak{g}_m)$) and hence $\mathfrak{h}_m^{(\geq 1)}$ acts on V. But since $\mathfrak{h}_m^{(\geq 1)}$ is commutative there exists some common eigenvector $v \in V$ for this action. Then by definition v is a highest weight vector of weight $(\lambda, \mu_1, \ldots, \mu_m) \in \mathfrak{h}_m^*$ for some $\mu_1, \ldots, \mu_m \in \mathfrak{h}^*$ where, as mentioned earlier, we identify \mathfrak{h}_m^* with $(\mathfrak{h}^*)^{m+1}$ in the natural manner.

3.1.2 Highest weight modules

We say M is a highest weight module of weight $\lambda \in \mathfrak{h}_m^*$ if M is generated by a highest weight vector of weight λ . The following result on highest weight filtrations is analogous to a result for BGG category \mathcal{O} , see for example [23, Corollary 1.2].

Lemma 3.1.5. Let $M \in \mathcal{O}(\mathfrak{g}_m)$. Then M has a finite filtration $0 = M_0 \subseteq M_1 \subseteq \cdots \subseteq M_k = M$ such that each M_{i+1}/M_i is a highest weight module. We call such a filtration a highest weight filtration of M.

Proof. By $(\mathcal{O}1)$ and $(\mathcal{O}2)$, there exists a finite set $\{v_1, v_2, \ldots, v_n\}$ of weight vectors which generate M. Let V be the $U(\mathfrak{n}_m \oplus \mathfrak{h}_m^{(\geq 1)})$ -module generated by the v_i , which is finite-dimensional by $(\mathcal{O}3)$, and proceed by induction on $\dim(V)$, observing that if m is maximal, then so is $(ht^i) \cdot m$ for any $h \in \mathfrak{h}$ and $0 \leq i \leq n$:

If $\dim(V) = 1$, then any non-zero element of V is a highest weight vector and generates M, so M is highest weight. If $\dim(V) > 1$ then pick some λ maximal among the weights of V. Then any vector of weight λ in V is maximal, so by a similar argument to the proof of Corollary 3.1.4 there is some $v \in V^{\lambda}$ which is a highest weight vector generating a highest weight submodule of M. Let M_1 be the highest weight submodule of M generated by v. Now consider $\overline{M} = M/M_1$, which is generated by the image \overline{V} of V in M/M_1 . Since $\dim(\overline{V}) < \dim(V)$ we are done by induction.

For $\lambda \in \mathfrak{h}_m^*$ we define the Verma module of weight λ , denoted $M(\lambda)$, to be

$$M(\lambda) := U(\mathfrak{g}_m) \otimes_{U(\mathfrak{b}_m)} \mathbb{C}_{\lambda}$$

where \mathbb{C}_{λ} is the one-dimensional $U(\mathfrak{b}_m)$ -module on which \mathfrak{h}_m acts via λ , and \mathfrak{n}_m acts by 0. The Verma modules are universal highest weight modules in the sense that every highest weight module is a quotient of a Verma module. They have the following properties generalising the classical case:

Lemma 3.1.6. Let $\lambda \in \mathfrak{h}_m^*$. Then:

- (i) dim $M(\lambda)^{\lambda_0} = 1$, and hence dim $M^{\lambda_0} = 1$ for every highest weight module M of weight $\lambda \in \mathfrak{h}_m^*$.
- (ii) Every highest weight module M admits a central character which depends only on its highest weight λ. In other words, for any λ ∈ ħ^{*}_m there is a homomorphism χ_λ: Z(g_m) → C such that for any highest weight module M of highest weight λ we have z · v = χ_λ(z)v for all z ∈ Z(g_m) and all v ∈ M.
- (iii) $M(\lambda)$ admits a unique maximal submodule and a unique simple quotient, which we denote $L(\lambda)$.
- (iv) Every simple object in $\mathcal{O}(\mathfrak{g}_m)$ is isomorphic to precisely one of these simple modules. Thus the modules

$$\{L(\lambda) \mid \lambda \in \mathfrak{h}_m^*\}$$

give a complete set of representatives for the isomorphism classes of simple modules in $\mathcal{O}(\mathfrak{g}_m)$.

Proof. The proofs of all four parts of this lemma are very similar to the classical case (see for example [23, Theorem 1.2]); we include them for the reader's convenience. For part (i), first observe that $M(\lambda)$ has a basis of the form $\{\prod_{\alpha\in\Phi^+,0\leq j\leq m}f^{i_{\alpha},j}_{\alpha,j}\otimes 1_{\lambda}:i_{\alpha,j}\in\mathbb{Z}_{\geq 0}\}$, where this product is taken over some fixed ordering on $\Phi^+\times\{0,1,\ldots,m\}$. The fact that $\dim M(\lambda)^{\lambda_0}=1$ then follows from the fact that $\prod_{\alpha\in\Phi^+,0\leq j\leq m}f^{i_{\alpha},j}_{\alpha,j}\otimes 1_{\lambda}$ has weight $\lambda_0-\sum_{\alpha,j}i_{\alpha,j}\alpha$. Now since $M(\lambda)$ lies in $\mathcal O$ which is closed under taking submodules, all submodules of $M(\lambda)$ are weight modules for the action of $\mathfrak{h}_m^{(0)}$. Since $1\otimes 1_{\lambda}$ generates $M(\lambda)$, for any proper submodule $N\subseteq M(\lambda)$ we have $N^{\lambda_0}=0$ and hence for any highest weight module $M:=(M(\lambda)/N)$ we have $\dim M^{\lambda_0}=1$ as required.

Part (ii) follows from the fact that $Z(\mathfrak{g}_m) \subseteq U(\mathfrak{g}_m)^{\mathfrak{h}}$ and hence any $z \in Z(\mathfrak{g}_m)$ preserves M^{λ_0} and hence acts on any $v \in M^{\lambda_0}$ by a scalar (using part (a)). But every element of M is of the form $u \cdot v$ for some $v \in M^{\lambda_0}$ and $u \in U(\mathfrak{g}_m)$, so since $z \cdot (u \cdot v) = u \cdot (z \cdot v)$ the result follows.

For part (iii), observe that since any non-zero element of $M(\lambda)^{\lambda_0}$ generates $M(\lambda)$, any proper submodule of $M(\lambda)$ must be weight (as observed in part (i)) and lie in a sum of weight spaces whose weights are strictly less than λ_0 . Hence the sum of all proper submodules of $M(\lambda)$ must again lie in a sum of weight spaces whose weights are strictly less than λ_0 and so is still proper, and is therefore the unique maximal submodule of $M(\lambda)$.

Part (iv) follows from Lemma 3.1.5 since it implies that any simple module must be highest weight and hence must be a quotient of some Verma module $M(\lambda)$, but the only such simple quotient is $L(\lambda)$.

3.1.3 Decomposition of category \mathcal{O}

A standard technique in studying the BGG category \mathcal{O} is to consider modules with a fixed central character for $U(\mathfrak{g})$. This refinement is also useful in our more general setting (see Theorem 3.2.2), but it will also be convenient to instead decompose $\mathcal{O}(\mathfrak{g}_m)$ by decomposing modules $M \in \mathcal{O}(\mathfrak{g}_m)$ into generalised eigenspaces for the action of $\mathfrak{h}_m^{(\geq 1)}$.

Fix $M \in \mathcal{O}(\mathfrak{g}_m)$ and $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$. We define the generalised eigenspace of eigenvalue μ to be

$$M^{(\mu)} = \{ v \in M \mid (h - \mu(h))^k v = 0 \text{ for all } k \gg 0, \ h \in \mathfrak{h}_m^{(\geq 1)} \}.$$

This definition then allows us to state the following lemma:

Lemma 3.1.7. Each $M \in \mathcal{O}(\mathfrak{g}_m)$ admits a direct sum decomposition:

$$M = \bigoplus_{\mu \in (\mathfrak{h}_m^{(\geq 1)})^*} M^{(\mu)}$$

and each $M^{(\mu)}$ is a $U(\mathfrak{g}_m)$ -submodule of M.

Proof. Since \mathfrak{h}_m preserves the weight spaces of M and each of these weight space is finite-dimensional, it follows that each M^{λ} decomposes into generalised eigenspaces for $\mathfrak{h}_m^{(\geq 1)}$. Therefore M admits a decomposition of the claimed form and it suffices to show that each summand $M^{(\mu)}$ is a $U(\mathfrak{g}_m)$ -submodule of M. In turn, it is then enough to show that $x \cdot v \in M^{(\mu)}$ for all $v \in M^{(\mu)}$ and $x = e_{\alpha,i}, h_{\alpha,i}$, or $f_{\alpha,i}$.

If $x = h_{\alpha,i}$, then $[x, (ht^j - \mu(ht^j))^n] = 0$ for any $h \in \mathfrak{h}$, $1 \leq j \leq m$ and $n \geq 0$, so $(ht^j - \mu(ht^j))^n \cdot (x \cdot v) = x \cdot ((ht^j - \mu(ht^j))^n \cdot v)$. But since $v \in M^{(\mu)}$, this is 0 for sufficiently large n and hence $x \cdot v \in M^{(\mu)}$. We now deal with the case $x = e_{\alpha,i}$; the case $x = f_{\alpha,i}$ is very similar. In the case i = m, since $e_{\alpha,m}$ and ht^j commute for $j \geq 1$ the same argument

as above shows $e_{\alpha,m} \cdot v \in M^{(\mu)}$ for any $v \in M^{(\mu)}$. Otherwise, observe that:

$$[e_{\alpha,i}, (ht^{j} - \mu(ht^{j}))^{n}] = \sum_{k=0}^{n-1} (ht^{j} - \mu(ht^{j}))^{k} [e_{\alpha,i}, ht^{j} - \mu(ht^{j})] (ht^{j} - \mu(ht^{j}))^{n-k-1}$$
$$= \sum_{k=0}^{n-1} (ht^{j} - \mu(ht^{j}))^{k} \alpha(h) e_{\alpha,i+j} (ht^{j} - \mu(ht^{j}))^{n-k-1}$$

Hence for $v \in M^{(\mu)}$, we have:

$$(ht^{j} - \mu(ht^{j}))^{n} \cdot (e_{\alpha,i} \cdot v) = e_{\alpha,i} \cdot ((ht^{j} - \mu(ht^{j}))^{n} \cdot v)$$
$$- \sum_{k=0}^{n-1} (ht^{j} - \mu(ht^{j}))^{k} \alpha(h) e_{\alpha,i+j} (ht^{j} - \mu(ht^{j}))^{n-k-1} \cdot v$$

By induction, $e_{\alpha,i+j} \cdot v \in M^{(\mu)}$ for $j \geq 1$, so for sufficiently large n this expression is 0 and hence $e_{\alpha,i} \cdot v \in M^{(\mu)}$.

Now we define the Jordan block of $\mathcal{O}(\mathfrak{g}_m)$ of weight $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$ to be the full subcategory $\mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ of $\mathcal{O}(\mathfrak{g}_m)$ whose objects are the modules $M \in \mathcal{O}$ such that $M = M^{(\mu)}$. We remark that these are not blocks in the sense of Definition 2.3.7 but are somewhat larger; however they play a similar role in the structure of category \mathcal{O} . The following direct sum decomposition of $\mathcal{O}(\mathfrak{g}_m)$, which we refer to as the Jordan decomposition of $\mathcal{O}(\mathfrak{g}_m)$, is an immediate corollary of Lemma 3.1.7.

Corollary 3.1.8.

$$\mathcal{O}(\mathfrak{g}_m) = igoplus_{\mu \in (\mathfrak{h}_m^{\geq 1})^*} \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$$

Remark 3.1.9. It is not hard to see that if $\lambda \in \mathfrak{h}_m^*$ and $\mu = \lambda|_{\mathfrak{h}_m^{(\geq 1)}}$ then both $M(\lambda)$ and $L(\lambda)$ lie in $\mathcal{O}^{(\mu)}(\mathfrak{g}_m)$. Combining Corollary 3.1.8 with the fact that Verma modules have unique maximal submodules and are therefore indecomposable, it follows that $L(\lambda)$ cannot occur as a subquotient of $M(\nu)$ unless $\lambda_{\geq 1} = \nu_{\geq 1}$

Let $\mathfrak{g}_m \to \mathfrak{g}_{m-1}$ be the natural quotient map with kernel $\mathfrak{g}_m^{(m)}$, and consider the pull-

back functor

$$p: \mathcal{O}(\mathfrak{g}_{m-1}) \longrightarrow \mathcal{O}(\mathfrak{g}_m)$$

induced by this map.

Lemma 3.1.10. Let $\lambda \in \mathfrak{h}_{m-1}^*$ and define $\nu \in \mathfrak{h}_m^*$ by $\nu(h_i) = \lambda(h_i)$ for $i = 0, \ldots, m-1$ and $\nu(h_m) = 0$ for all $h \in \mathfrak{h}$. Then $p(L(\lambda)) \cong L(\nu)$ as \mathfrak{g}_m -modules.

Proof. This follows immediately from the observations that $p(L(\lambda))$ is a simple highest weight module of weight ν , and that $L(\nu)$ is the unique such simple module in $\mathcal{O}(\mathfrak{g}_m)$. \square

Now we state and prove an easy equivalence between Jordan blocks of $\mathcal{O}(\mathfrak{g}_m)$ which arise by twisting by an automorphism of $U(\mathfrak{g}_m)$. We write $\mathfrak{g}' = [\mathfrak{g}, \mathfrak{g}]$ for the derived subalgebra of \mathfrak{g} . Note that $(\mathfrak{g}_m)' = (\mathfrak{g}')_m$, and so we may write \mathfrak{g}'_m unambiguously.

For $\lambda \in \mathfrak{h}_m^*$ we recall the notation $\lambda_{\geq 1} := \lambda|_{\mathfrak{h}_m^{(\geq 1)}}$. Any such λ can be extended to an element of \mathfrak{g}_m^* by declaring that $\lambda(\mathfrak{n}_m^{\pm}) = 0$, and we may abuse notation by identifying \mathfrak{h}_m^* with a subspace of \mathfrak{g}_m^* .

Lemma 3.1.11. Suppose that $\lambda, \nu \in \mathfrak{h}_m^*$ are such that $\lambda|_{\mathfrak{g}_m'} = \nu|_{\mathfrak{g}_m'}$. Then the categories $\mathcal{O}^{(\lambda_{\geq 1})}(\mathfrak{g}_m)$ and $\mathcal{O}^{(\nu_{\geq 1})}(\mathfrak{g}_m)$ are equivalent.

Proof. Since $\nu - \lambda$ is 0 on \mathfrak{g}'_m and $\mathfrak{g}_m = \mathfrak{g}'_m \oplus \mathfrak{z}(\mathfrak{g}_m)$, we can define an automorphism ϕ of $U(\mathfrak{g}_m)$ by declaring $\phi(x) = x - (\nu - \lambda)(x)$ for all $x \in \mathfrak{g}$ and then extending to all of $U(\mathfrak{g}_m)$. Twisting by ϕ then defines an autoequivalence of $U(\mathfrak{g}_m)$ -mod, and since ϕ preserves $U(\mathfrak{h}_m)$, $U(\mathfrak{h}_m^{(0)})$, and $U(\mathfrak{n}_m^+)$ it also defines an autoequivalence of $\mathcal{O}(\mathfrak{g}_m)$. Finally, suppose $M \in \mathcal{O}^{(\lambda_{\geq 1})}(\mathfrak{g}_m)$, and that $(x - \lambda(x))^n \cdot v = 0$ for some $v \in M$, $x \in \mathfrak{g}$, and $v \in M$. Then writing \cdot for the untwisted action and \cdot_{ϕ} for the twisted action, we have

$$(x - \nu(x))^n \cdot_{\phi} v = (x - \lambda(x))^n \cdot v = 0$$

and hence $\phi(M) \in \mathcal{O}^{(\nu_{\geq 1})}(\mathfrak{g}_m)$. It is then clear that twisting by ϕ defines an equivalence between $\mathcal{O}^{(\lambda_{\geq 1})}(\mathfrak{g}_m)$ and $\mathcal{O}^{(\nu_{\geq 1})}(\mathfrak{g}_m)$ with quasi-inverse given by twisting by ϕ^{-1} .

3.2 Parabolic induction

In this section we aim to prove the following theorem (a more precise version of Theorem 1.1.1), which allows us to relate the category $\mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ to a Jordan block of $\mathcal{O}(\mathfrak{l}_m)$ for a suitable Levi subalgebra $\mathfrak{l} \subseteq \mathfrak{g}$. Recall that if $\nu \in \mathfrak{h}^*$ then we extend ν to an element of \mathfrak{g}^* by setting $\nu(\mathfrak{n}^{\pm}) = 0$ and write \mathfrak{g}^{ν} for the coadjoint stabiliser of ν viewed as an element of \mathfrak{g}^* .

Theorem 3.2.1. Let $\lambda \in \mathfrak{h}_m^*$. Suppose that the stabiliser $\mathfrak{l} = \mathfrak{g}^{\lambda_m}$ is in standard Levi form, let $\mathfrak{p} := \mathfrak{l} + \mathfrak{n}^+ = \mathfrak{l} \oplus \mathfrak{r}$ be the standard parabolic subalgebra with Levi factor \mathfrak{l} containing \mathfrak{b} , and write \mathfrak{r} for the nilradical of \mathfrak{p} . Set $\mu = \lambda_{\geq 1} \in (\mathfrak{h}_m^{(\geq 1)})^*$. Then the categories $\mathcal{O}^{(\mu)}(\mathfrak{l}_m)$ and $\mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ are equivalent, and the functors inducing the equivalence are parabolic induction and \mathfrak{r}_m -invariants. The first is given by:

$$U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} (\bullet) : \mathcal{O}^{(\mu)}(\mathfrak{l}_m) \longrightarrow \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$$

sending $M \mapsto U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} M$, where we inflate a $U(\mathfrak{g}_m^{\mu_m})$ -module M into a $U(\mathfrak{p}_m)$ module by letting \mathfrak{r}_m act by 0. The second is given by

$$(ullet)^{\mathfrak{r}_m}:\mathcal{O}^{(\mu)}(\mathfrak{g}_m)\longrightarrow\mathcal{O}^{(\mu)}(\mathfrak{l}_m)$$

sending $M \longmapsto M^{\mathfrak{r}_m}$, where $M^{\mathfrak{r}_m} = \{v \in M : \mathfrak{r}_m \cdot v = 0\}$.

We first observe that $U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} (\bullet)$ is left adjoint to $(\bullet)^{\mathfrak{r}_m}$ since for M, N lying in the appropriate categories we have inverse isomorphisms

$$\operatorname{Hom}_{\mathfrak{l}_m}(M, N^{\mathfrak{r}_m}) \xrightarrow{\theta} \operatorname{Hom}_{\mathfrak{g}_m}(U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} M, N)$$

given by $\theta(f)(u \otimes v) = u \cdot f(v)$ and $\eta(g)(v) = g(1 \otimes v)$ for $u \in U(\mathfrak{g}_m)$ and $v \in M$.

In order to show that the adjoint functors $U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} (\bullet)$ and $(\bullet)^{\mathfrak{r}_m}$ are equivalences we consider the unit and counit of this adjunction. Let $\mathbb{1}_{\mathcal{C}}$ denote the identity endofunctor

of a category \mathcal{C} . The unit is the natural transformation $\psi: \mathbb{1}_{\mathcal{O}(\mathfrak{g}_m)} \to (U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} (\bullet))^{\mathfrak{r}_m}$ obtained by applying θ to the identity map $N^{\mathfrak{r}_m} \to N^{\mathfrak{r}_m}$, whilst the counit is the natural transformation $\varphi: U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} (\bullet)^{\mathfrak{r}_m} \to \mathbb{1}_{\mathcal{O}(\mathfrak{l}_m)}$ obtained by applying η to the identity map $U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} M \to U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} M$. In particular, for M, N in the appropriate categories we that have $\psi_M: M \to (U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} M)^{\mathfrak{r}_m}$ is given by $v \mapsto 1 \otimes v$ and that $\varphi_N: U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} N^{\mathfrak{r}_m} \to N$ is given by $u \otimes v \mapsto u \cdot v$.

To prove Theorem 3.2.1 it therefore suffices to show that ψ and φ are both natural isomorphisms. The proof is given in Section 3.2.2 and depends heavily on the exactness of the functor $(\bullet)^{\mathfrak{r}_m}$. The proof the exactness of this functor requires a careful study of the central characters of highest weight modules, which we now discuss.

3.2.1 Central characters

The main step in proving the exactness of the functor $(\bullet)^{\mathfrak{r}_m}$ is the following result which leads to a vanishing criterion for extensions. The proof relies on the well-known fact that if two $U(\mathfrak{g}_m)$ -modules admit different generalised central characters then there do not exist any non-split extensions between them.

For $\nu \in \mathfrak{h}^*$, we define $\Phi_{\nu} := \{\alpha \in \Phi : \nu(h_{\alpha}) = 0\} \subseteq \Phi$, the root system of the stabiliser \mathfrak{g}^{ν} . We also recall the notation χ_{λ} for central characters introduced in Lemma 3.1.6(ii). The main theorem we prove in this section is the following:

Theorem 3.2.2. Let $\lambda, \lambda' \in \mathfrak{h}_m^*$ such that $\lambda_{\geq 1} = \lambda'_{\geq 1}$ and \mathfrak{g}^{λ_m} is in standard Levi form. Then $\chi_{\lambda} = \chi_{\lambda'}$ if and only if $\lambda_0 - \lambda'_0 \in \mathbb{C}\Phi_{\lambda_m}$.

Before proceeding to the proof of this theorem, we state the following corollary, which will be a key ingredient in the proof of Theorem 3.2.1.

Corollary 3.2.3. Let $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$ be such that \mathfrak{g}^{μ_m} is in standard Levi form, and let $M \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ be indecomposable, Then there is a unique coset $\Xi_M \in \mathfrak{h}^*/\mathbb{C}\Phi_{\mu_m}$ such that if N is a highest weight subquotient of M of weight $\lambda \in \mathfrak{h}_m^*$, then $\lambda_{\geq 1} = \mu$ and $\lambda_0 + \mathbb{C}\Phi_{\mu_m} = \Xi_M$.

Proof. By Lemma 3.1.7 along with Remark 3.1.9, all subquotients of M must lie in the Jordan block $\mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ and hence $\lambda_{\geq 1} = \mu$. Since M has finite-dimensional weight spaces by Lemma 3.1.3 and is indecomposable it admits a generalised central character, i.e. there is a unique maximal ideal \mathfrak{m} of $Z(\mathfrak{g}_m)$ such that $\mathfrak{m}^k M = 0$ for $k \gg 0$. Thus all of the highest weight subquotients have the same central character by Lemma 3.1.6(ii). Corollary 3.2.3 now follows from Theorem 3.2.2.

Remark 3.2.4. Later we shall see that Theorem 3.2.2 and Corollary 3.2.3 hold even without the standard Levi type hypothesis; this will follow from Theorem 3.3.11. However since the proof of the latter theorem will use Theorem 3.2.2 we retain this hypothesis to keep these dependencies clear.

We now proceed to prove Theorem 3.2.2. We begin by recalling several facts from §2.5.1. First recall that the symmetric invariants $S(\mathfrak{g})^{\mathfrak{g}}$ are generated by $r = \operatorname{rank}(\mathfrak{g})$ algebraically independent homogeneous elements p_1, \ldots, p_r . Furthermore, we also recall the definitions of the maps $\partial^{(j)}, d^{(j)} : S(\mathfrak{g}_m^{(m)}) \to S(\mathfrak{g}_m)$. As discussed in §2.5.1 there is an embedding $S(\mathfrak{g})^{\mathfrak{g}} \hookrightarrow S(\mathfrak{g}_m)^{\mathfrak{g}_m}$; we will often abuse notation by identifying p_1, \ldots, p_r with their images under this embedding. We then have that $S(\mathfrak{g}_m)^{\mathfrak{g}_m}$ is generated by the r(m+1) elements $\partial^{(j)}p_i$, $i=1,\ldots,r,\ j=0,\ldots,m$. Finally, there is an isomorphism $\omega: S(\mathfrak{g}_m)^{\mathfrak{g}_m} \to Z(\mathfrak{g}_m)$. The elements of the centre that we are interested in are:

$$z_i^{(j)} := \omega(\partial^{(j)} p_i)$$
 for $i = 1, \dots, r$ and $j = 0, \dots, m$

Now let $U(\mathfrak{g}_m)^{\mathfrak{h}}$ be the invariant subalgebra under the adjoint action of \mathfrak{h} , and let $U(\mathfrak{g}_m)\mathfrak{n}_m^+$ be the left ideal of $U(\mathfrak{g}_m)$ generated by \mathfrak{n}_m^+ . The intersection $U(\mathfrak{g}_m)\mathfrak{n}_m^+ \cap U(\mathfrak{g}_m)^{\mathfrak{h}}$ is an ideal of $U(\mathfrak{g}_m)^{\mathfrak{h}}$, and it is not hard to see that the quotient by this ideal is isomorphic to $U(\mathfrak{h}_m)$. This yields a map $\pi': U(\mathfrak{g}_m)^{\mathfrak{h}} \to U(\mathfrak{h}_m)$ which can be described in the following way. Let x be a PBW monomial with the respect to the ordered basis of \mathfrak{g}_m defined in §2.5; in particular the order on this basis is compatible with the choice of Borel subalgebra $\mathfrak{b} \subseteq \mathfrak{g}$. Then π' sends x to itself if $x \in U(\mathfrak{h}_m)$ and 0 otherwise. The observations of [23, §1.7] are

still valid in our situation, and so we see that χ_{λ} coincides with the composition

$$U(\mathfrak{g}_m)^{\mathfrak{h}} \stackrel{\pi'}{\to} U(\mathfrak{h}_m) = \mathbb{C}[\mathfrak{h}_m^*] \stackrel{\text{ev}_{\lambda}}{\longrightarrow} \mathbb{C}$$

where the function $\operatorname{ev}_{\lambda}$ is the extension of $\lambda \in \mathfrak{h}_m^*$ to a map $\mathbb{C}[\mathfrak{h}_m^*] \cong S(\mathfrak{h}_m) \to \mathbb{C}$. This allows us to view χ_{λ} as a function not just from $Z(\mathfrak{g}_m)$ to \mathbb{C} but from $U(\mathfrak{g}_m)^{\mathfrak{h}}$ to \mathbb{C} .

Now for $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$ we define two maps $\xi_{\mu}, \eta_{\mu} : \mathfrak{h}^* \to \mathbb{C}^r$. For $\nu \in \mathfrak{h}^*$ we write (ν, μ) for the element of \mathfrak{h}_m^* which restricts to ν on $\mathfrak{h}_m^{(0)} \cong \mathfrak{h}$ and to μ on $\mathfrak{h}_m^{(\geq 1)}$. The first map ξ_{μ} is given by:

$$\xi_{\mu}(\nu) := ((\chi_{(\nu,\mu)} \circ \omega \circ d^{(m)})p_1, \dots, (\chi_{(\nu,\mu)} \circ \omega \circ d^{(m)})p_r) \in \mathbb{C}^r$$

whilst the second map η_{μ} is given by:

$$\eta_{\mu}(\nu) := ((\operatorname{ev}_{(\nu,\mu)} \circ d^{(m)} \circ \pi) p_1, \dots, (\operatorname{ev}_{(\nu,\mu)} \circ d^{(m)} \circ \pi) p_r) \in \mathbb{C}^r.$$

Here π denotes the composition $S(\mathfrak{g}) \to S(\mathfrak{h}) \hookrightarrow S(\mathfrak{g})$, where the map $S(\mathfrak{g}) \to S(\mathfrak{h})$ is given by restriction along the triangular decomposition $\mathfrak{g} = \mathfrak{n}^+ \oplus \mathfrak{h} \oplus \mathfrak{n}^-$.

Example 3.2.5. Consider the case $\mathfrak{g} = \mathfrak{sl}_2$ and m = 1. We let $\{e, h, f\}$ be the standard basis of \mathfrak{sl}_2 and write x_i for $x \otimes t^i$. The algebra $S(\mathfrak{sl}_2)^{SL_2}$ is a polynomial algebra generated by the element $p = \frac{1}{2}h^2 + 2ef$. Now, we can compute $\xi_{\mu}(\nu)$ as follows:

$$\xi_{\mu}(\nu) = (\operatorname{ev}_{(\nu,\mu)} \circ \pi' \circ \omega \circ d^{(1)}) (\frac{1}{2}h^{2} + 2ef)$$

$$= (\operatorname{ev}_{(\nu,\mu)} \circ \pi' \circ \omega) (2h_{0}h_{1} + 4e_{0}f_{1} + 4f_{0}e_{1})$$

$$= (\operatorname{ev}_{(\nu,\mu)} \circ \pi') (2h_{0}h_{1} + 2e_{0}f_{1} + 2f_{1}e_{0} + 2f_{0}e_{1} + 2e_{1}f_{0})$$

$$= (\operatorname{ev}_{(\nu,\mu)} \circ \pi') (2h_{0}h_{1} + 4f_{0}e_{1} + 4f_{1}e_{0} + 4h_{1})$$

$$= \operatorname{ev}_{(\nu,\mu)} (2h_{0}h_{1} + 4h_{1})$$

$$= 2\nu(h)\mu(h) + 4\mu(h)$$

Similarly, we can compute $\eta_{\mu}(\nu)$:

$$\eta_{\mu}(\nu) = (\operatorname{ev}_{(\nu,\mu)} \circ d^{(1)} \circ \pi) (\frac{1}{2}h^2 + 2ef)
= (\operatorname{ev}_{(\nu,\mu)} \circ d^{(1)}) (\frac{1}{2}h^2)
= \operatorname{ev}_{(\nu,\mu)} (2h_0h_1)
= 2\nu(h)\mu(h)$$

In particular, two Verma modules $M(\nu,\mu)$ and $M(\nu',\mu)$ lying in the same Jordan block have the same central character if and only if $\xi_{\mu}(\nu) = \xi_{\mu}(\nu')$. But by the above calculations, this is true if and only if $\eta_{\mu}(\nu) = \eta_{\mu}(\nu')$, and we see this occurs precisely when either $\mu = 0$ or $\nu = \nu'$.

Our approach to determining when Verma modules have the same central character in the general will follow the same approach as the example above. We start with the following lemma:

Lemma 3.2.6. $\xi_{\mu}(\nu) - \eta_{\mu}(\nu)$ depends only on μ_m , and in particular does not depend on ν .

Proof. It suffices to show that if we take any PBW monomial $x \in S(\mathfrak{g})^{\mathfrak{h}}$ and identify it with an element of $S(\mathfrak{g}_m)^{\mathfrak{h}}$ via the inclusion $S(\mathfrak{g}) \hookrightarrow S(\mathfrak{g}_m)$ the following holds:

$$d^{(m)}\pi(x) - \pi'\omega d^{(m)}(x) \in U(\mathfrak{h}_m^{(m)}).$$

This is because we can then apply $\operatorname{ev}_{(\nu,\mu)}$ to both terms to see that $\xi_{\mu}(\nu) - \eta_{\mu}(\nu)$ depends only on μ_m .

Let $x \in S(\mathfrak{g})^{\mathfrak{h}}$ be a monomial in the standard PBW ordering; we then consider four cases:

Case (1): $x \in S(\mathfrak{h})$. In this case, $\pi(x) = x$ and since $d^{(m)}(x) \in S(\mathfrak{h}_m)$, we have $d^{(m)}\pi(x) = d^{(m)}(x) = \pi'\omega d^{(m)}(x)$ since the restriction of $\pi' \circ \omega$ to $S(\mathfrak{h}_m)$ is the identity

map.

Case (2): x is of the form $f_{\beta}h_{\alpha_1}h_{\alpha_2}\dots h_{\alpha_k}e_{\beta}$, in which case:

$$d^{(m)}(\pi(x)) = d^{(m)}(0) = 0$$

$$\pi'(\omega d^{(m)}(x)) = \pi'(\omega (f_{\beta,m}h_{\alpha_1,m}h_{\alpha_2,m} \dots h_{\alpha_k,m}e_{\beta,0} + h_{\alpha_1,m}h_{\alpha_2,m} \dots h_{\alpha_k,m}e_{\beta,m}f_{\beta,0})$$

$$+ \sum_{j} f_{\beta,m}h_{\alpha_1,m}h_{\alpha_2,m} \dots h_{\alpha_{j-1},m}h_{\alpha_{j+1},m} \dots h_{\alpha_k,m}e_{\beta,m}h_{\alpha_j,0}))$$

$$= \frac{1}{2}h_{\alpha_1,m}h_{\alpha_2,m} \dots h_{\alpha_k,m}h_{\beta,m} + \frac{1}{2}h_{\alpha_1,m}h_{\alpha_2,m} \dots h_{\alpha_k,m}h_{\beta,m} + 0$$

$$= h_{\alpha_1,m}h_{\alpha_2,m} \dots h_{\alpha_k,m}h_{\beta,m}.$$

The first equality follows from the definition of $d^{(m)}$. To show the second equality, we first observe that π' and ω are both linear. Considering the first term, we see that for any $0 \le i \le j \le k$, we have:

$$\pi'(h_{\alpha_{1},m} \dots h_{\alpha_{i},m} e_{\beta,0} h_{\alpha_{i+1},m} \dots h_{\alpha_{j},m} f_{\beta,m} h_{\alpha_{j+1},m} \dots h_{\alpha_{k},m})$$

$$= \pi'(h_{\alpha_{1},m} \dots h_{\alpha_{i},m} e_{\beta,0} h_{\alpha_{i+1},m} \dots h_{\alpha_{k},m} f_{\beta,m})$$

$$= \pi'(h_{\alpha_{1},m} \dots h_{\alpha_{j},m} h_{\alpha_{j+1},m} \dots h_{\alpha_{k},m} [e_{\beta,0}, f_{\beta,m}]$$

$$+ \sum_{l \geq i} h_{\alpha_{1},m} \dots h_{\alpha_{l},m} [e_{\beta,0}, h_{\alpha_{l+1},m}] h_{\alpha_{l+2},m} \dots h_{\alpha_{k},m} f_{\beta,m}$$

$$+ f_{\beta,m} h_{\alpha_{1},m} h_{\alpha_{2},m} \dots h_{\alpha_{k},m} e_{\beta,0})$$

$$= \pi'(h_{\alpha_{1},m} h_{\alpha_{2},m} \dots h_{\alpha_{k},m} h_{\beta,m}$$

$$+ \sum_{l \geq i} \beta(h_{\alpha_{l}}) f_{\beta} h_{\alpha_{1},m} h_{\alpha_{2},m} \dots h_{\alpha_{k},m} e_{\beta,0}$$

$$+ f_{\beta,m} h_{\alpha_{1},m} h_{\alpha_{2},m} \dots h_{\alpha_{k},m} e_{\beta,0})$$

$$= h_{\alpha_{1},m} h_{\alpha_{2},m} \dots h_{\alpha_{k},m} h_{\beta,m}$$

and

$$\pi'(h_{\alpha_{1},m} \dots h_{\alpha_{i},m} f_{\beta,m} h_{\alpha_{i+1},m} \dots h_{\alpha_{j},m} e_{\beta,0} h_{\alpha_{j+1},m} \dots h_{\alpha_{k},m})$$

$$= \pi'(f_{\beta,m} h_{\alpha_{1},m} \dots h_{\alpha_{j},m} e_{\beta,0} h_{\alpha_{j+1},m} \dots h_{\alpha_{k},m})$$

$$= \pi'(\sum_{l \geq j} f_{\beta,m} h_{\alpha_{1},m} \dots h_{\alpha_{l},m} [e_{\beta,0}, h_{\alpha_{l+1},m}] h_{\alpha_{l+2},m} \dots h_{\alpha_{k},m}$$

$$+ f_{\beta,m} h_{\alpha_{1},m} h_{\alpha_{2},m} \dots h_{\alpha_{k},m} e_{\beta,0})$$

$$= \pi'(\sum_{l \geq j} \beta(h_{\alpha_{l}}) f_{\beta,m} h_{\alpha_{1},m} h_{\alpha_{2},m} \dots h_{\alpha_{k},m} e_{\beta,m}$$

$$+ f_{\beta,m} h_{\alpha_{1},m} h_{\alpha_{2},m} \dots h_{\alpha_{k},m} e_{\beta,0})$$

$$= 0.$$

Now, $\omega(f_{\beta,m}h_{\alpha_1,m}h_{\alpha_2,m}\dots h_{\alpha_k,m}e_{\beta,0})$ is a sum of (k+2)! terms. Half of these terms are of the form

$$\frac{1}{(k+2)!}h_{\alpha_1,m}\dots h_{\alpha_i,m}e_{\beta,0}h_{\alpha_{i+1},m}\dots h_{\alpha_j,m}f_{\beta,m}h_{\alpha_{j+1},m}\dots h_{\alpha_k,m}$$

and half are of the form

$$\frac{1}{(k+2)!}h_{\alpha_1,m}\dots h_{\alpha_i,m}f_{\beta,m}h_{\alpha_{i+1},m}\dots h_{\alpha_j,m}e_{\beta,0}h_{\alpha_{j+1},m}\dots h_{\alpha_k,m}$$

so $\pi'(\omega(f_{\beta,m}h_{\alpha_1,m}h_{\alpha_2,m}\dots h_{\alpha_k,m}e_{\beta,0})) = \frac{1}{2}h_{\alpha_1,m}h_{\alpha_2,m}\dots h_{\alpha_k,m}h_{\beta,m}$. Similar arguments for the other terms show that

$$\pi'(\omega(h_{\alpha_1,m}h_{\alpha_2,m}\dots h_{\alpha_k,m}e_{\beta,m}f_{\beta,0})) = \frac{1}{2}h_{\alpha_1,m}h_{\alpha_2,m}\dots h_{\alpha_k,m}h_{\beta,m}$$

and

$$\pi'(\omega(f_{\beta,m}h_{\alpha_1,m}h_{\alpha_2,m}\dots h_{\alpha_{j-1},m}h_{\alpha_{j+1},m}\dots h_{\alpha_k,m}e_{\beta,m}h_{\alpha_j,0}))=0.$$

Case (3): x is of the form $f_{\alpha_1} \dots f_{\alpha_n} h_{\beta_1} \dots h_{\beta_k} e_{\gamma}$ where n > 1 and $\gamma = \sum \alpha_i$ (so in

particular $\gamma > \alpha_i$ for all i). In this case:

$$d^{(m)}(\pi(x)) = d^{(m)}(0) = 0$$

$$\pi'(\omega d^{(m)}(x)) = \pi'(\omega (\sum_{i} f_{\alpha_{1},m} \dots f_{\alpha_{i-1},m} f_{\alpha_{i+1},m} \dots f_{\alpha_{n},m} h_{\beta_{1},m} \dots h_{\beta_{k},m} e_{\gamma,m} f_{\alpha_{i},0})$$

$$+ \sum_{j} f_{\alpha_{1},m} \dots f_{\alpha_{n},m} h_{\beta_{1},m} \dots h_{\beta_{j-1},m} h_{\beta_{j+1},m} \dots h_{\beta_{k},m} e_{\gamma,m} h_{\beta_{j},0}$$

$$+ f_{\alpha_{1},m} \dots f_{\alpha_{n},m} h_{\beta_{1},m} \dots h_{\beta_{k},m} e_{\gamma,0}))$$

$$= 0 + 0 + 0$$

Case (4): x is of the form $f_{\alpha_1} \dots f_{\alpha_n} h_{\beta_1} \dots h_{\beta_k} e_{\gamma_1} \dots e_{\gamma_p}$, where $n \geq 1, p > 1$ and $\sum \alpha_i = \sum \gamma_k$. In this case:

$$d^{(m)}(\pi(x)) = d^{(m)}(0) = 0$$

$$\pi'(\omega d^{(m)}(x)) = \pi'(\omega (\sum_{i} f_{\alpha_{1},m} \dots f_{\alpha_{i-1},m} f_{\alpha_{i+1},m} \dots f_{\alpha_{n},m} h_{\beta_{1},m} \dots h_{\beta_{k},m} e_{\gamma_{1},m} \dots e_{\gamma_{p},m} f_{\alpha_{i},0}$$

$$+ \sum_{j} f_{\alpha_{1},m} \dots f_{\alpha_{n},m} h_{\beta_{1},m} \dots h_{\beta_{j-1},m} h_{\beta_{j+1},m} \dots h_{\beta_{k},m} e_{\gamma_{1},m} \dots e_{\gamma_{p},m} h_{\beta_{j},0}$$

$$+ \sum_{l} f_{\alpha_{1},m} \dots f_{\alpha_{n},m} h_{\beta_{1},m} \dots h_{\beta_{k},m} e_{\gamma_{1},m} \dots e_{\gamma_{l-1},m} e_{\gamma_{l+1},m} \dots e_{\gamma_{p},m} e_{\gamma_{k},0}))$$

$$= 0 + 0 + 0$$

Lemma 3.2.7. For $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$ and $\nu, \nu' \in \mathfrak{h}^*$, we have $\chi_{(\nu,\mu)} = \chi_{(\nu',\mu)}$ if and only if $\chi_{(\nu,\mu)}(z_i^{(m)}) = \chi_{(\nu',\mu)}(z_i^{(m)})$ for $i = 1, \ldots, r$.

Proof. Since the centre $Z(\mathfrak{g}_m)$ is generated by the elements $\omega(\partial^{(k)}p_j)$ it follows that $\chi_{(\nu,\mu)} = \chi_{(\nu',\mu)}$ if and only if the characters coincide on these elements. Thus it suffices to show that $\chi_{(\nu,\mu)}(z_i^{(j)}) = \chi_{(\nu',\mu)}(z_i^{(j)})$ for all $j = 0, \ldots, m-1$.

We have $p_i \in S(\mathfrak{g}_m^{(m)})$ and it follows from the description of $\partial^{(j)}$ in §2.5.1 that $\partial^{(j)}p_i \in S(\mathfrak{g}_m^{(\geq m-j)})$. Hence we have $z_i^{(j)} \in U(\mathfrak{g}_m^{(\geq m-j)})$ and therefore $\chi_{(\nu,\mu)}(z_i^{(j)})$ only depends on μ . Hence $\chi_{(\nu,\mu)}(z_i^{(j)}) = \chi_{(\nu',\mu)}(z_i^{(j)})$ for all j < m, completing the proof.

We now apply Lemmas 3.2.6 and 3.2.7 to prove Theorem 3.2.2:

Proof of Theorem 3.2.2. Let λ, λ' satisfy the assumptions of the theorem. Thanks to Lemma 3.2.7 we must show that the given condition on λ_0, λ'_0 is equivalent to the condition that $\chi_{\lambda}(z_i^{(m)}) = \chi_{\lambda'}(z_i^{(m)})$ for all i.

Using Lemma 2.5.5, we have $z_i^{(m)} = \omega(d^{(m)}p_i) + \omega(q_i^{(m)})$. Since $q_i^{(m)} \in S(\mathfrak{g}_m^{\geq 1})$ it follows that $\chi_{\lambda}(z_i^{(m)}) = \chi_{\lambda'}(z_i^{(m)})$ if and only if $\chi_{\lambda} \circ \omega \circ d^{(m)}(p_i) = \chi_{\lambda'} \circ \omega \circ d^{(m)}(p_i)$ for all $i = 1, \ldots, r$. We note that this second equality is well-defined because by Lemma 2.5.6 $d^{(m)}$ sends \mathfrak{h} -invariants to \mathfrak{h} -invariants and ω is \mathfrak{h}_m -equivariant.

Setting $\mu := \lambda_{\geq 1} = \lambda'_{\geq 1}$ we can apply Lemma 3.2.6 to see that the central characters χ_{λ} and $\chi_{\lambda'}$ coincide if and only if $\eta_{\mu}(\lambda_0) = \eta_{\mu}(\lambda'_0)$. Let $\pi : \mathfrak{h}^* \to \mathfrak{h}^*/W \cong \mathbb{C}^r$ be the quotient map. It follows from the Chevalley restriction theorem [11, 3.1.37] that we can write this in coordinates as $\pi(\nu) \to (p_1|_{\mathfrak{h}^*}(\nu), \dots, p_r|_{\mathfrak{h}^*}(\nu))$. Now it is easy to see by a direct comparison of the two definitions that $\eta_{\mu}(\lambda_0)$ coincides with the differential $d_{\lambda_m}\pi(\lambda_0)$ of the quotient map π .

We now claim that $\ker \eta_{\mu} = \ker d_{\lambda_m} \pi = \mathbb{C} \Phi_{\lambda_m}$. This implies that $\eta_{\mu}(\lambda_0) = \eta_{\mu}(\lambda'_0)$ if and only if $\lambda_0 - \lambda'_0 \in \mathbb{C} \Phi_{\mu_m}$ completing the proof.

To see that $\ker d_{\lambda_m}\pi = \mathbb{C}\Phi_{\lambda_m}$, first observe that it follows from Chevalley's restriction theorem and a result of Richardson [46, Proposition 1.2] that $\dim \ker d_{\lambda_m}\pi = \dim \mathfrak{h} - \dim \mathfrak{z}(\mathfrak{g}^{\lambda_m})$. Note that $\mathbb{C}\Phi_{\lambda_m}$ has a basis consisting of simple roots α such that $h_{\alpha} \in [\mathfrak{g}^{\lambda_m},\mathfrak{g}^{\lambda_m}]$, so we have $\dim \mathbb{C}\Phi_{\lambda_m} = \dim \ker d_{\lambda_m}\pi = \dim \ker \eta_{\mu}$ and so the claim will follow if we can prove the inclusion $\mathbb{C}\Phi_{\lambda_m} \subseteq \ker \eta_{\mu}$.

Let Δ_{λ_m} be a system of simple roots for Φ_{λ_m} and let $\alpha \in \Delta_{\lambda_m}$. Then it is easily verified that $f_{\alpha,m} \otimes 1_{\lambda}$ is highest weight of weight (λ_0, μ) , and hence $M(\lambda_0, \mu)$ and $M(\lambda_0 + \alpha, \mu)$ have the same central character. Therefore $\alpha \in \ker \eta_{\mu}$ for any $\alpha \in \Delta_{\lambda_m}$, and hence $\mathbb{C}\Phi_{\lambda_m} \subseteq \ker \eta_{\mu}$ as required.

3.2.2 The parabolic induction functor

Fix a standard parabolic subalgebra \mathfrak{p} of \mathfrak{g} such that $\mathfrak{b} \subseteq \mathfrak{p}$ with corresponding Levi decomposition $\mathfrak{p} = \mathfrak{l} \oplus \mathfrak{r}$. In this section we prove that for $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$ such that $\mathfrak{g}^{\mu_m} = \mathfrak{l}$ the functor $(\bullet)^{\mathfrak{r}_m}$ is an exact functor from $\mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ to $\mathcal{O}^{(\mu)}(\mathfrak{l}_m)$, and then use this to prove Theorem 3.2.1. First we need the following lemma.

Proposition 3.2.8. Let $M \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ be an indecomposable module and let $\Xi_M \in \mathfrak{h}^*/\mathbb{C}\Phi_{\mu_m}$ be the coset determined by M as in Corollary 3.2.3. If $\mathfrak{l} = \mathfrak{g}^{\mu_m}$ is a standard Levi subalgebra then

$$M^{\mathfrak{r}_m} = \bigoplus_{\nu \in \Xi_M} M^{\nu}.$$

Proof. First observe that since M is indecomposable, then the weight λ of any highest weight subquotient of M must satisfy $\lambda_m = \mu_m$. We also note that since \mathfrak{g}^{μ_m} is the Levi factor of a standard parabolic, we have $\mathfrak{r} = \operatorname{span}\{e_\alpha : \alpha \in \Phi^+ \setminus \Phi_{\mu_m}\}$. By Lemma 3.1.5 we have a finite filtration $0 \subseteq M_1 \subseteq \cdots \subseteq M_k = M$ such that M_i/M_{i-1} has highest weight $\lambda^{(i)} \in \mathfrak{h}_m^*$. By Lemma 3.1.3, the weights of M lie in the set

$$\bigcup_{i=1}^{k} \{ \lambda_0^{(i)} - \mathbb{Z}_{\geq 0} \Phi^+ \}.$$

Furthermore, by Corollary 3.2.3 we have $\lambda_0^{(i)} + \mathbb{C}\Phi_{\mu_m} = \Xi_M$ for all i. It follows that the weights of M actually lie in the set $\lambda^{(i)} + \mathbb{C}\Phi_{\mu_m} - \sum_{\beta \in \Phi^+} \mathbb{Z}_{\geq 0}\beta$, for any choice of i.

In particular if $\nu \in \mathfrak{h}^*$ satisfies $\nu \in \lambda_0^{(j)} + \mathbb{C}\Phi_{\mu_m}$ for some j, then for any $\alpha \in \Phi^+ \setminus \Phi_{\mu_m}$, we have that $\nu + \alpha$ does not lie in $\lambda_0^{(i)} - \mathbb{Z}_{\geq 0}\Phi^+$ for any i. Therefore $\mathfrak{r}_m \cdot M^{\nu} = 0$.

Conversely, suppose $v \in M^{\mathfrak{r}_m}$ is of weight $v \in \mathfrak{h}^*$. Since \mathfrak{h}_m acts locally finitely and preserves weight spaces we can find a common eigenvector for \mathfrak{h}_m of weight v in $U(\mathfrak{h}_m) \cdot v$. Suppose the eigenvalue of this eigenvector is $\lambda' \in \mathfrak{h}_m^*$ (which by assumption satisfies $\lambda'_0 = v$). Then a quotient of $M(\lambda')$ occurs as a submodule of M. All highest weight modules are indecomposable since they admit unique maximal submodules, so the generalised central character of M must be $\chi_{\lambda'}$. Now Theorem 3.2.2 implies that

 $\nu \in \lambda'_0 + \mathbb{C}\Phi_{\mu_m}$. By Corollary 3.2.3 we see that for all i, λ'_0 and $\lambda_0^{(i)}$ lie in the same coset of \mathfrak{h}^* modulo $\mathbb{C}\Phi_{\mu_m}$ and so $\nu \in \lambda_0^{(i)} + \mathbb{C}\Phi_{\mu_m} = \Xi_M$.

Corollary 3.2.9. Suppose that $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$ such that $\mathfrak{g}^{\mu_m} = \mathfrak{l}$ is a standard Levi subalgebra. Then the functor $(\bullet)^{\mathfrak{r}_m} : \mathcal{O}^{(\mu)}(\mathfrak{g}_m) \to \mathcal{O}^{(\mu)}(\mathfrak{g}_m^{\mu_m})$ is exact.

Proof. Let $0 \to L \xrightarrow{f} M \xrightarrow{g} N \to 0$ be a short exact sequence in $\mathcal{O}^{(\mu)}(\mathfrak{g}_m)$. Since f is injective, the restriction of f to $L^{\mathfrak{r}_m}$ is still injective. If $m \in \ker(g) \cap M^{\mathfrak{r}_m}$, then m = f(l) for some $l \in L$ and for any $r \in \mathfrak{r}_m$, we have $f(r \cdot l) = r \cdot m = 0$, so $r \cdot l \in \ker(f) = 0$. Hence $l \in L^{\mathfrak{r}_m}$, and therefore $0 \to L^{\mathfrak{r}_m} \xrightarrow{f} M^{\mathfrak{r}_m} \xrightarrow{g} N^{\mathfrak{r}_m}$ is exact, i.e. we have left exactness of $(\bullet)^{\mathfrak{r}_m}$.

To complete the proof, we need only to show that $g: M^{\mathfrak{r}_m} \to N^{\mathfrak{r}_m}$ is surjective, and by Theorem 3.2.2 it suffices to consider the case where M and N both have filtrations by highest weight modules of weights (λ_i, μ) where $\lambda_i - \lambda_j \in \mathbb{Z}\Phi_{\mu_m}$ for all i, j. Let $v \in N^{\mathfrak{r}_m}$ have weight λ . By Proposition 3.2.8, we must have $\lambda \in \lambda_i + \mathbb{Z}\Phi_{\mu_m}$ for any λ_i . There then exists some $w \in g^{-1}(v)$ which is also of weight λ , so it is enough to show that any element of M of weight $\lambda \in \lambda_i + \mathbb{Z}\Phi_{\mu}$ is in $M^{\mathfrak{r}_m}$. But the weight of any element of M lies in $\bigcup (\lambda_i - \mathbb{Z}_{\geq 0}\Phi^+)$. In particular, if $\alpha \in \Phi^+ \setminus \Phi_{\mu_m}$ and $\lambda \in \lambda_i + \mathbb{C}\Phi_{\mu_m}$, then $\lambda + \alpha \nleq \lambda_i$ for any λ_i , so $M^{\lambda+\alpha} = 0$. Hence if $v \in M$ is of weight $\lambda \in \lambda_i + \mathbb{C}\Phi_{\mu_m}$, then for any $e_{\alpha,m} \in \mathfrak{r}_m$ we have $e_{\alpha,m} \cdot v \in M^{\lambda+\alpha} = 0$ so $v \in M^{\mathfrak{r}_m}$ as required.

Now we have exactness of $(\bullet)^{\mathfrak{r}_m}$, we can finally prove Theorem 3.2.1.

Proof of Theorem 3.2.1. Recall from the discussion following the statement of Theorem 3.2.1 that it suffices to show that the maps $\psi_M: M \to (U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} M)^{\mathfrak{r}_m}$ sending $v \mapsto 1 \otimes v$ and $\varphi_N: U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} N^{\mathfrak{r}_m} \to N$ sending $u \otimes v \mapsto u \cdot v$ are always isomorphisms.

The PBW theorem implies that $U(\mathfrak{g}_m)$ is free over $U(\mathfrak{p}_m)$ and hence $U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} (\bullet)$ is an exact functor. Furthermore $(\bullet)^{\mathfrak{r}_m}$ is exact by Corollary 3.2.9. It is therefore enough to check that φ_N and ψ_M are isomorphisms on highest weight modules, since a standard argument using the length of a highest weight filtration from Lemma 3.1.5 will allow

us to conclude that φ_N and ψ_M are in fact isomorphisms for all $N \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ and all $M \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m^{\mu_m})$

The map ψ_M is an isomorphism for highest weight modules M thanks to Proposition 3.2.8. Now suppose that $N \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ is a highest weight module with highest weight generator v. Since $\mathfrak{r}_m \cdot v = 0$ it follows that v lies in the image of φ_N and so φ_N is surjective. To prove injectivity, let $K = \ker(\varphi_N)$ and consider the short exact sequence

$$0 \to K \hookrightarrow U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} N^{\mathfrak{r}_m} \stackrel{\varphi_N}{\to} N \to 0.$$

By Corollary 3.2.9 we have another short exact sequence

$$0 \to K^{\mathfrak{r}_m} \hookrightarrow (U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} N^{\mathfrak{r}_m})^{\mathfrak{r}_m} \to N^{\mathfrak{r}_m} \to 0.$$

Now set $M = N^{\mathfrak{r}_m}$, which is a highest weight $U(\mathfrak{g}_m^{\mu_m})$ -module generated by v. The map $(U(\mathfrak{g}_m) \otimes_{U(\mathfrak{p}_m)} M)^{\mathfrak{r}_m} \to M$ is a $U(\mathfrak{g}_m)$ -equivariant map uniquely determined by $1 \otimes m \mapsto m$. Therefore it is the left inverse of ψ_M , which we already know to be bijective. It follows that $K^{\mathfrak{r}_m} = 0$, but since every nonzero object in $\mathcal{O}(\mathfrak{g}_m)$ admits a nonzero highest weight vector, it follows that K = 0 and so φ_N is an isomorphism for all highest weight modules N, completing the proof.

3.3 Twisting Functors

3.3.1 Definition of twisting functors

Now we proceed to state and prove a precise version of Theorem 1.1.2. Throughout this section we fix a simple root $\alpha \in \Delta$ and write $U := U(\mathfrak{g}_m)$ for the sake of brevity.

Recall that a right Ore set S in a non-commutative ring R is a multiplicatively closed subset such that for all $r \in R$ and $s \in S$ there exist $r' \in R$ and $s' \in S$ such that rs' = sr'.

A left Ore set is defined dually. In order for R to admit a right ring of quotients with respect to S it is necessary that S be a right Ore set. When R has no zero divisors this condition is also sufficient, and if S is both a right and left Ore set the left and right quotients are isomorphic; see [35, §2.1].

Also recall the notation $e_{\alpha,i}, h_{\alpha,i}, f_{\alpha,i}$ from Section 2.5. Let F_{α} be the multiplicative subset of U generated by $\{f_{\alpha,0}, \ldots, f_{\alpha,m}\}$ and note that these elements are pairwise commutative.

Lemma 3.3.1. F_{α} is both a left and right Ore set in U.

Proof. We show that F_{α} is a left Ore set; the proof that it is a right Ore set is almost identical. Let $x \in \mathfrak{g}_m \subseteq U$, fix $n = 0, \ldots, m$ and let $j \geq 0$. It is easily verified by induction that

$$f_{\alpha,n}^{j}x = \sum_{k=0}^{j} {j \choose k} (\operatorname{ad} f_{\alpha,n})^{k}(x) f_{\alpha,n}^{j-k}$$

But $\mathrm{ad}(f_{\alpha,n})$ is nilpotent, so there exists l such that $(\mathrm{ad}\,f_{\alpha,n})^l=0$. Hence for sufficiently large j, we have $f^j_{\alpha,n}x=u_jf^{j-l}_{\alpha,n}$ for some $u_j\in U$. Applying this repeatedly, we see that given a PBW monomial $v\in U$ and some $i\geq 0$, we can find $u\in U$ and $j\geq 0$ such that $f^j_{\alpha,n}v=uf^i_{\alpha,n}$. But U is spanned by such monomials v, so we have shown that $\{f^i_{\alpha,n}:i\geq 0\}$ is a left Ore set. Combining these statements for $n=0,\ldots,m$ it follows that F_α is a left Ore set since F_α is commutative.

Now let U_{α} be the localisation of U with respect to F_{α} . We wish to explicitly describe a basis for U_{α} . We introduce the following notation: Let

$$a:\Phi^+\times\{0,1,\ldots,m\}\to\mathbb{Z}_{\geq 0}$$

$$b: \Delta \times \{0, 1, \dots, m\} \to \mathbb{Z}_{\geq 0}$$

$$c: \Phi^+ \setminus \{\alpha\} \times \{0, 1, \dots, m\} \to \mathbb{Z}_{\geq 0}$$

be arbitrary maps of sets, and define

$$v(a,b,c) := \left(\prod_{i=0}^{n} \prod_{\beta \in \Phi^{+}} e_{\beta,i}^{a_{\beta,i}}\right) \left(\prod_{i=0}^{n} \prod_{\beta \in \Delta} h_{\beta,i}^{b_{\beta,i}}\right) \left(\prod_{i=0}^{n} \prod_{\beta \in \Phi^{+} \setminus \{\alpha\}} f_{\beta,i}^{c_{\beta,i}}\right) \in U$$

where we take the product with respect to some fixed choice of ordering on the basis of \mathfrak{g} . These elements can also be described as precisely the PBW monomials in U which have no factor in F_{α} .

Lemma 3.3.2. Let a, b, c, and v(a, b, c) be as above. Then

- (i) A basis for U_{α} is given by the elements $f_{\alpha,0}^{i_0} \cdots f_{\alpha,m}^{i_m} v(a,b,c)$ where $i_j \in \mathbb{Z}$.
- (ii) A basis for U_{α} is given by the elements $v(a,b,c)f_{\alpha,0}^{i_0}\cdots f_{\alpha,m}^{i_m}$ where $i_j\in\mathbb{Z}$.

Proof. Note that U_{α} is spanned by unordered monomials in \mathfrak{g}_m and the $f_{\alpha,i}^{-1}$. Using the left Ore condition we can rewrite any such monomial as a span of monomials of the form described in (i). Furthermore, if there is a linear dependence between the latter monomials, we can left multiply by appropriate elements of F_{α} to obtain a linear dependence between PBW monomials in U, which must be zero. This proves (i), and (ii) follows by a symmetrical argument.

We now give more precise relations between certain generators of U_{α} , which will be useful later.

Lemma 3.3.3. For any $i, j \in \mathbb{Z}$, the following relations hold in U_{α} , for any $h \in \mathfrak{h}$ and any $\beta \in \Phi^+ \setminus \{\alpha\}$:

(i)
$$[e_{\alpha,i}, f_{\alpha,j}^{-1}] = -f_{\alpha,j}^{-2} h_{\alpha,i+j} - 2f_{\alpha,j}^{-3} f_{\alpha,i+2j}$$
.

(ii)
$$[h_i, f_{\alpha,j}^{-1}] = \alpha(h) f_{\alpha,j}^{-2} f_{\alpha,i+j}$$
.

(iii)
$$[e_{\beta,i}, f_{\alpha,j}^{-1}] = r_1 f_{\alpha,j}^{-2} e_{\beta-\alpha,i+j} + r_2 f_{\alpha,j}^{-3} e_{\beta-2\alpha,i+2j} + r_3 f_{\alpha,j}^{-4} e_{\beta-3\alpha,i+3j}$$

for some $r_1, r_2, r_3 \in \mathbb{C}$. We adopt the convention that $e_{\gamma,i} = 0$ if $\gamma \notin \Phi$ or if i > m.

Proof. These relations can be verified by multiplying by powers of $f_{\alpha,j}$ to obtain an expression which holds in U. We give an explicit calculation for part (i); the other relations can be proved similarly (in part (iii) we require the general fact that $\beta - 4\alpha$ is never a root for any roots α and β). Using the relations in U we have:

$$f_{\alpha,j}^3 e_{\alpha,i} = f_{\alpha,j}^2 e_{\alpha,i} f_{\alpha,j} - f_{\alpha,j} h_{\alpha,i+j} f_{\alpha,j} - 2f_{\alpha,i+2j} f_{\alpha,j}.$$

We then multiply on the left by $f_{\alpha,j}^{-3}$ and on the right by $f_{\alpha,j}^{-1}$ to obtain (i).

Let $V_{\alpha} \subseteq U_{\alpha}$ be the span of the monomials appearing in Lemma 3.3.2(i) such that $i_j \geq 0$ for some $j = 0, \ldots, m$. We then have the following lemma:

Lemma 3.3.4. V_{α} is a *U-U-sub-bimodule of* U_{α} .

Proof. We first show that V_{α} is a right U-module. Let i_l, j_l be such that at least one $i_l \geq 0$ and all $j_l \geq 0$, and let v = v(a, b, c) and v' = v(a', b', c') for appropriate functions a, b, c and a', b', c'. Then by the PBW theorem we have $vf_{\alpha,0}^{j_0} \dots f_{\alpha_m}^{j_m} v' = \sum_p (f_{\alpha,0}^{k_{p_0}} \dots f_{\alpha_m}^{k_{p_m}} v_p)$ for some $k_{p_l} \geq 0$, and $v_p = v(a_p, b_p, c_p)$ for suitable functions a_p, b_p, c_p . Hence:

$$f_{\alpha,0}^{i_0} \dots f_{\alpha,m}^{i_m} v \cdot (f_{\alpha,0}^{j_0} \dots f_{\alpha,m}^{j_m} v') = \sum_p (f_{\alpha,0}^{i_0} \dots f_{\alpha,m}^{i_m} f_{\alpha,0}^{k_{p_0}} \dots f_{\alpha,m}^{k_{p_m}} v_p)$$

$$= \sum_{p} (f_{\alpha_0}^{i_0+k_{p_0}} \dots f_{\alpha,m}^{i_m+k_{p_m}} v_p)$$

In particular this expression lies in V_{α} . But U is spanned by elements of the form $f_{\alpha,0}^{j_0} \dots f_{\alpha_m}^{j_m} v'$, so V_{α} is a right U-module.

We now claim that V_{α} is a span of monomials of the type appearing in 3.3.2(ii) with $i_j \geq 0$ for some j; once we have this, it follows that V_{α} is a left U-module by a similar argument to above. In general, if $i_0, \ldots, i_m \in \mathbb{Z}$ and v = v(a, b, c) for appropriate a, b, c then using the fact that F_{α} is an Ore set, in U_{α} we have:

$$v f_{\alpha,0}^{i_0} \dots f_{\alpha,m}^{i_m} = \sum_{p} f_{\alpha,0}^{i_{p_0}} \dots f_{\alpha,m}^{i_{p_m}} v_p$$

where $i_{p_l} \geq 0$ and the v_p are of the same form as v. In fact, we see that if $i_j \geq 0$, then $i_{p_j} \geq 0$ for all p, so in this case $vf_{\alpha,0}^{i_0} \dots f_{\alpha,m}^{i_m} \in V_{\alpha}$. Hence

$$V_{\alpha} \supseteq \operatorname{span}\{v(a,b,c)f_{\alpha,0}^{i_0}\dots f_{\alpha,m}^{i_m} \mid \text{ at least one of the } i_j \geq 0\}.$$

The inclusion

$$V_{\alpha} \subseteq \operatorname{span}\{v(a,b,c)f_{\alpha,0}^{i_0}\dots f_{\alpha,m}^{i_m} \mid \text{ at least one of the } i_j \geq 0\}$$

follows by a similar argument, completing the proof.

We now consider the U-U-bimodule $S_{\alpha} := U_{\alpha}/V_{\alpha}$. First we observe the following consequence of Lemma 3.3.2.

Corollary 3.3.5. S_{α} has a basis given by

$$\{f_{\alpha,0}^{i_0}\dots f_{\alpha,m}^{i_m}v(a,b,c) \mid i_j < 0 \text{ for all } j\}$$

and another basis given by:

$$\{v(a,b,c)f_{\alpha,0}^{i_0}\dots f_{\alpha,m}^{i_m}\mid i_j<0 \text{ for all } j\}$$

We will abuse notation slightly and usually denote an element of U_{α} and its coset in S_{α} by the same symbol. At several points we will use the fact that in S_{α} the element $f_{\alpha,0}^{i_0} \dots f_{\alpha,m}^{i_m}$ is zero if any $i_j \geq 0$.

Now we pick a special automorphism of \mathfrak{g}_m . Just as in §2.3.3, the simple root α we fixed earlier gives rise to a reflection $s_{\alpha} \in W = N_G(\mathfrak{h})/\mathfrak{h}$. We lift s_{α} arbitrarily to an element of $N_G(\mathfrak{h})$, which gives an automorphism of \mathfrak{g} via the adjoint representation. We denote this automorphism by ϕ_{α} . Note that ϕ_{α} permutes the root spaces as s_{α} , i.e. it sends \mathfrak{g}_{β} to $\mathfrak{g}_{s_{\alpha}(\beta)}$ and preserves \mathfrak{h} . Furthermore, after rescaling e_{α} and f_{α} if necessary, we may assume that $\phi_{\alpha}(e_{\alpha}) = f_{\alpha}$ and $\phi_{\alpha}(f_{\alpha}) = e_{\alpha}$. We extend ϕ_{α} to an automorphism

of \mathfrak{g}_m using the rule $\phi_{\alpha}(xt^i) = \phi_{\alpha}(x)t^i$ for all $x \in \mathfrak{g}$; by abuse of notation we also denote this automorphism ϕ_{α} .

If M is a left U-module, we let $\phi_{\alpha}(M)$ denote the module obtained by twisting the left action on M by ϕ_{α} and write \cdot_{α} for this action. More precisely, if $v \in M$ and $u \in U$ then $u \cdot_{\alpha} v := \phi_{\alpha}(u) \cdot v$. Similarly, we can also twist the action on M by ϕ_{α}^{-1} , and we use the notation $\cdot_{\alpha^{-1}}$ for the twisted action in this case.

Now let \mathcal{C} be the full subcategory of U-mod whose objects are the \mathfrak{h} -semisimple modules.

Lemma 3.3.6. $\mathcal{O}(\mathfrak{g}_m)$ is a Serre subcategory of the category \mathcal{C} .

Proof. By Corollary 3.1.8, it suffices to show that for all $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$ and all $M_1, M_2 \in \mathcal{O}^{(\mu)}$, we have that $M_1, M_2 \in \mathcal{C}$ and any \mathfrak{h} -semisimple extension M between M_1 and M_2 lies in $\mathcal{O}^{(\mu)}$. By the definition of $\mathcal{O}^{(\mu)}$, if $M_i \in \mathcal{O}^{(\mu)}$ then M_i must be \mathfrak{h} -semisimple and hence lie in \mathcal{C} . On the other hand, to show $M \in \mathcal{O}^{(\mu)}$, we must show that:

- (i) M is finitely generated.
- (ii) For all $h \in \mathfrak{h}$ and $1 \leq i \leq m$, $h_i \mu_i(h)$ acts locally nilpotently on M (where we recall that we use h_i as shorthand for $h \otimes t^i$).
- (iii) The subalgebra $\mathfrak{h}_{(\geq 1)}$ acts locally finitely on M.
- (iv) The subalgebra $\mathfrak{n}_m \subseteq \mathfrak{g}_m$ acts locally nilpotently on M.

For (i), let X_1 be a finite generating set for M_1 and let X_2 be a finite generating set for M_2 . Let X_2' be a set containing a choice of preimage for each element of X_2 under the surjection $M \to M_2$. Then $X_1 \cup X_2'$ is a finite generating set for M. For (ii), let $v \in M$. Then for some $j \geq 0$, $(h_i - \mu_i(h))^j \cdot (v + M_1) = 0 \in M/M_1 \cong M_2$, so $(h_i - \mu_i(h))^j \cdot v =: v' \in M_1$, and for some $k \geq 0$, $(h_i - \mu_i(h))^k \cdot v' = 0$, so we have $(h_i - \mu_i(h))^{j+k} \cdot v = 0$. Finally, we have that $\dim(M^{\lambda}) = \dim(M_1^{\lambda}) + \dim(M_2^{\lambda})$ for any $\lambda \in \mathfrak{h}^*$, so in particular the weight spaces of M are finite-dimensional and bounded above. Hence (iii) and (iv) hold.

Let $\mathcal{H}: U\operatorname{-mod} \to \mathcal{C}$ be the functor given by letting $\mathcal{H}(M)$ be the sum of the \mathfrak{h} -weight spaces in M, and then define two endofunctors of \mathcal{C} by setting:

$$T_{\alpha}M := \phi_{\alpha}(S_{\alpha} \otimes_{U} M)$$

$$G_{\alpha}M := \mathcal{H}(\operatorname{Hom}_{U}(S_{\alpha}, \phi_{\alpha}^{-1}(M))).$$

Note that the action of U on $G_{\alpha}M$ is given by

$$(u \cdot f)(s) = f(s \cdot u)$$
 for $u \in U$, $f \in G_{\alpha}M$, $s \in S_{\alpha}$.

Also note that if $M, N \in \mathcal{C}$ and $\chi \in \operatorname{Hom}_{\mathcal{C}}(M, N)$, then $T_{\alpha}(\chi) : T_{\alpha}M \to T_{\alpha}N$ and $G_{\alpha}(\chi) : G_{\alpha}M \to G_{\alpha}N$ are given by:

$$T_{\alpha}(\chi)(s \otimes m) = s \otimes \chi(m) \text{ for } s \in S_{\alpha}, \ m \in M$$

$$G_{\alpha}(\chi)(\rho) = \chi \circ \rho \text{ for } \rho \in \operatorname{Hom}_{U}(S_{\alpha}, \phi_{\alpha}^{-1}(M)))$$

In order to see these functors are well defined, the only non-trivial condition to check is that $T_{\alpha}M \in \mathcal{C}$ for any $M \in \mathcal{C}$. For $M \in \mathcal{C}$, we have that $T_{\alpha}M$ is spanned by $\{f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes w : i_j > 0, w \in M\}$, and so $T_{\alpha}M \in \mathcal{C}$ will follow from:

Lemma 3.3.7. For any $w \in M^{\lambda}$, we have

$$f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m}\otimes w\in (T_{\alpha}M)^{s_{\alpha}(\lambda)-(i_0+\cdots+i_m)\alpha}$$

Proof. For $w \in M^{\lambda}$ and $h \in \mathfrak{h}$ we have

$$h \cdot_{\alpha} (f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes w) = (s_{\alpha}(h) f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) \otimes w$$

$$= f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} (s_{\alpha}(h) - (i_0 + \dots + i_m) \alpha(h)) \otimes w$$

$$= f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes (s_{\alpha}(h) - (i_0 + \dots + i_m) \alpha(h)) w$$

$$= f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes (s_{\alpha}(\lambda) - (i_0 + \dots + i_m) \alpha)(h) w$$

$$= (s_{\alpha}(\lambda) - (i_0 + \dots + i_m) \alpha)(h) (f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes w).$$

and the desired result follows.

The following result, characterising weight elements of $G_{\alpha}M$, will be very useful later.

Lemma 3.3.8. Let $g \in \operatorname{Hom}_U(S_\alpha, \phi_\alpha^{-1}(M))$. Then g has weight $\lambda \in \mathfrak{h}^*$ if and only if $g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$ has weight $\lambda + (i_0 + \dots + i_m)\alpha$ in $\phi_\alpha^{-1}(M)$ for all $i_0, \dots, i_m \in \mathbb{Z}_{\geq 0}$, i.e. if and only if this vector has weight $s_\alpha(\lambda) - (i_0 + \dots + i_m)\alpha$ in M.

Proof. First assume that g has weight λ . Then for any $h \in \mathfrak{h}$ and $i_j \geq 0$:

$$\lambda(h)g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) = (h \cdot g)(f_{\alpha_0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$$

$$= g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}h)$$

$$= g((h - (i_0 + \dots + i_m)\alpha(h))f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$$

$$= (h - (i_0 + \dots + i_m)\alpha(h)) \cdot g(f_{\alpha,0}^{-i_0} f_{\alpha,m}^{-i_m})$$

so $g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$ has weight $\lambda + (i_0 + \dots + i_m)\alpha$ in $\phi_{\alpha}^{-1}(M)$, and hence has weight $s_{\alpha}(\lambda) - (i_0 + \dots + i_m)\alpha$ in M.

On the other hand, suppose $g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$ has weight $s_{\alpha}(\lambda) - (i_0 + \dots + i_m)\alpha$ in M for all $i_j \geq 0$. Then for any $h \in \mathfrak{h}, i_j \geq 0$, u = v(a,b,c) for functions a,b,c of the

appropriate form we have:

$$\begin{split} (h \cdot g)(uf_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) &= g(uf_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} h) \\ &= u \cdot g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} h) \\ &= u \cdot g((h - (i_0 + \dots + i_m)\alpha(h))f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) \\ &= u \cdot (h - (i_0 + \dots + i_m)\alpha(h)) \cdot g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) \\ &= u \cdot \lambda(h)g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) \\ &= \lambda(h)g(uf_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) \end{split}$$

and hence g has weight λ .

We then obtain the following corollary:

Corollary 3.3.9. Let $g \in \text{Hom}_U(S_\alpha, \phi_\alpha^{-1}(M))$, Then g is a weight vector if and only if $g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$ is a weight vector for all $i_j \geq 0$. In particular, any element of $G_\alpha(M)$ is the direct sum of such weight vectors.

Proof. Suppose $g \in \operatorname{Hom}_U(S_{\alpha}, \phi_{\alpha}^{-1}(M))$ is such that $g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$ is a weight vector for all $i_j \geq 0$. Since $e_{\alpha,k} \cdot_{\alpha^{-1}} g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) = g(f_{\alpha_0}^{-i_0} \dots f_{\alpha,k}^{-i_k+1} \dots f_{\alpha,m}^{-i_m})$, in $\phi_{\alpha}^{-1}(M)$ we have that $\operatorname{wt}(g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})) + \alpha = \operatorname{wt}(g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,k}^{-i_k+1} \dots f_{\alpha,m}^{-i_m}))$ for all $0 \leq k \leq m$. Hence if in $\phi_{\alpha}^{-1}(M)$ the weight of $g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$ is λ' , we have that the weight of $g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$ is $\lambda' - 2\alpha + (i_0 + \dots + i_m)\alpha$. Applying Lemma 3.3.8, we see that g is indeed a weight vector.

The following easy lemma is the first step in our ultimate goal of showing that the functors T_{α} and G_{α} are equivalences between appropriate Jordan blocks of \mathcal{O} .

Lemma 3.3.10. T_{α} is right exact and G_{α} is left exact.

Proof. For any module $M \in U$ -mod, we have

$$\phi_{\alpha}(S_{\alpha} \otimes_{U} M) \cong \phi_{\alpha}(S_{\alpha}) \otimes_{U} M$$

and since $\phi_{\alpha}(S_{\alpha}) \otimes_{U} (\bullet)$ is right exact, T_{α} is also right exact.

Similarly, G_{α} is a composition of two left exact functors $\operatorname{Hom}_{U}(S_{\alpha}, \phi_{\alpha}^{-1}(\bullet))$ and \mathcal{H} and so is left exact.

3.3.2 Twisting functors between blocks of category \mathcal{O}

We now observe that the Weyl group action on \mathfrak{h}^* extends naturally to an action on $(\mathfrak{h}_m^{\geq 1})^*$ by letting W act diagonally through the identification $(\mathfrak{h}_m^{\geq 1})^* = (\mathfrak{h}^*)^m$, and this vector space parameterises the Jordan blocks of $\mathcal{O}(\mathfrak{g}_m)$. Retaining the notation of the previous section, we can now state a precise version of Theorem 1.1.2.

Theorem 3.3.11. Let $\mu \in (\mathfrak{h}_m^{\geq 1})^*$ be such that $\mu(h_{\alpha,m}) \neq 0$. The functors T_{α} and G_{α} restrict to functors

$$T_{\alpha} : \mathcal{O}^{(\mu)}(\mathfrak{g}_m) \longrightarrow \mathcal{O}^{(s_{\alpha}(\mu))}(\mathfrak{g}_m)$$

$$G_{\alpha} : \mathcal{O}^{(s_{\alpha}(\mu))}(\mathfrak{g}_m) \longrightarrow \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$$

which form a quasi-inverse pair of equivalences.

The proof of Theorem 3.3.11 is broken down into a series of lemmas which we record and prove over the course of this section. The theorem will follow directly from combining Lemmas 3.3.12, 3.3.19, 3.3.22, 3.3.23 and 3.3.24.

For the rest of the section we keep $\alpha \in \Delta$ fixed and let $\mu \in (\mathfrak{h}_m^{\geq 1})^*$ be such that $\mu(h_{\alpha,m}) \neq 0$. The following result is the first step in the proof of Theorem 3.3.11.

Lemma 3.3.12. T_{α} restricts to a functor $\mathcal{O}^{(\mu)}(\mathfrak{g}_m) \to \mathcal{O}^{(s_{\alpha}(\mu))}(\mathfrak{g}_m)$.

Proof. For $M \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ let l(M) denote the minimal length of a filtration $0 = M_0 \subseteq M_1 \subseteq \cdots \subseteq M_{k-1} \subseteq M_k = M$ such that the sections M_i/M_{i-1} are all highest weight modules; such a filtration exists by Lemma 3.1.5. We have an exact sequence:

$$0 \to M_1 \to M \to M/M_1 \to 0$$

and since T_{α} is right exact, we have another exact sequence

$$T_{\alpha}M_1 \to T_{\alpha}M \to T_{\alpha}(M/M_1) \to 0$$

Using Lemma 3.3.6 and the fact that $T_{\alpha}M_1$ is a highest weight module, we can reduce the claim that $T_{\alpha}(M) \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ to the case where l(M) = 1, i.e. M is a highest weight module.

Since $T_{\alpha}M$ is \mathfrak{h} -semisimple and a quotient of a highest weight module is still highest weight, by induction and Lemma 3.3.6 it suffices to show that if M is highest weight of weight $(\lambda, \mu_1, \ldots, \mu_m)$, then $T_{\alpha}M$ is highest weight of weight $(\lambda', s_{\alpha}(\mu_1), \ldots, s_{\alpha}(\mu_m))$ for some $\lambda' \in \mathfrak{h}^*$.

For the rest of the proof, fix M highest weight of weight $\lambda \in \mathfrak{h}_m^*$, and let $v \in M$ be a highest weight generator of M. We aim to show that $f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1} \otimes v$ is highest weight and generates $T_{\alpha}M$, which will complete the proof of the lemma.

To see that $f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1} \otimes v$ is maximal, we use Lemma 3.3.7 to see that for any $i \geq 0$, the element $e_{\alpha,i}f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1} \otimes v \in T_{\alpha}M$ lies in an \mathfrak{h} -eigenspace whose weight is not a weight of $T_{\alpha}M$. To see that it is a genuine highest weight vector, one can use Lemma 3.3.3(ii) to show that h_i acts via $s_{\alpha_i}(h_i)$.

To see that $f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1} \otimes v$ generates $T_{\alpha}M$ we use the fact that Uv = M and the two bases of S_{α} from Corollary 3.3.5, to check that every element of $T_{\alpha}M$ lies in the submodule generated by the set $\{f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes v \mid i_k > 0\}$. Let L denote the span of this set. Note that it is an $(\mathfrak{sl}_2)_m$ -module, where $(\mathfrak{sl}_2)_m$ is the truncated current Lie algebra on $(\mathfrak{sl}_2)_{\alpha} := \langle e_{\alpha}, h_{\alpha}, f_{\alpha} \rangle$. To complete the proof we show that L is a simple $(\mathfrak{sl}_2)_m$ -module.

Let $\mathfrak{t}_m \subseteq (\mathfrak{sl}_2)_m$ denote the span of $h_{\alpha,0},\ldots,h_{\alpha,m}$ and let $\gamma=\lambda|_{\mathfrak{t}_m}\in\mathfrak{t}_m^*$ give the action of \mathfrak{t}_m on $f_{\alpha,0}^{-1}\ldots f_{\alpha,m}^{-1}\otimes v$. If $M(\gamma)$ denotes the Verma module of highest weight γ then there is a nonzero homomorphism $M(\gamma)\to L$ and the dimensions of the weight spaces are the same. Therefore it remains to show that the Verma module $M(\gamma)$ is simple. By Lemma 3.3.3(ii) we see that $\gamma(h_{\alpha,m})=(s_\alpha\lambda)(h_{\alpha,m})\neq 0$ and so we can apply Theorem 3.2.1 to see

that $\mathcal{O}^{(\gamma_{\geq 1})}((\mathfrak{sl}_2)_m)$ is equivalent to $\mathcal{O}^{(\gamma_{\geq 1})}(\mathfrak{t}_m)$. In the latter category, all highest weight modules are simple, and so it follows that $M(\gamma)$ is simple as required.

Our next steps towards proving Theorem 3.3.11 are to prove the following results on the invariants of a module $M \in \mathcal{O}^{(\mu)}$ with respect to a certain subalgebra of \mathfrak{g} , and then prove a result on elements of the tensor product $S_{\alpha} \otimes_{U} M$. Throughout, we let $(\mathfrak{sl}_{2})_{\alpha} = \langle e_{\alpha}, h_{\alpha}, f_{\alpha} \rangle \subseteq \mathfrak{g}$.

Lemma 3.3.13. Let $\mu \in (\mathfrak{h}_m^{\geq 1})^*$ be such that $\mu_m(h_\alpha) \neq 0$, let $M \in \mathcal{O}^{(\mu)}$, and let $\mathfrak{a} = \langle f_{\alpha,0}, \ldots, f_{\alpha,m} \rangle \subseteq \mathfrak{g}_m$. Then M is free as a $U(\mathfrak{a})$ -module, and any basis of $M^{\langle e_{\alpha,0}, \ldots, e_{\alpha,m} \rangle}$ as a vector space is a free generating set for M.

Proof. It suffices to consider only the case where M is indecomposable. First consider the case where $\mathfrak{g} = \mathfrak{sl}_2$ and M is a Verma module. In this case, $M^{\langle e_{\alpha,0},\dots,e_{\alpha,m}\rangle}$ is one-dimensional, spanned by any highest weight vector of M. But by definition M is generated freely by this as a $U(\mathfrak{n}_m^-)$ -module, and in this case $\mathfrak{a} = \mathfrak{n}_m^-$ so the lemma holds.

If $\mathfrak{g} = \mathfrak{sl}_2$ and M is indecomposable but not a Verma module, then by Lemma 3.1.5 and [56, Theorem 7.1] M has a filtration $0 = M_0 \subseteq M_1 \subseteq \cdots \subseteq M_{k-1} \subseteq M_k = M$ such that each quotient is a Verma module. To show any basis of $M^{\langle e_{\alpha,0},\dots,e_{\alpha,m}\rangle}$ freely generates M as a $U(\mathfrak{a})$ -module, it suffices to check one choice.

Choose $\Omega = \Psi \cup \{v\}$, where Ψ is a basis of $M_{k-1}^{\langle e_{\alpha,0},\dots,e_{\alpha,m}\rangle}$ and $v \in M^{\langle e_{\alpha,0},\dots,e_{\alpha,m}\rangle} \setminus M_{k-1}$. Let $M' \subseteq M$ be the $U(\mathfrak{a})$ -submodule generated by Ω . Then observe that M/M_{k-1} is a Verma module with highest weight generator $v + M_{k-1}$. Hence for any $w \in M$, there is some $w' \in M'$ such that $w - w' \in M_{k-1}$. But $\Psi \subseteq \Omega$, and by induction Ψ generates M_{k-1} as a $U(\mathfrak{a})$ -module, so $M_{k-1} \subseteq M'$. Hence $w - w' \in M'$, and so $w \in M'$ for any $w \in M$, i.e. M = M'.

Now we show that Ω generates M freely, i.e. that the set $\Omega' = \{f_{\alpha,0}^{i_0} \dots f_{\alpha,m}^{i_m} v : i_0, \dots, i_m \geq 0, v \in \Omega\}$ is linearly independent. We already know that Ω' spans M, so it is enough to show that the number of $f_{\alpha,0}^{i_0} \dots f_{\alpha,m}^{i_m} v$ of weight λ' is equal to the dimension of $M^{\lambda'}$ for any $\lambda' \in \mathfrak{h}^*$. Now, every quotient M_i/M_{i-1} is isomorphic to the same Verma

module $M(\lambda, \mu)$ and hence all $v \in \Omega$ have weight λ and also $\dim(M^{\lambda'}) = k \dim(M(\lambda, \mu)^{\lambda'})$. These two facts together imply that the number of $f_{\alpha,0}^{i_0} \dots f_{\alpha,m}^{i_m} v$ of weight λ' is equal to the dimension of $M^{\lambda'}$, so Ω generates M freely and the lemma holds in the case $\mathfrak{g} = \mathfrak{sl}_2$.

Finally, we deal with the case where \mathfrak{g} is any reductive Lie algebra. Let Υ be a basis of $M^{\langle e_{\alpha,0},\dots,e_{\alpha,m}\rangle}$. Let $v\in M$ and let N be the $((\mathfrak{sl}_2)_{\alpha})_m$ -submodule of M generated by v. It is easy to check that $N\in\mathcal{O}^{(\mu_1(h_{\alpha}),\dots,\mu_m(h_{\alpha}))}((\mathfrak{sl}_2)_{\alpha})$, since all the axioms except finite generation follow from the fact $M\in\mathcal{O}^{\mu}$, while by definition N is generated by one element. Hence we can apply the lemma in the \mathfrak{sl}_2 case to see that $v=\sum_j f_{\alpha,0}^{i_{j,0}}\dots f_{\alpha,m}^{i_{j,m}}v_k$, for some $i_{j,k}\geq 0$ and $v_k\in N^{\langle e_{\alpha,0},\dots,e_{\alpha,m}\rangle}\subseteq M^{\langle e_{\alpha,0},\dots,e_{\alpha,m}\rangle}=\operatorname{span}\Upsilon$, so v is in the $U(\mathfrak{a})$ -module generated by Υ .

Suppose Υ does not generate M freely as a $U(\mathfrak{a})$ -module. Then $\sum_j f_{\alpha,0}^{i_j,0} \dots f_{\alpha,m}^{i_j,m} b_j = 0$ for some $i_{j,k} \geq 0$ and $b_j \in \Upsilon$. But since this sum is finite, we can let N be the $((\mathfrak{sl}_2)_{\alpha})_m$ -module generated by $\{b_1, \dots, b_k\}$ and obtain a contradiction to the \mathfrak{sl}_2 case. Hence the lemma holds.

Corollary 3.3.14. Again, let $\mu \in (\mathfrak{h}_m^{\geq 1})^*$ be such that $\mu_m(h_\alpha) \neq 0$, let $M \in \mathcal{O}^{(\mu)}$, and let N_1 be a finitely generated $((\mathfrak{sl}_2)_\alpha)_m$ -submodule of M. Then there exists a $U(\mathfrak{a})$ -submodule N_2 of M such that $M = N_1 \oplus N_2$, where again $\mathfrak{a} = \langle f_{\alpha,0}, \ldots, f_{\alpha,m} \rangle$.

Proof. Let Ψ be a basis for $N_1^{\langle e_{\alpha,0},\dots,e_{\alpha,m}\rangle}$, which freely generates N_1 as a $U(\mathfrak{a})$ -module by the previous lemma. Since $N_1^{\langle e_{\alpha,0},\dots e_{\alpha,m}\rangle} \subseteq M^{\langle e_{\alpha,0},\dots,e_{\alpha,m}\rangle}$, we can extend Ψ to a basis Ω for $M^{\langle e_{\alpha,0},\dots,e_{\alpha,m}\rangle}$, so if we let N_2 be the $U(\mathfrak{a})$ -module generated by $\Omega \setminus \Psi$ then by the previous lemma we have $M = N_1 \oplus N_2$.

We now recall the following general result about tensor products. Let R be a ring, and let M and N be right and left R-modules respectively, and let V be a \mathbb{C} -vector space. We say a \mathbb{C} -bilinear map $\varphi: M \times N \to V$ is R-balanced if for any $v \in M$, $w \in N$, and $r \in R$, we have $\varphi(v \cdot r, w) = \varphi(v, r \cdot w)$. We then have the following standard result on the tensor product $M \otimes_R N$:

Lemma 3.3.15. The element $v \otimes w$ of $M \otimes_R N$ is zero if and only if for any vector space V and R-balanced map $\varphi : M \times N \to V$, we have that $\varphi(v, w) = 0$.

We observe that if we set $\mathfrak{a} = \langle f_{\alpha,0}, \dots, f_{\alpha,m} \rangle$ as above and set $A = \operatorname{span}\{f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}: i_0, \dots, i_m > 0\} \subseteq S_{\alpha}$, which is a $U(\mathfrak{a})$ -sub-bimodule of S_{α} , then we have (by considering the two bases of S_{α} from Corollary 3.3.5) that as U- $U(\mathfrak{a})$ -bimodules:

$$S_{\alpha} \cong U \otimes_{U(\mathfrak{a})} A$$
,

and as $U(\mathfrak{a})$ -U-bimodules:

$$S_{\alpha} \cong A \otimes_{U(\mathfrak{a})} U$$
.

In particular, for any left U-module M, we have that as left $U(\mathfrak{a})$ -modules:

$$S_{\alpha} \otimes_{U} M \cong (A \otimes_{U(\mathfrak{a})} U) \otimes_{U} M \cong A \otimes_{U(\mathfrak{a})} M.$$

The following lemma, which we will prove using Corollary 3.3.14, Lemma 3.3.15 and the above observation, we be very useful in next part of the proof of Theorem 3.3.11:

Lemma 3.3.16. Suppose $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$ is such that $\mu_m(h_\alpha) \neq 0$ and let $M \in \mathcal{O}^{(\mu)}$. Let $v \in M^\lambda \setminus \{0\}$ for some $\lambda \in \mathfrak{h}^*$. Then the following are equivalent:

- (a) In $T_{\alpha}M$, we have $f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes v = 0$.
- (b) For any vector space V and $U(\mathfrak{a})$ -balanced map $\varphi: A \times M \to V$ we have that $\varphi(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, v) = 0.$
- (c) There exist $v_0, \ldots v_m \in M$ such that $v = f_{\alpha,0}^{i_0} \cdot v_0 + \cdots + f_{\alpha,m}^{i_m} \cdot v_m$.

Proof. We first note that (a) and (b) are equivalent by the above observation and Lemma 3.3.15. Also, if $m = f_{\alpha,0}^{i_0} \cdot v_0 + \dots + f_{\alpha,m}^{i_m} \cdot v_m$ for some $v_0, \dots, v_m \in M$, then $f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes v = f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes f_{\alpha}^{i_0} \cdot v_0 + \dots + f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes f_{\alpha,m}^{i_m} \cdot v_m = f_{\alpha,0}^0 \dots f_{\alpha,m}^{-i_m} \otimes v_0 + \dots + f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^0 \otimes v_0 + \dots + f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_0} \otimes v_0 + \dots + f_{\alpha,0}^{-i_0} \otimes v_0 + \dots +$

we will use later). To show (b) implies (c), suppose $v \in M$ cannot be written in the form $f_{\alpha,0}^{i_0} \cdot v_0 + \cdots + f_{\alpha,m}^{i_m} \cdot v_m$. We seek a $U(\mathfrak{a})$ -balanced map $\varphi : A \times M \to V$ such that $\varphi(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, v) \neq 0$. It suffices to consider the case where M is indecomposable.

First consider the case where $\mathfrak{g} = \mathfrak{sl}_2$ and M is the Verma module $M(\lambda, \mu_1, \ldots, \mu_m)$ where $\mu_m \neq 0$. Let $v \in M$. Then $v = \sum k_{a_0, \ldots, a_m} f_{\alpha, 0}^{a_0} \ldots f_{\alpha, m}^{a_m} \otimes 1$, where all but finitely many of the k_{a_0, \ldots, a_m} are zero. Observe that v is of the form $v = f_{\alpha, 0}^{i_0} \cdot v_0 + \cdots + f_{\alpha, m}^{i_m} \cdot v_m$ for some $v_0, \ldots, v_m \in M$ if and only if $k_{a_0, \ldots, a_m} = 0$ for all $a_k < i_k$.

Now suppose v is not of the form $v = f_{\alpha,0}^{i_0} \cdot v_0 + \cdots + f_{\alpha,m}^{i_m} \cdot v_m$. Then pick some $0 < a_k \le i_k$ such that $k_{i_0-a_0,\dots,i_m-a_m} \ne 0$. We can define a $U(\mathfrak{a})$ -balanced map $\varphi : A \times M \to \mathbb{C}$ by setting:

$$\varphi(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, f_{\alpha,0}^{j_0} \dots f_{\alpha,m}^{j_m} \otimes 1) = \begin{cases} 1 & \text{if } i_k - j_k = a_k \text{ for all } 0 \le k \le m \\ 0 & \text{otherwise} \end{cases}$$

and extending bilinearly. By construction we have that φ is a $U(\mathfrak{a})$ -balanced map such that $\varphi(f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m},v)\neq 0$.

Now we consider the case where $\mathfrak{g} = \mathfrak{sl}_2$ and M is indecomposable. If M is a Verma module, we are done by the above. If not, let $0 = M_0 \subseteq M_1 \subseteq \cdots \subseteq M_{k-1} \subseteq M_k = M$ be a filtration of M such that each section is a highest weight module. In fact, by [56, Theorem 7.1], in this case each Verma module is simple, so each section is in fact a Verma module. Let $v \in M$ be such that v cannot be written as $f_{\alpha,0}^{i_0} \cdot v_0 + \cdots + f_{\alpha,m}^{i_m} \cdot v_m$. Consider the quotient M/M_1 . We have one of two cases:

- (1) If $v+M_1$ cannot be written as $f_{\alpha,0}^{i_0} \cdot v_0 + \cdots + f_{\alpha,m}^{i_m} \cdot v_m + M_1$, then by induction on k we can find a $U(\mathfrak{a})$ -balanced map $\varphi: A \times (M/M_1) \to \mathbb{C}$ such that $\varphi(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, v + M_1) \neq 0$. We can then lift this to a $U(\mathfrak{a})$ -balanced map $\overline{\varphi}: A \times M \to \mathbb{C}$ by setting $\overline{\varphi}(u,v') = \varphi(u,v'+M_1)$, which clearly satisfies $\overline{\varphi}(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, v) \neq 0$.
- (2) If $v + M_1 = f_{\alpha,0}^{i_0} \cdot v_0' + \dots + f_{\alpha,m}^{i_m} \cdot v_m' + M_1$, then let $w = v f_{\alpha,0}^{i_0} \cdot v_0' \dots f_{\alpha,m}^{i_m} \cdot v_m' \in M_1$. Now w cannot be written as $f_{\alpha,0}^{i_0} \cdot w_0 + \dots + f_{\alpha,m}^{i_m} \cdot w_m$ or else we could take $v_0 = v_0' + w_0$

and obtain a contradiction. But M_1 is a Verma module, so we have already shown there exists a $U(\mathfrak{a})$ -balanced map $\varphi: A \times M_1 \to \mathbb{C}$ such that $\varphi(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, w) \neq 0$. Now by Corollary 3.3.14, there exists a $U(\mathfrak{a})$ -submodule M' of M such that M = M'. Hence

$$(\varphi \oplus 0) : A \times M \cong A \times (M_1 \oplus M') \cong (A \times M_1) \oplus (A \times M') \to \mathbb{C}$$

is a $U(\mathfrak{a})$ -balanced map such that:

$$(\varphi \oplus 0)(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, v) = (\varphi \oplus 0)(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, w + f_{\alpha,0}^{i_0} \cdot v_0' + \dots + f_{\alpha,m}^{i_m} \cdot v_m')$$
$$= \varphi(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, w) \neq 0$$

Finally, let \mathfrak{g} be an arbitrary reductive Lie algebra, let $M \in \mathcal{O}^{(\mu)}$, and let $v \in M$ be such that v cannot be written as $v = f_{\alpha,0}^{i_0} \cdot v_0 + \cdots + f_{\alpha,m}^{i_m} \cdot v_m$. Let M_1 be the $((\mathfrak{sl}_2)_{\alpha})_m$ -module generated by v, and let M_2 be a $U(\mathfrak{a})$ -module such that $M = M_1 \oplus M_2$ as in Corollary 3.3.14. By the \mathfrak{sl}_2 case, we have a $U(\mathfrak{a})$ -balanced map $\varphi : A \times M_1 \to \mathbb{C}$ such that $\varphi(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, v) \neq 0$. Then

$$(\varphi \oplus 0) : A \times M \cong A \times (N_1 \oplus N_2) \cong (A \times N_1) \oplus (A \times N_2) \to \mathbb{C}$$

is a $U(\mathfrak{a})$ -balanced map such that $(\varphi \oplus 0)(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, v) = \varphi(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}, v) \neq 0$ as required.

In the applications for which we need this lemma, the following corollary will often be sufficient:

Corollary 3.3.17. Retain the setup of the previous lemma and let $\lambda \in \mathfrak{h}^*$. Then there is some $k \in \mathbb{N}$ such that $M^{\lambda+k\alpha} = 0$. Furthermore for any such k and any $v \in M^{\lambda}$, we have that $f_{\alpha,0}^{-k} \dots f_{\alpha,m}^{-k} \otimes v = 0$ if and only if v = 0.

Proof. The existence of $k \in \mathbb{N}$ such that $M^{\lambda+k\alpha} = 0$ follows from Lemma 3.1.3. By

Lemma 3.3.16, we have that $f_{\alpha,0}^{-k} \dots f_{\alpha,m}^{-k} \otimes v = 0$ if and only if there exist $v_0, \dots, v_m \in M$ such that $v = f_{\alpha,0}^k \cdot v_0 + \dots + f_{\alpha,m}^k \cdot v_m$. Replacing v_0, \dots, v_m with their projections onto the $\lambda + k\alpha$ weight space if necessary, we may assume $v_0, \dots, v_k \in M^{\lambda + k\alpha} = 0$, so v = 0. \square

Finally, we need one more lemma:

Lemma 3.3.18. Let $M \in U$ -mod. Then any element of $S_{\alpha} \otimes_{U(\mathfrak{g}_m)} M$ can be written in the form $f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes v$, for some $i_0,\dots,i_m>0$ and $v \in M$.

Proof. Certainly any element of $S_{\alpha} \otimes M$ can be written as a sum of elements of the form $s \otimes m$ for some $s \in S_{\alpha}$ and $m \in M$. We have that any $s \in S_{\alpha}$ is the sum of elements of the form $f_{\alpha,0}^{-i,0} \dots f_{\alpha,m}^{-i_m} u$, where u = v(a,b,c), and $f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} u \otimes v = f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes (u \cdot v)$ for any $v \in M$, so any element of $S_{\alpha} \otimes M$ can be written as the sum of elements of the form $f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes v$.

Now let $x \in S_{\alpha} \otimes M$, and write $x = \sum_{a} f_{\alpha,0}^{-i_{0,a}} \dots f_{\alpha,m}^{-l_{m,a}} \otimes v_{a}$. Let $j_{k} = \max_{a} i_{k,a}$. Then:

$$x = \sum_{a} (f_{\alpha,0}^{-i_{0,a}} \dots f_{\alpha,m}^{-i_{m,a}} \otimes v_{a}) = \sum_{a} (f_{\alpha,0}^{-j_{0}} \dots f_{\alpha,m}^{-j_{m}} \otimes f_{\alpha,0}^{j_{0}-k_{0,a}} \dots f_{\alpha,m}^{j_{m}-i_{m,a}} v_{a})$$
$$= f_{\alpha,0}^{-j_{0}} \dots f_{\alpha,m}^{-j_{m}} \otimes \sum_{a} (f_{\alpha}^{j_{0}-i_{0,a}} \dots f_{\alpha}^{j_{m}-i_{m,a}} v_{a})$$

We can now complete the next step in the proof of Theorem 3.3.11, which is to show that there is an isomorphism between M and $G_{\alpha}T_{\alpha}M$.

Lemma 3.3.19. For any $M \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$, the map $\psi_M : M \to G_\alpha T_\alpha M$ defined by $\psi_M(v)(s) = s \otimes v$ is an isomorphism.

Proof. First observe that ψ_M is certainly a homomorphism since for any $v \in M$, $u \in U$ and $s \in S_\alpha$, we have $(u \cdot \psi_M(v))(s) = \psi_M(v)(s \cdot u) = (s \cdot u) \otimes v = s \otimes (u \cdot v) = \psi_M(u \cdot v)(s)$.

To see that ψ_M is always injective, let $M \in \mathcal{O}^{(\mu)}$ and let $v \in M$ be a non-zero weight vector of weight λ say. Then, applying Corollary 3.3.17, there exists $k \in \mathbb{N}$ such that

 $M^{\lambda+k\alpha}=0$, and $\psi_M(v)(f_{\alpha,0}^{-k}\dots f_{\alpha,m}^{-k})=f_{\alpha,0}^{-k}\dots f_{\alpha,m}^{-k}\otimes v=0$ if and only if v=0. Hence $\psi_M(v)=0$ if and only if v=0. Since M and $G_\alpha T_\alpha M$ are both \mathfrak{h} -semisimple, this suffices to show that ψ_M is injective.

We now show that ψ_M is surjective. Let $g \in G_{\alpha}T_{\alpha}M$. This g is a map $S_{\alpha} \to S_{\alpha} \otimes M$, and we may assume g is a weight element of $G_{\alpha}T_{\alpha}M$, of weight λ say. We first use Lemma 3.3.18 to see that, for all $i_0, \ldots, i_m > 0$ we can find $j_{i_0, \ldots, i_m, l} > 0$, and $v_{i_0, \ldots, i_m} \in M$ such that $g(f_{\alpha,0}^{-i_0} \ldots f_{\alpha,m}^{-i_m}) = f_{\alpha,0}^{-j_{i_0, \ldots, i_m, 0}} \ldots f_{\alpha,m}^{-j_{i_0, \ldots, i_m, m}} \otimes v_{i_0, \ldots, i_m}$.

If $i_l > j_{i_0,\dots,i_m,l}$, then we can replace m_{i_0,\dots,i_m} with $f_{\alpha}^{i_l-j_{i_0,\dots,i_m,l}} \cdot m_{i_0,\dots,i_m}$ and set $j_{i_0,\dots,i_m} = i_l$. Hence we may assume that $i_l \leq j_{i_0,\dots,i_m,l}$. On the other hand, suppose $i_l < j_{i_0,\dots,i_m,l}$. Then since g is a $U(\mathfrak{g}_m)$ -homomorphism, we have:

$$0 = f_{\alpha,l}^{i_l} \cdot g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) = f_{\alpha,0}^{-j_{i_0,\dots,i_m,0}} \dots f_{\alpha,l}^{-j_{i_0,\dots,i_m,l}+i_l} \dots f_{\alpha,m}^{-j_{i_0,\dots,i_m,m}} \otimes v_{i_0,\dots,i_m}$$

By Lemma 3.3.16 we have that $v_{i_0,...,i_m} = f_{\alpha,0}^{j_{i_0,...,i_m,0}} \cdot v_0' + \cdots + f_{\alpha,l}^{j_{i_0,...,i_m,l-i_l}} \cdot v_l' + \cdots + f_{\alpha,m}^{j_{i_0,...,i_m,m}} \cdot v_m'$ for some $v_0', \ldots, v_m' \in M$. Hence:

$$g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) = f_{\alpha,0}^{-j_{i_0,\dots,i_m,0}} \dots f_{\alpha,m}^{-j_{i_0,\dots,i_m,m}} \otimes$$

$$(f_{\alpha,0}^{j_{i_0,\dots,i_m,0}} \cdot v'_0 + \dots + f_{\alpha,l}^{j_{i_0,\dots,i_m,l-i_l}} \cdot v'_l + \dots + f_{\alpha,m}^{j_{i_0,\dots,i_m,m}} \cdot v'_m)$$

$$= f_{\alpha,0}^{-j_{i_0,\dots,i_m,0}} \dots f_{\alpha,l}^{-i_l} \dots f_{\alpha,m}^{-j_{i_0,\dots,i_m,m}} \otimes v'_l$$

so, replacing $v_{i_0,...,i_m}$ with v'_l , we may assume that $j_{i_0,...,i_m,l} = i_l$.

Since g has weight λ , we have that for any $h \in \mathfrak{h}$,

$$\lambda(h)g(f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m}) = (h\cdot g)(f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m}) = g(f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m}h)$$

$$= g((h - (i_0 + \dots + i_m)\alpha(h))f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m})$$

$$= (h - (i_0 + \dots + i_m)\alpha(h)) \cdot g(f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m})$$

so $g(f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m})=f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m}\otimes v_{i_0,\dots i_m}$ has weight $\lambda+(i_0+\dots+i_m)\alpha$. But by

a similar calculation as Lemma 3.3.7 (but using the untwisted action on $S_{\alpha} \otimes M$ rather than the twisted action used there), if $v \in M$ has weight λ' , then $f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes v$ has weight $\lambda' + (i_0 + \dots + i_m)\alpha$ in $S_{\alpha} \otimes M$. Hence the v_{i_0,\dots,i_m} all have weight λ in M.

Now we choose k such that $M^{\lambda+k\alpha}=0$. Then for any $i_0,\ldots,i_m\geq k$, we have

$$f_{\alpha,0}^{-k} \dots f_{\alpha,m}^{-k} \otimes v_{k,\dots,k} = g(f_{\alpha,0}^{-k} \dots f_{\alpha,m}^{-k}) = f_{\alpha,0}^{i_0-k} \dots f_{\alpha,m}^{i_m-k} \cdot g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$$
$$= f_{\alpha,0}^{-k} \dots f_{\alpha,m}^{-k} \otimes v_{i_0,\dots,i_m}$$

so $f_{\alpha,0}^{-k} \dots f_{\alpha,m}^{-k} \otimes (v_{k,\dots,k} - v_{i_0,\dots,i_m}) = 0$. But, again applying Corollary 3.3.17, we must have $v_{k,\dots,k} - v_{i_0,\dots,i_m} = 0$, i.e. $v_{k,\dots,k} = v_{i_0,\dots,i_m}$. Let $v = v_{k,\dots,k}$. Then we argue that $f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes v = f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes v_{i_0,\dots,i_m}$ for all $i_0,\dots,i_m > 0$: if $i_0,\dots,i_m \geq k$ we have just seen this, and if not then we can use that fact that

$$f_{\alpha,0}^{-i_0} \dots f_{\alpha,l}^{-i_l+1} \dots f_{\alpha,m}^{-i_m} \otimes v_{i_0,\dots,i_m} = f_{\alpha,l} \cdot g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) = g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,l}^{-i_l+1} \dots f_{\alpha,m}^{-i_m})$$

$$= f_{\alpha,0}^{-i_0} \dots f_{\alpha,l}^{-i_l+1} \dots f_{\alpha,m}^{-i_m} \otimes v_{i_0,\dots i_{l-1},\dots i_m}$$

Since S_{α} is generated by $\{f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} : i_0, \dots, i_m > 0\}$, we conclude that $g = \psi_M(v)$ for this v chosen above, completing the proof that ψ_M is surjective.

The next result is an easy but useful consequence of Frobenius reciprocity.

Lemma 3.3.20. Let M be a U-module, and let $A \subseteq S_{\alpha}$ and $\mathfrak{a} \subseteq \mathfrak{g}_m$ be as in Lemma 3.3.13. Let $\varphi : A \to M$ be a $U(\mathfrak{a})$ -homomorphism. Then φ extends uniquely to a U-homomorphism $\varphi : S_{\alpha} \to M$.

Proof. Frobenius reciprocity states that for algebras $R \subseteq S$, an R-module N and an S-module M, there is an isomorphism $\operatorname{Hom}_S(S \otimes_R N, M) \cong \operatorname{Hom}_R(N, M)$ given by the restriction map $\varphi \mapsto \varphi|_{1 \otimes R}$. Setting $R = U(\mathfrak{a}), S = U$, and N = A, the result follows from the earlier observation that $S_{\alpha} \cong U \otimes_{U(\mathfrak{a})} A$ as left U-modules. \square

The next result is a technical result which will be used later to construct certain elements of $G_{\alpha}M$.

Lemma 3.3.21. Let $M \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ and let \mathcal{I} be a subset of \mathbb{Z}^{m+1} satisfying:

- (1) $(i_0, \ldots, i_m) \in \mathcal{I}$ whenever any $i_k \leq 0$.
- (2) If $(i_0, \ldots, i_m) \in \mathcal{I}$, then $(i_0, \ldots, i_k 1, \ldots, i_m) \in \mathcal{I}$ for any $0 \le k \le m$.

and let $\{v_{i_0,\dots,i_m} \in M : (i_0,\dots,i_m) \in \mathcal{I}\}$ be a collection of elements of M satisfying:

- (i) $v_{i_0,\ldots,i_m} = 0$ whenever any $i_k \leq 0$.
- (ii) $e_{\alpha,k} \cdot v_{i_0,...,i_m} = v_{i_0,...,i_{k-1},...,i_m}$ whenever $(i_0,...,i_m) \in \mathcal{I}$

Then there exists a $U(\mathfrak{a})$ -homomorphism $\varphi: A \to \phi_{\alpha}^{-1}(M)$ such that $\varphi(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) = v_{i_0,\dots,i_m}$, which by Lemma 3.3.20 extends to a U-homomorphism $\varphi: S_{\alpha} \to \phi_{\alpha}^{-1}(M)$. Moreover, if there exists $\lambda \in \mathfrak{h}^*$ such that the weight of v_{i_0,\dots,i_m} is $\lambda - (i_0 + \dots + i_m)\alpha$ for all $(i_0,\dots,i_m) \in \mathcal{I}$, then we can choose φ to also be weight and hence lie in $G_{\alpha}M$.

Proof. We construct elements v_{i_0,\dots,i_m} for $(i_0,\dots,i_m)\in\mathbb{Z}^{m+1}\setminus\mathcal{I}$ inductively such that the v_{i_0,\dots,i_m} satisfy conditions (i) and (ii) above for any $(i_0,\dots,i_m)\in\mathbb{Z}^{m+1}$. Then observe that (i) and (ii) ensure that setting $\varphi(f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m})=v_{i_0,\dots,i_m}$ defines a $U(\mathfrak{a})$ -homomorphism $\varphi:A\to\phi_{\alpha}^{-1}(M)$, since it is enough to check that $v_{i_0,\dots,i_k-1,\dots i_m}=f_{\alpha,k}\cdot_{\alpha^{-1}}\varphi(f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m})=f_{\alpha,k}\cdot_{\alpha^{-1}}v_{i_0,\dots,i_m}=e_{\alpha,k}\cdot v_{i_0,\dots,i_m}$ for any $0\le k\le m$ and $(i_0,\dots,i_m)\in\mathbb{Z}^{m+1}$.

To construct such v_{i_0,\ldots,i_m} , let $(i_0,\ldots,i_m)\in\mathbb{Z}^{m+1}\setminus\mathcal{I}$ be such that $i_0+\cdots+i_m$ is minimal among elements of $\mathbb{Z}^{m+1}\setminus\mathcal{I}$. Then in particular, $(i_0,\ldots,i_k-1,\ldots,i_m)\in\mathcal{I}$ for each $0\leq k\leq m$. Recall that $(\mathfrak{sl}_2)_{\alpha}:=\langle e_{\alpha},h_{\alpha},f_{\alpha}\rangle$ and let N be the $((\mathfrak{sl}_2)_{\alpha})_m$ -module generated by $\{v_{i_0,\ldots,i_k-1,\ldots,i_m}:0\leq k\leq m\}$, so $N\in\mathcal{O}^{(\mu)}(((\mathfrak{sl}_2)_{\alpha})_m)$. Here we make a slight abuse of notation, identifying μ with its restriction to $\mathfrak{h}_m\cap((\mathfrak{sl}_2)_{\alpha})_m$. Then we use the following claim to construct $v_{i_0,\ldots,i_m}\in N\subseteq M$ such that $e_{\alpha,k}\cdot v_{i_0,\ldots,i_m}=m_{i_0,\ldots,i_k-1,\ldots,i_m}$.

Claim (*). Consider the maps:

$$N \xrightarrow{\theta_1} N^{m+1} \xrightarrow{\theta_2} N^{\frac{1}{2}m(m+1)}$$

given by $\theta_1(x) = (e_{\alpha,0} \cdot x, \dots, e_{\alpha,m} \cdot x)$ and $\theta_2(y_0, \dots, y_m) = (e_{\alpha,k} \cdot y_l - e_{\alpha,l} \cdot y_k)_{0 \le k < l \le m}$. Then $\ker(\theta_2) = \operatorname{im}(\theta_1)$. Furthermore, if $(y_0, \dots, y_m) \in \ker(\theta_2)$ and $y_k \in N \cap M^{\lambda}$ for all $0 \le k \le m$, then there exists $x \in M^{\lambda - \alpha}$ such that $\theta_1(x) = (y_0, \dots, y_m)$.

We now proceed to prove this claim. First observe that $\theta_2 \circ \theta_1 = 0$, so certainly $\ker(\theta_2) \supseteq \operatorname{im}(\theta_1)$. Hence we only need to show that $\operatorname{im}(\theta_1) \supseteq \ker(\theta_2)$. Throughout the proof of this claim, we write e_k for $e_{\alpha,k}$ and f_k for $f_{\alpha,k}$.

We first deal with the case where m=1 and $N=M(\gamma)$ is a Verma module. In this case, we consider the restriction of these maps to certain weight spaces in the following way (for any $\lambda \in \mathfrak{h}^*$):

$$N^{\lambda} \xrightarrow{\theta_1} (N^{\lambda+\alpha})^2 \xrightarrow{\theta_2} N^{\lambda+2\alpha}$$

Now, either $\dim(N^{\lambda+\alpha})=0$, in which case $\operatorname{im}(\theta_1)=\ker(\theta_2)$ automatically, or the dimensions of these weight spaces satisfy $\dim(N^{\lambda})=m+1$, $\dim(N^{\lambda+\alpha})=m$, and $\dim(N^{\lambda+2\alpha})=m-1$. Hence by considering dimensions and the fact that $\ker(\theta_2)\supseteq \operatorname{im}(\theta_1)$, it is enough to show that θ_1 is injective and θ_2 is surjective. We can compute that $e_{\alpha,1}$ acts on the basis vectors $f_{\alpha,1}^i f_{\alpha,0}^j \otimes 1$ by:

$$e_{\alpha,1} \cdot (f_{\alpha,1}^i f_{\alpha,0}^j \otimes 1) = \mu j f_{\alpha,1}^i f_{\alpha,0}^{j-1} \otimes 1 - j(j-1) f_{\alpha,1}^{i+1} f_{\alpha,0}^{j-2} \otimes 1$$

so by considering $\theta_2(x,0)$, we see θ_2 is surjective using an inductive argument. We also see that $e_{\alpha,1} \cdot v = 0$ if and only if $v = f_{\alpha,1}^i \otimes 1$. Since $\mu \neq 0$, we can show that $e_{\alpha,0} \cdot f_1^i \otimes 1 \neq 0$, so θ_1 is injective as required.

We now deal with the case where $m \geq 1$ and $N = M(\lambda)$ is a Verma module in $\mathcal{O}^{(\mu)}(\mathfrak{sl}_2)_m)$, so $\lambda_{\geq 1} = \mu$. We use the following facts, which can be verified by computing the action of e_m on the basis elements $f_0^{i_0} \dots f_m^{i_m} \otimes 1$ of $M(\lambda)$ and recalling that $\mu(h_\alpha) \neq 0$:

- (a) $e_m \cdot N = N$, which can be shown by an inductive argument using a computation similar the $\mathfrak{g} = \mathfrak{sl}_2$ Verma case.
- (b) $e_m \cdot v = 0$ if and only if $v \in \text{span}\{f_1^{i_1} \dots f_m^{i_m} \otimes 1 : i_1, \dots, i_m \geq 0\}$. The latter is isomorphic as a $U(\langle e_0, \dots, e_{m-1} \rangle)$ -module to the Verma module $M(\gamma)$ over $\text{span}\{e_{\alpha,i}, h_{\alpha,i}, f_{\alpha,i} \mid i = 0, \dots, m-1\} \cong ((\mathfrak{sl}_2)_{\alpha})_{m-1}$ where $\gamma(h_{\alpha,i}) := \mu(h_{\alpha,i+1})$.
- (c) Applying (b) repeatedly, we see that for any $k \geq 1$, we have that $e_l \cdot v = 0$ for all $k \leq l \leq m$ if and only if $v \in \text{span}\{f_{m-k+1}^{i_{m-k+1}} \dots f_m^{i_m} \otimes 1 : i_{m-k+1}, \dots, i_m \geq 0\}$.

To ease notation slightly we will write $\mathfrak{a}_{(i)} = \langle e_0, \dots, e_i \rangle$, write $(\mathfrak{sl}_2)_{(i)}$ for the Lie algebra $((\mathfrak{sl}_2)_{\alpha})_i$ and $\gamma_{(i)}$ for the character of span $\{h_{\alpha,0},\dots,h_{\alpha,i}\}$ given by $\gamma_{(i)}(h_{\alpha,j}) := \mu(h_{\alpha,m-i})$.

Now let $(y_0, \ldots, y_m) \in \ker(\theta_2)$, i.e. $e_k \cdot y_l - e_l \cdot y_k = 0$ for all $0 \le k, l \le m$. By fact (a), there certainly exists an x_m such that $e_m \cdot x_m = y_m$. We then inductively construct x_k for $2 \le k \le m$ such that $e_l \cdot x_k = y_l$ for all $k \le l \le m$. The cases k = 0, 1 will be dealt with by an additional argument immediately afterwards, and we will then obtain some x_0 such that $\theta_1(x_0) = (y_0, \ldots, y_m)$.

Suppose we have constructed x_{k+1} such that $e_l \cdot x_{k+1} = y_l$ for all $k+1 \le l \le m$. For any $k+1 \le l \le m$, consider $e_l \cdot (e_k \cdot x_{k+1} - y_k) = e_k \cdot (e_l \cdot x_{k+1}) - e_l \cdot y_k = e_k \cdot y_l - e_l \cdot y_k = 0$. Hence by fact (c), we have $e_k \cdot x_{k+1} - y_k \in \text{span}\{f_{m-k}^{i_{m-k}} \dots f_m^{i_m} \otimes 1 : i_{m-k}, \dots, i_m \ge 0\}$. But as a $U(\mathfrak{a}_{(k)})$ -module, this is isomorphic to $M(\gamma_{(k)})$, so by fact (a) there exists $x_k' \in \text{span}\{f_{m-k}^{i_{m-k}} \dots f_m^{i_m} \otimes 1 : i_{m-k}, \dots, i_m \ge 0\}$ such that $e_k \cdot x_k' = e_k \cdot x_{k+1} - y_k$, and by fact (c) $e_l \cdot x_k' = 0$ for all $k+1 \le l \le m$. Setting $x_k = x_{k+1} - x_k'$, we see that $e_k \cdot x_k = e_k \cdot x_{k+1} - e_k \cdot x_k' = e_k \cdot x_{k+1} - (e_k \cdot x_{k+1} - y_k) = y_k$, and for $k+1 \le l \le m$ we have $e_l \cdot x_k = e_l \cdot x_{k+1} - e_l \cdot x_k' = y_l - 0$. Hence we have constructed x_k with the desired properties.

Now consider $y_1' = e_1 \cdot x_2 - y_1$ and $y_0' = e_0 \cdot x_2 - y_0$. For $2 \leq l \leq n$, we have $e_l \cdot y_1' = e_l \cdot (e_1 \cdot x_2 - y_1) = e_1 \cdot y_1 - e_l \cdot y_1 = 0$, and similarly $e_l \cdot y_0' = 0$, so $y_0', y_1' \in \text{span}\{f_{m-1}^{i_{m-1}}f_m^{i_m} \otimes 1 : i_{m-1}, i_m \geq 0\}$ which is isomorphic to $M(\gamma_{(1)})$, the Verma module over $(\mathfrak{sl}_2)_{(1)}$, as a $U(\langle e_0, e_1 \rangle)$ -module. In addition, since $(y_0, \ldots, y_m) \in \text{ker } \theta$ we have

 $e_0 \cdot y_1' - e_1 \cdot y_0' = 0$, so we can apply the case where N is a Verma module and n = 1, proved earlier, to find x' such that $e_0 \cdot x' = y_0'$, $e_1 \cdot x' = y_1'$, and $e_l \cdot x' = 0$ for $2 \le l \le n$. Then setting $x_0 = x_2 - x'$, we have that $\theta_1(x_0) = (y_0, \dots, y_m)$.

If N is not a Verma module, let $0 = N_0 \subseteq N_1 \subseteq \cdots \subseteq N_{k-1} \subseteq N_k = N$ be a filtration of N such that each quotient is a highest weight module. Since $\mu(h_{\alpha,m}) \neq 0$, all these highest weight modules must in fact be Verma modules by [56, Theorem 7.1]. Then, given $(y_0, \ldots, y_m) \in \ker(\theta_2)$, in the quotient N/N_{k-1} we have that there exists $x \in N$ such that $e_l \cdot x + N_{k-1} = y_l + N_{k-1}$ for all $0 \leq l \leq m$. Hence $e_l \cdot x - y_l \in N_{k-1}$ for all $0 \leq l \leq n$. But $e_{l'} \cdot (e_l \cdot x - y_l) = e_l \cdot (e_{l'} \cdot x - y_{l'})$ for all $0 \leq l, l', \leq m$, so by induction on k, there exists $x' \in N_{k-1}$ such that $e_l \cdot x' = e_l \cdot x - y_l$ for all $0 \leq l \leq mn$. Note that in this final argument we have used the fact that $y'_l := e_l \cdot x - y_l$ gives a collection of elements lying in the kernel of θ , which allows us to apply the inductive hypothesis. Hence $e_l \cdot (x - x') = y_l$ for all $0 \leq l \leq m$, so $\operatorname{im}(\theta_1) \supseteq \ker(\theta_2)$ as required.

Finally, if $y_0, \ldots, y_m \in M^{\lambda} \cap N$, then given $x \in N \subseteq M$ such that $\theta_1(x) = (y_0, \ldots, y_m)$, by considering weight spaces we may replace x with its component $x^{\lambda-\alpha}$ in the $\lambda-\alpha$ weight space and $\theta_1(x) = \theta_1(x^{\lambda-\alpha})$, proving the final part of Claim (*).

Since $e_{\alpha,k} \cdot v_{i_0,\dots,i_l-1,\dots,i_m} = v_{i_0,\dots,i_k-1,\dots,i_l-1,\dots i_m} = e_{\alpha,l} \cdot m_{i_0,\dots,i_k-1,\dots i_m}$ for all $0 \le k < l \le m$, this claim then allows us to pick v_{i_0,\dots,i_m} such that $e_{\alpha,k} \cdot v_{i_0,\dots,i_m} = v_{i_0,\dots,i_k-1,\dots i_m}$ for all $0 \le k \le m$, and if $v_{i_0,\dots,i_k-1,\dots,i_m}$ has weight λ for all $0 \le k \le m$, then we can choose v_{i_0,\dots,i_m} to have weight $\lambda - \alpha$. Hence applying this inductively, we can construct for all $(i_0,\dots,i_m) \in \mathbb{Z}^{m+1}$ elements $v_{i_0,\dots,i_m} \in M$ satisfying conditions (i) and (ii).

Suppose there exists $\lambda \in \mathfrak{h}^*$ such that $\operatorname{wt}(m_{i_0,\dots,i_m}) = \lambda - (i_0 + \dots + i_m)\alpha$ for all $(i_0,\dots,i_m) \in \mathcal{I}$. Then by construction v_{i_0,\dots,i_m} is weight for all $(i_0,\dots,i_m) \in \mathbb{Z}^{m+1}$, so by Corollary 3.3.9 the function we have constructed is a weight vector.

We are now ready to establish the main remaining ingredient needed to finish the proof of Theorem 3.3.11.

Lemma 3.3.22. For any $N \in \mathcal{O}^{(s_{\alpha}(\mu))}(\mathfrak{g}_m)$, the map $\epsilon_N : T_{\alpha}G_{\alpha}N \to N$ given by $\epsilon_N(s \otimes s)$

g) = g(s) is an isomorphism.

Proof. First we show ϵ_N is a homomorphism. Let $u \in U$, let $s \in S_\alpha$, and let $g \in G_\alpha N$. Then

$$u \cdot \epsilon_N(s \otimes g) = u \cdot g(s) = \phi_{\alpha}^{-1} \phi_{\alpha}(u) \cdot g(s) = g(\phi_{\alpha}(u) \cdot s) = \epsilon_N((\phi_{\alpha}(u) \cdot s) \otimes g) = \epsilon_N(u \cdot (s \otimes g))$$

so ϵ_N is certainly a *U*-homomorphism.

Now let $v \in N$ be a weight vector, and choose a_0, \ldots, a_m such that $e_{\alpha,k}^{a_k} \cdot v = 0$ for each k. Let

$$\mathcal{I} = \{(i_0, \dots, i_m) : i_k \leq 0 \text{ for some } k \text{ or } i_k \leq a_k \text{ for all } k\} \subseteq \mathbb{Z}^{m+1}$$

$$v_{i_0, \dots, i_m} = e_{\alpha, 0}^{a_0 - i_0} \dots e_{\alpha, m}^{a_m - i_m} \cdot v \text{ if } i_k \leq a_k \text{ for all } k$$

$$v_{i_0, \dots, i_m} = 0 \text{ otherwise.}$$

Then we can use Lemma 3.3.21 to construct $g \in \text{Hom}_U(S_\alpha, \phi_\alpha^{-1}(M))$ which is a weight vector and therefore in $G_\alpha N$ such that $g(f_{\alpha,0}^{-a_0} \dots f_{\alpha,m}^{-a_m}) = v$. Hence $\epsilon_N(f_{\alpha,0}^{-a_0} \dots f_{\alpha,m}^{-a_m} \otimes g) = v$, so ϵ_N is surjective.

We now show ϵ_N is injective. By Lemma 3.3.18 any element of $T_{\alpha}G_{\alpha}N$ may be written as $f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes g$ for some $g \in G_{\alpha}N$. Suppose $\epsilon_N(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes g) = g(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) = 0$. We may assume g is weight (i.e. an \mathfrak{h} -eigenvector). If not, we may write g as a sum of $g_{\lambda} \in (G_{\alpha}N)^{\lambda}$, which by considering weight spaces must all satisfy $g_{\lambda}(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) = 0$, and then apply the following argument to each g_{λ} .

Observe that $g(f_{\alpha,0}^{-j_0}\dots f_{\alpha,m}^{-j_m})=0$ if either $j_k\leq 0$ for some $0\leq k\leq n$ or all $j_k\leq i_k$,

and choose

$$\mathcal{I} = \left\{ (j_0, \dots, j_m) \in \mathbb{Z}^{m+1} : \begin{array}{c} j_k \le i_k \text{ for all } 0 \le k \le m-1 \text{ or } j_m \le i_m \\ \text{or } j_k \le 0 \text{ for some } 0 \le k \le m \end{array} \right\} \subseteq \mathbb{Z}^{m+1}$$

$$v_{j_0,\ldots,j_m}=g(f_{\alpha,0}^{-j_0}\ldots f_{\alpha,m}^{-j_m})$$
 if $j_k\leq i_k$ for all $0\leq k\leq m-1$

$$v_{j_0,\dots,j_m}=0$$
 if $j_m\leq i_m$ or $j_k\leq 0$ for some $0\leq k\leq m$.

Then applying Lemma 3.3.21 to this, we construct $g_m \in \text{Hom}_U(S_\alpha, \phi_\alpha^{-1}(M))$ which is a weight vector (and hence in $G_\alpha N$) such that:

(a)
$$g_m(f_{\alpha,0}^{-j_0} \dots f_{\alpha,m}^{-j_m}) = 0$$
 if $j_m \le i_m$

(b)
$$g_m(f_{\alpha,0}^{-j_0}\dots f_{\alpha,m}^{-j_m}) = g(f_{\alpha,0}^{-j_0}\dots f_{\alpha,m}^{-j_m})$$
 if $j_k \leq i_k$ for all $0 \leq k \leq m-1$

Note that these two conditions are not mutually exclusive but they are consistent by our earlier observation.

By (a), we can define $g'_m \in G_\alpha N$ by setting

$$g'_m(f_{\alpha,0}^{-j_0}\dots f_{\alpha,m-1}^{-j_{m-1}}f_{\alpha,m}^{-j_m+i_m})=g_m(f_{\alpha,0}^{-j_0}\dots f_{\alpha,m}^{-j_m})$$

and so $g_m = f_{\alpha,m}^{i_m} \cdot g_m'$. By (b) we have $(g - g_m)(f_{\alpha,0}^{-j_0} \dots f_{\alpha,m}^{-j_m}) = 0$ if either some $j_k \leq 0$ or if $j_k \leq i_k$ for all $0 \leq k \leq m-1$.

We now construct $g_s \in G_{\alpha}N$ inductively by setting

$$\mathcal{I} = \left\{ (j_0, \dots, j_m) \in \mathbb{Z}^{m+1} : \begin{array}{c} j_k \le i_k \text{ for all } 0 \le k \le s - 1 \\ \text{or } j_s \le i_s \text{ or } j_k \le 0 \text{ for some } 0 \le k \le m \end{array} \right\} \subseteq \mathbb{Z}^{m+1}$$

$$v_{j_0,...,j_m} = (g - g_m - \dots - g_{s+1})(f_{\alpha,0}^{-j_0} \dots f_{\alpha,m}^{-j_m}) \text{ if } j_k \le i_k \text{ for all } 0 \le k \le s-1$$

$$v_{j_0,\ldots,j_m}=0$$
 if $j_s\leq i_s$ or $j_k\leq 0$ for some $0\leq k\leq m$

Again, these two conditions are not mutually exclusive, but they are consistent.

These g_s satisfy:

(a)
$$g_s(f_{\alpha,0}^{-j_0} \dots f_{\alpha,m}^{-j_m}) = 0 \text{ if } j_s \leq i_s$$

(b)
$$g_s(f_{\alpha,0}^{-j_0} \dots f_{\alpha,m}^{-j_m}) = (g - g_m - \dots - g_{s+1})(f_{\alpha,0}^{-j_0} \dots f_{\alpha,m}^{-j_m})$$
 if $j_k \leq i_k$ for all $0 \leq k \leq s-1$

By (a), we can write this g_s in the form $g_s = f_{\alpha,s}^{i_s} \cdot g_s'$ for some $g_s' \in G_\alpha N$. By (b) we have that $(g - g_m - \dots - g_s)(f_{\alpha,0}^{-j_0} \dots f_{\alpha,m}^{-j_m}) = 0$ if $j_k \leq i_k$ for all $0 \leq k \leq s-1$. Hence we see that $g = f_{\alpha,0}^{-i_0} \cdot g_0' + \dots + f_{\alpha,m}^{-i_m} \cdot g_m'$ for some $g_0', \dots, g_m' \in G_\alpha N$, so $f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m} \otimes g = \sum (f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) \otimes f_{\alpha,k}^{i_k} g_k' = 0$. Hence ϵ_N is injective.

The last main step in proving Theorem 3.3.11 is to show that G_{α} genuinely a functor between appropriate blocks of $\mathcal{O}(\mathfrak{g}_m)$ as claimed.

Lemma 3.3.23. G_{α} restricts to a functor $\mathcal{O}^{(s_{\alpha}(\mu))}(\mathfrak{g}_m) \to \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$.

Proof. Let $N \in \mathcal{O}^{(s_{\alpha}(\mu))}$. Let $0 = N_0 \subseteq N_1 \subseteq \cdots \subseteq N_{k-1} \subseteq N_k = N$ be a filtration of N such that each section is a highest weight module. We have a short exact sequence

$$0 \to N_1 \to N \to (N/N_1) \to 0$$

and hence, since G_{α} is left exact, an exact sequence

$$0 \to G_{\alpha} N_1 \to G_{\alpha} N \to G_{\alpha} (N/N_1).$$

Hence $G_{\alpha}N$ is an extension of a submodule of $G_{\alpha}(N/N_1)$ by $G_{\alpha}N_1$. Since $G_{\alpha}N$ is automatically \mathfrak{h} -semisimple and $\mathcal{O}^{(s_{\alpha}(\mu))}$ is closed under taking submodules, we can use induction on k and Lemma 3.3.6 to reduce to the case where N is a highest weight module of weight $(\lambda, s_{\alpha}(\mu_1), \ldots, s_{\alpha}(\mu_m))$.

Let v be a highest weight generator of N. By Lemma 3.3.21, we can find $g_v \in G_{\alpha}N$ such that $g_v(f_{\alpha,0}^{-1}\dots f_{\alpha,m}^{-1})=v$ and g_v is a weight vector. We aim to show that g_v is a highest weight generator of $G_{\alpha}N$ of weight $(s_{\alpha}(\lambda)-(m+1)\alpha,\mu_1,\dots,\mu_m)$. By Lemma 3.3.8, we certainly have that g_v is a weight vector of weight $s_{\alpha}(\lambda)-(m+1)\alpha$.

Let $\beta \in \Phi^+ \setminus \{\alpha\}$. Then $e_{\beta,k} \cdot g_v$ has weight $s_{\alpha}(\lambda) - (m-1)\alpha + \beta$, so by Lemma 3.3.8, $(e_{\beta,k} \cdot g_v)(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m})$ has weight $\lambda - (i_0 + \dots + i_m - (m+1))\alpha + s_{\alpha}(\beta)$. Now, since $s_{\alpha}(\beta) \in \Phi^+ \setminus \{\alpha\}$, $\lambda - (i_0 + \dots + i_m - (m+1))\alpha + s_{\alpha}(\beta) \nleq \lambda$ and so $N^{\lambda - (i_0 + \dots + i_m - (m+1))\alpha + s_{\alpha}(\beta)} = 0$. In particular, $(e_{\beta,k} \cdot g_v)(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) = 0$ for all $i_0, \dots, i_m > 0$, so $e_{\beta,k} \cdot g_v = 0$ for all $0 \le k \le m$.

To see that $e_{\alpha,k} \cdot g_v = 0$, first consider the $((\mathfrak{sl}_2)_{\alpha})_m$ -module $\bigoplus_{l \in \mathbb{Z}_{\geq 0}} N^{\lambda - l\alpha}$. Since N is highest weight, this is generated as a $((\mathfrak{sl}_2)_{\alpha})_m$ -module by v, which is highest weight of weight $(\lambda(h_{\alpha}), s_{\alpha}(\mu_1)(h_{\alpha}), \ldots, s_{\alpha}(\mu_m)(h_{\alpha}))$. Hence since $s_{\alpha}(\mu_m)(h_{\alpha}) \neq 0$ and this module is clearly non-zero, by [56, Theorem 7.1] this module is isomorphic to

$$M(\lambda(h_{\alpha}), s_{\alpha}(\mu_1)(h_{\alpha}), \ldots, s_{\alpha}(\mu_m)(h_{\alpha}))$$

and so in particular the only elements $x \in \bigoplus_{l \in \mathbb{N}} N^{\lambda - l\alpha}$ such that $e_{\alpha,k} \cdot x = 0$ for all $0 \le k \le m$ are scalar multiples of v. Now suppose that $e_{\alpha,k} \cdot g_v \ne 0$. Then we have a non-zero element $g \in (G_{\alpha}N)^{s_{\alpha}(\lambda)-m\alpha}$. By Lemma 3.3.8, $g(f_{\alpha,0}^{-1}\dots f_{\alpha,m}^{-1}) = x$ must have weight $\lambda - m\alpha$. But also $e_{\alpha,k} \cdot x = 0$ for all $0 \le k \le m$, so by the above observation x is a scalar multiple of v and so has weight λ . Hence x = 0, and we can now show inductively by a similar argument that $g(f_{\alpha,0}^{-i_0}\dots f_{\alpha,m}^{-i_m}) = 0$ for all $i_0,\dots,i_m > 0$, so g = 0 giving a contradiction.

Finally, let $h \in \mathfrak{h}$ and let $1 \leq k \leq m$. Then $h_k \cdot g_v \in (G_{\alpha}N)^{s_{\alpha}(\lambda)-(m+1)\alpha}$. We have:

$$(h_k \cdot g_v)(f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1}) = g_v(f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1} h_k)$$

$$= g_v(h_k f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1} + \alpha(h) \sum_{0 \le l \le m-k} f_{\alpha,0}^{-1} \dots f_{\alpha,l}^{-2} \dots f_{\alpha,l+k}^{0} \dots f_{\alpha,m}^{-1})$$

$$= g_v(h_k f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1})$$

$$= h_k \cdot v = \mu_k(h)v$$

Now suppose $h_k \cdot g_v \neq \mu_k(h)g_v$. Let (i_0, \dots, i_m) be such that $i_0 + \dots + i_m$ is minimal subject to $x := (h_k \cdot g_v - \mu_k(h)g_v)(f_{\alpha,0}^{-i_0} \dots f_{\alpha,m}^{-i_m}) \neq 0$, and note that $(i_0, \dots, i_m) \neq (1, \dots, 1)$ by

the calculation above. Then by the minimality of $i_0 + \cdots + i_m$, we have $e_{\alpha,k} \cdot x = 0$ for each $0 \le k \le m$. But by Lemma 3.3.8, we have $x \in N^{\lambda - (i_0 + \cdots + i_m - (m+1))\alpha}$, which gives a contradiction by the earlier observation that the only elements $x \in \bigoplus_{l \in \mathbb{N}} N^{\lambda - l\alpha}$ such that $e_{\alpha,k} \cdot x = 0$ for all $0 \le k \le m$ are scalar multiples of m. Hence $h_k \cdot g_v = \mu_k(h)g_v$, completing the proof that g_v is highest weight of weight $(s_{\alpha}(\lambda) - (m+1)\alpha, \mu_1, \dots, \mu_m)$ as claimed.

Now, by 3.3.22, there is an isomorphism ϵ_N between $T_{\alpha}G_{\alpha}N$ and N, and furthermore this isomorphism takes $f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1} \otimes g_v$ to $g_v(f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1}) = v$. Hence $f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1} \otimes g_v$ is a highest weight generator of $T_{\alpha}G_{\alpha}N$. Now let L be the submodule of $G_{\alpha}N$ generated by g_v , and consider the inclusion $\iota: L \hookrightarrow G_{\alpha}N$. We wish to show that ι is an isomorphism, which will complete the proof that N is highest weight and therefore the proof that G_{α} is a functor from $\mathcal{O}^{s_{\alpha}(\mu)}$ to $\mathcal{O}^{(\mu)}$. First we compute that for any $l \in L, s \in S_{\alpha}$:

$$((G_{\alpha}(\epsilon_{N}) \circ G_{\alpha}T_{\alpha}(\iota) \circ \psi_{L})(l))(s) = \epsilon_{N}((G_{\alpha}T_{\alpha}(\iota) \circ \psi_{L})(l)(s))$$

$$= \epsilon_{N}(T_{\alpha}(\iota) \circ \psi_{L}(l)(s))$$

$$= \epsilon_{N}(T_{\alpha}(\iota)(s \otimes l))$$

$$= \epsilon_{N}(s \otimes \iota(l))$$

$$= \iota(l)(s)$$

The first and second equalities hold since by the definition of G_{α} if $\chi \in G_{\alpha}T_{\alpha}G_{\alpha}N$ and ρ is a map from $T_{\alpha}G_{\alpha}N$ to either N or L, then $G_{\alpha}(\rho)(\chi) = \rho \circ \chi$. The final three equalities hold by the definitions of ψ_L , $T_{\alpha}(\iota)$, and ϵ_N respectively.

Hence $\iota = G_{\alpha}(\epsilon_N) \circ G_{\alpha}T_{\alpha}(\iota) \circ \psi_L$. Since L is highest weight, by Lemma 3.3.19 the map ψ_L is certainly an isomorphism. Similarly, by Lemma 3.3.22, the map ϵ_N is an isomorphism, so $G_{\alpha}(\epsilon_N)$ is also an isomorphism. Hence to show ι is an isomorphism it suffices to show $G_{\alpha}T_{\alpha}(\iota)$ is an isomorphism, and to show this it suffices to show $T_{\alpha}(\iota)$ is an isomorphism. We can also conclude from this calculation that $G_{\alpha}T_{\alpha}(\iota)$ is injective.

Now we consider $T_{\alpha}(\iota): T_{\alpha}L \to T_{\alpha}G_{\alpha}N$. We have that $(T_{\alpha}(\iota))(f_{\alpha,0}^{-1}\dots f_{\alpha,m}^{-1}\otimes g_v) =$

 $f_{\alpha,0}^{-1} \dots f_{\alpha,m}^{-1} \otimes g_v$ which as discussed above generates $T_{\alpha}G_{\alpha}N$, so $T_{\alpha}(\iota)$ is certainly surjective. Let $K = \ker(T_{\alpha}(\iota))$ and observe since that K is a submodule of $T_{\alpha}L \in \mathcal{O}^{(\mu)}$, we have $K \in \mathcal{O}^{(\mu)}$.

We have a short exact sequence:

$$0 \to K \xrightarrow{\varphi} T_{\alpha}L \xrightarrow{T_{\alpha}(\iota)} T_{\alpha}G_{\alpha}N \to 0$$

and so since G_{α} is left exact, we have an exact sequence:

$$0 \to G_{\alpha}K \overset{G_{\alpha}(\varphi)}{\to} G_{\alpha}T_{\alpha}L \overset{G_{\alpha}T_{\alpha}(\iota)}{\to} G_{\alpha}T_{\alpha}G_{\alpha}N$$

But $G_{\alpha}T_{\alpha}(\iota)$ is injective, so $\operatorname{im}(G_{\alpha}(\varphi)) = 0$ and hence $G_{\alpha}K = 0$. Now, if $K \neq 0$ then since $K \in \mathcal{O}^{(\mu)}$ we can use Lemma 3.3.21 to construct a non-zero element of $G_{\alpha}K$, so K = 0, i.e. $T_{\alpha}(\iota)$ is injective. Hence $T_{\alpha}(\iota)$ is an isomorphism, so by the earlier discussion ι is an isomorphism, so $G_{\alpha}N$ is highest weight as required.

Finally, to complete the proof of Theorem 3.3.11 we need a small argument to check that ψ and ϵ are both natural transformations:

Lemma 3.3.24. The transformations

$$\psi: \mathrm{id}_{\mathcal{O}^{(\mu)}(\mathfrak{g}_m)} \to G_{\alpha} T_{\alpha}$$

and

$$\epsilon: T_{\alpha}G_{\alpha} \to \mathrm{id}_{\mathcal{O}^{(s_{\alpha}(\mu))}(\mathfrak{q}_{m})}$$

are natural.

Proof. To show ψ is natural, we must show that for any $M, N \in \mathcal{O}^{(\mu)}$ and $f: M \to N$ that $\psi_N \circ f = G_\alpha T_\alpha f \circ \psi_M$. Using that $G_\alpha(\rho)(\chi) = \rho \circ \chi$, we see that for any $s \in S_\alpha$ and

 $v \in M$:

$$(G_{\alpha}T_{\alpha}f \circ \psi_{M}(v))(s) = G_{\alpha}T_{\alpha}f(s \otimes v)$$

$$= T_{\alpha}f(s \otimes v)$$

$$= s \otimes f(v)$$

$$= \psi_{N}(f(v))(s)$$

$$= (\psi_{N} \circ f)(v)(s).$$

To show ϵ is natural, we must show that for any $M, N \in \mathcal{O}^{(s_{\alpha}(\mu))}$ and $f: M \to N$ that $f \circ \epsilon_M = \epsilon_N \circ T_{\alpha} G_{\alpha} f$. This holds since for any $s \in S_{\alpha}$ and $g \in G_{\alpha} M$, we have:

$$(\epsilon_N \circ T_\alpha G_\alpha f)(s \otimes g) = \epsilon_N (s \otimes (G_\alpha f)(g))$$

$$= \epsilon_N (s \otimes (f \circ g))$$

$$= f(g(s))$$

$$= f(\epsilon_M (s \otimes g))$$

$$= (f \circ \epsilon_M)(s \otimes g)$$

Theorem 3.3.11 now follows by combining Lemmas 3.3.12, 3.3.19, 3.3.22, 3.3.23, and 3.3.24.

We end this section by remarking that Lemma 3.1.11, Theorem 3.2.1, and Theorem 3.3.11 can be combined (together with a small result on the action of the Weyl group on \mathfrak{h}^* ; see Proposition 3.4.7) to show that every Jordan block $\mathcal{O}^{(\mu_1,\ldots,\mu_m)}(\mathfrak{g}_m)$ is equivalent to another Jordan block $\mathcal{O}^{(\nu_1,\ldots,\nu_{m-1},0)}(\mathfrak{l}_m)$ where \mathfrak{l} is some Levi factor of \mathfrak{g} . We will discuss this in more detail later during our calculation of composition multiplicities in §3.4.2. For now, we illustrate this via the following example:

Example 3.3.25. Let $\mathfrak{g} = \mathfrak{sl}_3$ and m = 1, let $\Phi^+ = \{\alpha, \beta, \alpha + \beta\}$ be the positive roots

of \mathfrak{sl}_3 labelled in the usual way, and consider the Jordan blocks $\mathcal{O}^{(\mu)}((\mathfrak{sl}_3)_1)$ as μ ranges over \mathfrak{h}^* . The possible stabilisers $(\mathfrak{sl}_3)^{\mu}$ are:

$$\mathfrak{l}_{1} = \begin{pmatrix} * & * & * \\ * & * & * \\ * & * & * \end{pmatrix}, \qquad \mathfrak{l}_{2} = \begin{pmatrix} * & 0 & 0 \\ 0 & * & 0 \\ 0 & 0 & * \end{pmatrix}, \qquad \mathfrak{l}_{3} = \begin{pmatrix} * & * & 0 \\ * & * & 0 \\ 0 & 0 & * \end{pmatrix}$$

$$\mathfrak{l}_{4} = \begin{pmatrix} * & 0 & 0 \\ 0 & * & * \\ 0 & * & * \end{pmatrix}, \qquad \mathfrak{l}_{5} = \begin{pmatrix} * & 0 & * \\ 0 & * & 0 \\ * & 0 & * \end{pmatrix}$$

We then have the following cases:

- If $(\mathfrak{sl}_3)^{\mu} = \mathfrak{l}_1$ then $\mu = 0$ so we are already in the Jordan block $\mathcal{O}^{(0)}((\mathfrak{sl}_3)_1)$.
- If $(\mathfrak{sl}_3)^{\mu} = \mathfrak{l}_2 = \mathfrak{h}$, then $(\mathfrak{sl}_3)^{\mu}$ is the Levi factor of the standard parabolic $\mathfrak{p} = \mathfrak{b}$. Then by Theorem 3.2.1 and Lemma 3.1.11 we have that $\mathcal{O}^{(\mu)}((\mathfrak{sl}_3)_1) \cong \mathcal{O}^{(0)}(\mathfrak{h}_1)$.
- If $(\mathfrak{sl}_3)^{\mu} = \mathfrak{l}_3$ or $(\mathfrak{sl}_3)^{\mu} = \mathfrak{l}_4$ then is it the Levi factor of a standard parabolic \mathfrak{p}_3 or \mathfrak{p}_4 respectively, where:

$$\mathfrak{p}_3 = \begin{pmatrix} * & * & * \\ * & * & * \\ 0 & 0 & * \end{pmatrix}, \qquad \mathfrak{p}_4 = \begin{pmatrix} * & * & * \\ 0 & * & * \\ 0 & * & * \end{pmatrix}$$

Hence we can again apply Theorem 3.2.1 and Lemma 3.1.11 to show that $\mathcal{O}^{(\mu)}((\mathfrak{sl}_3)_1) \cong \mathcal{O}^{(0)}((\mathfrak{gl}_2)_1)$, since $\mathfrak{l}_3 \cong \mathfrak{gl}_2 \cong \mathfrak{l}_4$.

• If $(\mathfrak{sl}_3)^{\mu} \cong \mathfrak{l}_5$ then there is no parabolic subalgebra \mathfrak{p}_5 containing \mathfrak{b} whose Levi factor is \mathfrak{l}_5 . Hence we cannot directly apply Theorem 3.2.1. Instead, we first use Theorem 3.3.11 to obtain an equivalence $\mathcal{O}^{(\mu)}((\mathfrak{sl}_3)_1) \cong \mathcal{O}^{(s_{\alpha}(\mu))}((\mathfrak{sl}_3)_1)$ via the twisting functor T_{α} . Then one can check that $\mathfrak{g}^{s_{\alpha}(\mu)} = \mathfrak{l}_4$, so by the previous case we see that $\mathcal{O}^{(\mu)}((\mathfrak{sl}_3)_1) \cong \mathcal{O}^{(0)}((\mathfrak{gl}_2)_1)$.

3.4 Composition multiplicaties of Verma modules

3.4.1 Definition of composition multiplicities

Let $M \in \mathcal{O}(\mathfrak{g}_m)$. As in the reductive case covered in §2.3.2, we define the *character* of M to be the function $\mathrm{ch}(M):\mathfrak{h}^*\to\mathbb{Z}_{\geq 0}$ which sends λ to $\dim(M^\lambda)$, and the *support* of M, denoted $\mathrm{supp}(M)$, to be the set $\mathrm{supp}(M)=\{\lambda\in\mathfrak{h}^*:M^\lambda\neq 0\}$.

Since modules in $\mathcal{O}(\mathfrak{g}_m)$ need not have finite length, we require an alternative way to define composition multiplicities. We do this using the using the following result, generalising the case $\mathfrak{g} = \mathfrak{sl}_2$, m = 1 treated in [34, Proposition 8].

Lemma 3.4.1. Let $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$. For any $M \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$, there exist unique $\{k_{\lambda}(M) \in \mathbb{Z}_{\geq 0} : \lambda \in \mathfrak{h}_m^*, \lambda_{\geq 1} = \mu\}$, which we refer to as composition multiplicities, such that

$$\operatorname{ch}(M) = \sum k_{\lambda}(M) \operatorname{ch}(L(\lambda, \mu))$$

Proof. We first show uniqueness. Suppose we have

$$\operatorname{ch}(M) = \sum_{\lambda \in \mathfrak{h}^*} a_{\lambda} \operatorname{ch}(L(\lambda, \mu)) = \sum_{\lambda \in \mathfrak{h}^*} b_{\lambda} \operatorname{ch}(L(\lambda, \mu))$$

for some $a_{\lambda}, b_{\lambda} \in \mathbb{Z}_{\geq 0}$, and $a_{\lambda} \neq b_{\lambda}$ for some $\lambda \in \mathfrak{h}^*$. Then we let $X = \{\lambda \in \mathfrak{h}^* : a_{\lambda} > b_{\lambda}\}$, $Y = \operatorname{supp}(M) \setminus X$, and

$$\chi = \sum_{\lambda \in X} (a_{\lambda} - b_{\lambda}) \operatorname{ch}(L(\lambda, \mu)) = \sum_{\lambda \in Y} (b_{\lambda} - a_{\lambda}) \operatorname{ch}(L(\lambda, \mu))$$

Since $a_{\lambda} \neq b_{\lambda}$ for some λ , we have that $\chi \neq 0$. Let $\nu \in \mathfrak{h}^*$ be such that $\chi(\nu) \neq 0$ but $\chi(\nu') = 0$ for all $\nu' \geq \nu$. Now, if $\nu \in X$, then the coefficient of $\operatorname{ch}(L(\nu,\mu))$ in the second sum must be 0, and since $\chi(\nu') = 0$ whenever $\nu' \geq \nu$, so must the coefficients of $L(\nu',\mu)$ for $\nu' \geq \nu$. Hence $\chi(\nu) = 0$ giving a contradiction. If instead $\nu \in Y$ we obtain contradiction by an identical argument.

To show existence of the $k_{\lambda}(M)$, we use induction on $n = \sum_{\lambda' \geq \lambda} \dim(M^{\lambda'})$, which is finite by Lemma 3.1.3. If n = 0, then in particular $M^{\lambda} = 0$, so we must have $k_{\lambda}(M) = 0$.

Now, let Γ be the set of non-negative integer sums of positive roots, and choose $\nu \in \lambda + \Gamma$ such that ν is maximal subject to the condition that $M^{\nu} \neq 0$. Then there must exist a highest weight vector of weight ν in M, generating a highest weight submodule K of M. Consider the quotient map $M(\nu,\mu) \to K$, and let $K' \subseteq K$ be the image of the unique maximal submodule $N(\nu,\mu) \subseteq M(\nu,\mu)$ under this quotient map. Both $\sum_{\lambda' \geq \lambda} \dim(M/K)^{\lambda'}$ and $\sum_{\lambda' \geq \lambda} \dim(K')^{\lambda'}$ are < n, so we have already constructed $k_{\lambda}(M/K)$ and $k_{\lambda}(K')$. We then set:

$$k_{\lambda}(M) = k_{\lambda}(M/K) + k_{\lambda}(K') \text{ if } \nu \neq \lambda$$

 $k_{\lambda}(M) = k_{\lambda}(M/K) + k_{\lambda}(K') + 1 \text{ if } \nu = \lambda$

We now give another interpretation of these quantities $k_{\lambda}(M)$ which is closer to the notion of composition multiplicities in Artinian categories.

Lemma 3.4.2. Let $\mu \in (\mathfrak{h}_n^{\geq 1})^*$ and $M \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$. Then there exists $\mathcal{I} \subseteq (\mathbb{Z}_{\geq 0})^m$ and a descending filtration of M indexed by \mathcal{I} with the lexicographic ordering, such that:

- (a) If $(i_1, \ldots, i_{k-1}, i_k, \ldots, i_m) \in \mathcal{I}$ with $i_k > 0$ then $(i_1, \ldots, i_{k-1}, i_k 1, 0, \ldots, 0) \in \mathcal{I}$.
- (b) Each quotient $M_{(i_1,...,i_m-1)}/M_{(i_1,...,i_m)}$ is simple.
- (c) The intersection of all the $M_{(i_1,...,i_m)}$ is trival.

Furthermore, for any such filtration $L(\lambda)$ appears as a quotient $M_{(i_1,...,i_{m-1})}/M_{(i_1,...,i_m)}$ precisely $k_{\lambda}(M)$ times.

Proof. First observe that by Theorems 3.2.1 and 3.3.11 we may reduce to the case $\mu(\mathfrak{h}_m^{(m)}) = 0$. Suppose m = 1. In this case M has a filtration $M \supseteq \mathfrak{g}_1^{(1)} M \supseteq (\mathfrak{g}_1^{(1)})^2 M \supseteq \cdots$.

Each quotient $(\mathfrak{g}_1^{(1)})^i M/(\mathfrak{g}_1^{(1)})^{i+1} M$ lies in BGG category \mathcal{O} for \mathfrak{g} and hence has finite length, so this may be refined to a filtration satisfying (a) and (b).

For m > 1 consider the filtration $M \supseteq \mathfrak{g}_m^{(m)} M \supseteq (\mathfrak{g}_m^{(m)})^2 M \supseteq \cdots$. Each quotient $(\mathfrak{g}_m^{(m)})^i M/(\mathfrak{g}_m^{(m)})^{i+1} M$ lies in the category $\mathcal{O}^{(\mu)}(\mathfrak{g}_{m-1})$, where we identify μ with an element of $(\mathfrak{h}_{m-1}^{\geq 1})^*$ in the obvious manner, so we may set $M_{(i,0,\dots,0)} = (\mathfrak{g}_m^{(m)})^i M$ and argue by induction that this may be refined to a filtration satisfying (a) and (b).

To show this filtration satisfies (c), it is enough to show that whenever $\mu(\mathfrak{h}_m^{(m)}) = 0$ every $M \in \mathcal{O}^{(\mu)}(\mathfrak{g}_m)$ has the property that $\bigcap_{i\geq 0}(\mathfrak{g}_m^{(m)})^i M = 0$. By considering weight spaces, this property is preserved by taking quotients and extensions, so it is enough to verify this in the case where $M = M(\lambda)$ is a Verma module. Since $\lambda_m = \mu_m = 0$ this Verma module is graded: we place a grading on $U(\mathfrak{g}_m)$ by setting, for $x \in \mathfrak{g}$,

$$\deg x_i = \begin{cases} 0 & \text{if } i = 0, \dots, m - 1, \\ 1 & \text{if } i = m. \end{cases}$$

We transfer the induced grading on $U(\mathfrak{n}_m^-)$ to $M(\lambda)$ via the isomorphism of \mathfrak{n}_m^- -modules $U(\mathfrak{n}_m^-) \cong M(\lambda)$. Now $M(\lambda)$ is a positively graded \mathfrak{g}_m -module and $\bigcap_{i\geq 0} (\mathfrak{g}_m^{(m)})^i M(\lambda)$ is contained in the intersection of all graded components, which is zero.

We claim that $(\mathfrak{g}_m^{(m)})^i M(\lambda, \mu_1, \dots, \mu_{m-1}, 0) = \{u(\mathbf{v_0}, \dots, \mathbf{v_m}) \otimes 1 : |\mathbf{v_m}| \geq i\}$, where $u(\mathbf{v_0}, \dots, \mathbf{v_m}) = \prod f_{\alpha,0}^{v_{0_{\alpha}}} \cdots \prod f_{\alpha,n}^{v_{m_{\alpha}}} \text{ for } \mathbf{v_0}, \dots, \mathbf{v_m} \in \mathbb{Z}_{\geq 0}^{\Phi^+}$, which immediately implies that

$$\bigcap_{i\geq 0} (\mathfrak{g}_m^{(m)})^i M(\lambda, \mu_1, \dots, \mu_{m-1}, 0) = 0.$$

We prove this claim by induction; it clearly holds when i = 0. Suppose

$$(\mathfrak{g}_m^{(m)})^i M(\lambda, \mu_1, \dots, \mu_{m-1}, 0) = \{ u(\mathbf{v_0}, \dots, \mathbf{v_n}) \otimes 1 : |\mathbf{v_m}| \ge i \}.$$

Then $(\mathfrak{g}_m^{(m)})^{i+1}M(\lambda,\mu_1,\ldots,\mu_{m-1},0)$ is spanned by elements of the form $x_m \cdot u(\mathbf{v_0},\ldots,\mathbf{v_n}) \otimes 1$, where $x = e_{\alpha}, f_{\alpha}$, or h_{α} . But $x_m \cdot u(\mathbf{v_0},\ldots,\mathbf{v_n}) \otimes 1 = f_{\alpha_1,0}x_m u(\mathbf{v_0}',\ldots,\mathbf{v_n}) \otimes 1 + 1$

 $[x, f_{\alpha_1}]_n u(\mathbf{v_0}', \dots, \mathbf{v_m}) \otimes 1$, where $|\mathbf{v_0}'| = |\mathbf{v_0}| - 1$, so by induction on $|\mathbf{v_0}|$ we may assume that $\mathbf{v_0} = \mathbf{0}$. Now x_m commutes with $u(\mathbf{0}, \mathbf{v_1}, \dots, \mathbf{v_m})$, so if $x = e_\alpha$ or h_α then $x_m \cdot u(\mathbf{0}, \dots, \mathbf{v_m}) \otimes 1 = 0$, while if $x = f_\alpha$ then $x_m \cdot u(\mathbf{0}, \dots, \mathbf{v_m}) \otimes 1$ lies in $\{u(\mathbf{v_0}, \dots, \mathbf{v_m}) \otimes 1 : |\mathbf{v_m}| \geq i + 1\}$. Hence the claim holds.

Finally, we observe that $\operatorname{ch}(M) = \sum_{(i_0,\dots,i_m)\in\mathcal{I}, i_m>0} \operatorname{ch}(M_{i_0,\dots,i_m-1}/M_{i_0,\dots,i_m})$, so by the uniqueness part of Lemma 3.4.1 the final part of the Lemma holds.

This result justifies the use of the terminology composition multiplicities, for $k_{\lambda}(M)$. From now on we use the notation $[M:L(\lambda,\mu)]:=k_{\lambda}(M)$ for $M\in\mathcal{O}^{(\mu)}$.

Corollary 3.4.3. The parabolic induction and invariants functors and the twisting functors T_{α} , G_{α} preserve composition multiplicities.

Proof. Let $M = M_0 \supseteq M_1 \supseteq M_2 \supseteq \cdots$ be a filtration of M of the form described in Lemma 3.4.2, and let F be T_{α} , G_{α} , or a parabolic induction or invariants functor. Then $F(M) = F(M_0) \supseteq F(M_1) \supseteq F(M_2) \supseteq \cdots$ is also a filtration of this form, and for all i we have $F(M_i)/F(M_{i+1}) \cong F(M_i/M_{i+1})$, so $[M:L(\lambda,\mu)] = [F(M):F(L(\lambda,\mu))]$.

3.4.2 Computation of composition multiplicities of Verma modules

We now wish to compute the composition multiplicities $[M(\lambda) : L(\nu)]$. Thanks to Lemma 3.1.7 and Lemma 3.4.2 we know that $[M(\lambda) : L(\nu)] = 0$ unless $\lambda_m = \nu_m$ and so, using a combination of twisting functors and parabolic induction functors, we can reduce to the case $\lambda_m = \nu_m = 0$.

Let $p: \mathbb{Z}_{\geq 0}\Phi^+ \to \mathbb{Z}_{\geq 0}$ be Kostant's partition function. For $\alpha \in \mathbb{Z}\Phi$ we write $\alpha_0 \in \mathfrak{h}_{m-1}^*$ to be the element satisfying $\alpha_0(h_i) = 0$ for i > 0 and $\alpha_0(h_0) = \alpha(h)$, where $h \in \mathfrak{h}$. Also, for any $\lambda \in \mathfrak{h}_m^*$ we identify $\lambda_{\leq m-1}$ with an element of \mathfrak{h}_{m-1}^* in the obvious manner.

Lemma 3.4.4. If $M(\lambda_{\leq m-1} - \alpha_0)(\mathfrak{g}_{m-1})$ is the Verma module of weight $\lambda_{\leq m-1} - \alpha_0$ for

 \mathfrak{g}_{m-1} then

$$\operatorname{ch} M(\lambda) = \sum_{\alpha \in \mathbb{Z}_{>10}\Phi^+} p(\alpha) \operatorname{ch}(M(\lambda_{\leq m-1} - \alpha_0)(\mathfrak{g}_{m-1})).$$

Proof. Note that ch M can be defined for any semisimple \mathfrak{h} -module M that has finite-dimensional weight spaces. Let \mathbb{C}_{λ_0} be the \mathfrak{h} -module of weight λ_0 . Since $M(\lambda) \cong U(\mathfrak{n}_m^-) \otimes \mathbb{C}_{\lambda_0}$ as \mathfrak{h} -modules the lemma follows from the facts:

- (i) The character of $U(\mathfrak{n}_m^-)$ is equal to the character of $S(\mathfrak{n}_m^-)$.
- (ii) $S(\mathfrak{n}_m^-)$ is a free module over $S(\mathfrak{n}_{m-1}^-)$ and, for $\alpha \in \mathbb{Z}_{\geq 0}\Phi^+$, there are $p(\alpha)$ basis vectors of weight $-\alpha$.

Corollary 3.4.5. If $\lambda, \nu \in \mathfrak{h}_m^*$ satisfy $\lambda_m = \nu_m = 0$, and $\lambda_{\geq 1} = \nu_{\geq 1}$, and $L(\nu_{\leq m-1})(\mathfrak{g}_{m-1})$ denotes the simple \mathfrak{g}_{m-1} -module of highest weight $\nu_{\leq m-1} \in \mathfrak{h}_m^*$ then

$$[M(\lambda):L(\nu)] = \sum_{\alpha \in \mathbb{Z}_{\geq 0}\Phi^+} p(\alpha)[M(\lambda_{\leq m-1} - \alpha_0)(\mathfrak{g}_{m-1}):L(\nu_{\leq m-1})(\mathfrak{g}_{m-1})].$$

Proof. By Lemma 3.1.10 we have $\operatorname{ch} L(\nu) = \operatorname{ch} L(\nu_{\leq m-1})$ for all $\nu \in \mathfrak{h}_m^*$ satisfying $\nu_m = 0$, and so the claim follows from Lemma 3.4.4.

We now introduce the notation $\mu_m \in \mathfrak{h}^*$, and write $\mathfrak{g}_{m-1}^{\mu_m} := (\mathfrak{g}^{\mu_m})_{m-1} = (\mathfrak{g}_{m-1})^{\mu_m}$.

Corollary 3.4.6. Let $\mu \in (\mathfrak{h}_m^{(\geq 1)})^*$ such that \mathfrak{g}^{μ_m} is the Levi factor of a standard parabolic. Then for any $\lambda, \nu \in \mathfrak{h}_m^*$ such that $\lambda_{\geq 1} = \nu_{\geq 1} = \mu$ we have

$$[M(\lambda) : L(\nu)] = \sum_{\alpha \in \mathbb{Z}_{>0} \Phi^+} p(\alpha) [M(\lambda_{\leq m-1} - \alpha_0)(\mathfrak{g}_{m-1}^{\mu_m}) : L(\nu_{\leq m-1})(\mathfrak{g}_{m-1}^{\mu_m})].$$

Proof. It can be verified that the functors in Lemma 3.1.11 and Theorem 3.2.1 send highest weight modules to highest weight modules of the corresponding weight. The result then follows from said Lemma and Theorem along with Lemma 3.4.2, and Corollary 3.4.5. \Box

To complete the computation of composition multiplicities for all Verma modules we must show that for any Jordan block $\mathcal{O}^{(\mu)}$, we can apply twisting functors to obtain a

Jordan block $\mathcal{O}^{(\mu')}$ such that $\mathfrak{g}^{\mu'}$ is the Levi factor of a standard parabolic. This is possible thanks to the following result:

Proposition 3.4.7. Let $\mu \in \mathfrak{h}^*$. Then there exists $w \in W$ and a Levi factor \mathfrak{l} of a standard parabolic \mathfrak{p} such that $\mathfrak{g}^{w(\mu)} = \mathfrak{l}$. Furthermore, if w is chosen to be of minimal length subject to this condition and $w = s_{\alpha_n} s_{\alpha_{n-1}} \dots s_{\alpha_1}$ is a reduced expression for w, then for each $1 \leq i \leq n$, we have $((s_{\alpha_{i-1}} \dots s_{\alpha_1})\mu)(h_{\alpha_i}) \neq 0$.

Proof. By [12, Lemma 3.8.1], we can certainly find $w = s_{\alpha_n} s_{\alpha_{n-1}} \dots s_{\alpha_1}$ and \mathfrak{l} the Levi factor of a standard parabolic subalgebra satisfying $\mathfrak{g}^{w(\mu)} = \mathfrak{l}$. Now, suppose that w has minimal length such that $g^{w(\mu)} = \mathfrak{l}$ and that for some i we have $((s_{\alpha_{i-1}} \cdots s_{\alpha_1})\mu)(h_{\alpha_i}) = 0$. In general, if $\mu(h_{\alpha}) = 0$ for some $\alpha \in \Phi$ then $s_{\alpha}(\mu) = \mu - \mu(h_{\alpha})\alpha = \mu$, so in particular, if $w' := s_{\alpha_n} \dots s_{\alpha_{i+1}} s_{\alpha_{i-1}} \dots s_{\alpha_1}$, then $w(\mu) = w'(\mu)$ and so $\mathfrak{l} = \mathfrak{g}^{w(\mu)} = \mathfrak{g}^{w'(\mu)}$. But w' has shorter length than w, giving a contradiction.

We now define an action \bullet_m of W on \mathfrak{h}^* by $w \bullet_m \lambda := w(\lambda + m\rho) - m\rho$ where ρ is half the sum of the positive roots. Note this generalises the dot action of W on \mathfrak{h}^* , which corresponds to the case m = 0, and which controls the central characters of \mathfrak{g}_0 (see [23, §1.9]). We remark that this m-dot action first appeared in the work of Geoffriau [19] on the centre of the enveloping algebra of \mathfrak{g}_m .

If α is any simple root, then since $s_{\alpha}(\alpha) = -\alpha$ and s_{α} permutes the other positive roots we have that $s_{\alpha}(\rho) = \rho - \alpha$. We then have:

$$s_{\alpha} \bullet_{m} \lambda = s_{\alpha}(\lambda + m\rho) - m\rho$$
$$= s_{\alpha}(\lambda) + m\rho - m\alpha - m\rho$$
$$= s_{\alpha}(\lambda) - m\alpha.$$

We extend this to a W-action on \mathfrak{h}_m^* by setting

$$(s_{\alpha} \bullet_{m} \lambda)(h_{i}) = \begin{cases} (s_{\alpha} \bullet_{m} \lambda_{i})(h_{i}) & \text{for } i = 0\\ (s_{\alpha} \lambda_{i})(h_{i}) & \text{for } i \neq 0. \end{cases}$$

By the proof of Lemma 3.3.12, the twisting functors T_{α} take highest weight modules of weight $\lambda \in \mathfrak{h}_m^*$ to highest weight modules of weight $s_{\alpha} \bullet_m \lambda$, and hence take $M(\lambda)$ to $M(s_{\alpha} \bullet_m \lambda)$ and similarly take $L(\lambda)$ to $L(s_{\alpha} \bullet_m \lambda)$. Applying Corollary 3.4.3 again we obtain the following.

Corollary 3.4.8. Let $\lambda, \nu \in \mathfrak{h}_m^*$ be such that $\lambda_{\geq 1} = \nu_{\geq 1}$. Let $w \in W$ have a reduced expression $w = s_{\alpha_k} s_{\alpha_{k-1}} \cdots s_{\alpha_1}$ for simple reflections s_{α_i} such that:

- (a) $\mathfrak{g}^{w(\mu_m)}$ is the Levi factor of a standard parabolic.
- (b) For each $1 \leq i \leq m$, we have $((s_{\alpha_{i-1}} \cdots s_{\alpha_1})\mu)(h_{\alpha_i}) \neq 0$.

Then we have

$$\begin{split} [M(\lambda):L(\nu)] &= [M(w \bullet_m \lambda):L(w \bullet_m \nu)] \\ &= \sum_{\alpha \in \mathbb{Z}_{\geq 0} \Phi^+} p(\alpha)[M((w \bullet_m \lambda)_{\leq m-1} - \alpha_0)(\mathfrak{g}_{m-1}^{w(\lambda_m)}):L((w \bullet_m \nu)_{\leq m-1})(\mathfrak{g}_{m-1}^{w(\mu_m)})]. \end{split}$$

CHAPTER 4

MODULAR REPRESENTATION THEORY OF TRUNCATED CURRENT LIE ALGEBRAS

4.1 Structure theory

We now consider truncated current Lie algebras in positive characteristic, starting with some structural results and then moving on to consider their representation theory. Fix an algebraically closed field k of characteristic p > 0. Recall from §2.4.1 the definition of a restricted Lie algebra, that is, a Lie algebra \mathfrak{g} equipped with a map $(\bullet)^{[p]}: \mathfrak{g} \to \mathfrak{g}$ satisfying certain conditions. If \mathfrak{g} is a restricted Lie algebra, then the corresponding truncated current Lie algebra \mathfrak{g}_m has a natural restricted structure given by:

$$(xt^i)^{[p]} = x^{[p]}t^{ip}$$

for $x \in \mathfrak{g}$ and $0 \le i \le m$. If \mathfrak{g} is the Lie algebra of an algebraic group G this restricted structure can be also be obtained from the group G_m described in Theorem 2.5.2. We abuse notation slightly and write $(\bullet)^{[p]}$ for both the p^{th} power map on both \mathfrak{g} and on \mathfrak{g}_m . From now on, we fix a standard reductive group G with Lie algebra \mathfrak{g} which carries a natural restricted structure (see §2.4.2).

Recall from §2.5 that there is a non-degenerate G_m -invariant symmetric associative

form on \mathfrak{g}_m defined by

$$\kappa_m : \mathfrak{g}_m \times \mathfrak{g}_m \longrightarrow \mathbb{k},$$

$$xt^i, yt^j \longmapsto \delta_{i+j,m}\kappa(x,y).$$

where $\kappa : \mathfrak{g} \times \mathfrak{g} \to \mathbb{R}$ is the nondegenerate form from axiom (H3) of the standard hypotheses (see §2.4.2). Using this non-degenerate pairing we can define an isomorphism of $U(\mathfrak{g}_m)$ -modules:

$$\hat{\kappa}_m : \mathfrak{g}_m \longrightarrow \mathfrak{g}_m^*$$

$$x \longmapsto \kappa_m(x,\cdot).$$

Note that this isomorphism preserves the gradings on these modules: it sends homogeneous elements $x \in \mathfrak{g}_m^{(i)}$ to linear functions which vanish on $\bigoplus_{j \neq m-i} \mathfrak{g}_m^{(j)}$. We call such elements of \mathfrak{g}^* homogeneous with support in degree m-i.

Using Definition 2.4.3, we obtain from the p^{th} power map $(\bullet)^{[p]}$ a notion of semisimple and nilpotent elements. Our first step in studying the representation theory of \mathfrak{g}_m is to classify such elements. We start with the semisimple elements:

Proposition 4.1.1. Every semisimple element of \mathfrak{g}_m is conjugate under G_m to a semisimple element of $\mathfrak{g}_m^{(0)}$.

Proof. By [21, Corollary 15.3], since \mathfrak{g}_m is the Lie algebra of the algebraic group G_m , any two Cartan subalgebras of \mathfrak{g}_m are conjugate under the adjoint action of G_m . Furthermore, observe that \mathfrak{h}_m is one such Cartan subalgebra; it is nilpotent, and since \mathfrak{h} is self-normalising in \mathfrak{g} it follows that \mathfrak{h}_m is self-normalising in \mathfrak{g}_m . Now, any semisimple element s is contained in some maximal torus, whose centraliser is a Cartan subalgebra (see [50, 2.4.2]) and so s is conjugate to an element $\sum_{i=0}^m h_i t^i$ of \mathfrak{h}_m . But since \mathfrak{h} is abelian, we see that the Jordan decomposition of $\sum_{i=0}^m h_i t^i$ must be $h_0 + \sum_{i=1}^m h_i t^i$ (using the fact that $\mathfrak{g}_m^{(\geq 1)}$ consists of nilpotent elements). Since this is conjugate to the semisimple element s, the nilpotent part of this decomposition must be 0 and so we see that in fact we

must have $h_i = 0$ for i > 0. Hence s is conjugate to h_0 which is a semisimple element of $\mathfrak{g}_m^{(0)}$.

Proposition 4.1.2. The nilpotent elements of \mathfrak{g}_m are precisely those elements of the form

$$x_0 + \sum_{i=1}^m x_i t^i$$

where x_0 is a nilpotent element of \mathfrak{g} and $x_1, \ldots, x_m \in \mathfrak{g}$.

Proof. Let $x \in \mathfrak{g}$ be of the form above. Using the axioms of a restricted Lie algebra inductively, along with the fact that $\mathfrak{g}_m^{(\geq 1)}$ is a nilpotent ideal we see that $x^{[p]^i} = 0$ for sufficiently large i, so x is nilpotent.

Now suppose that x is not of the form above and assume for a contradiction that x is nilpotent. Then writing $x = \sum_{i=0}^m x_i t^i$ we see that $x_0 = x_0^s + x_0^n$ is the Jordan decomposition of x_0 in $\mathfrak{g}_m^{(0)} \cong \mathfrak{g}$ with $x_0^s \neq 0$. According to [50, Theorem 2.3.4] there exists a k > 0 such that $x^{[p]^k}$ is a semisimple element, and hence we must have $x^{[p]^k} = 0$ since x is nilpotent. However since $\mathfrak{g}_m^{(\geq 1)}$ is a nilpotent ideal which is stable under the p^{th} power map it follows from the axioms for restricted Lie algebras that if we write $x^{[p]^k} = \sum_{i=0}^m x_{i,k} t^i$ then $x_0^{[p]^k} = x_{0,k}$. However $x_0^{[p]^k} \neq 0$ for any k > 0 since we assumed that x_0 was not nilpotent, giving a contradiction.

Remark 4.1.3. The isomorphism $\hat{\kappa}_m : \mathfrak{g}_m \to \mathfrak{g}_m^*$ allows us to transport the notations of semisimple and nilpotent to *p*-characters $\chi \in \mathfrak{g}_m^*$ and define the Jordan decomposition $\chi = \chi_s + \chi_n$ of χ ; we shall often make use without further comment.

4.2 Some Morita equivalences

We now proceed to prove several Morita equivalences between certain reduced enveloping algebras $U_{\chi}(\mathfrak{g}_m)$. Fix $0 \leq k \leq m$. We say that $\chi \in \mathfrak{g}^*$ is supported in degree less than or

equal to k if

$$\chi\left(\mathfrak{g}_m^{(>k)}\right) = 0.$$

Using the isomorphism $\hat{\kappa}_m$, every $\chi \in \mathfrak{g}^*$ can be expressed uniquely as $\hat{\kappa}_m(x)$ for some $x \in \mathfrak{g}$; χ is then supported in degree less than or equal to k if and only if $x \in \mathfrak{g}_m^{(\geq m-k)}$.

The following result will be useful the classification of simple $U(\mathfrak{g}_m)$ -modules.

Proposition 4.2.1. Let $\chi \in \mathfrak{g}_m^*$ be supported in degree less than or equal to k, and let $\psi = \chi|_{\mathfrak{g}_k} \in \mathfrak{g}_k^*$, where we identify \mathfrak{g}_k with $\mathfrak{g}_m^{(\leq k)} \subseteq \mathfrak{g}_m$ as vector spaces.

- (i) The full subcategory of $U_{\chi}(\mathfrak{g}_m)$ -mod whose objects are the modules annihilated by $\mathfrak{g}_m^{(>k)}$ is equivalent to $U_{\psi}(\mathfrak{g}_k)$ -mod. This subcategory contains all simple $U_{\chi}(\mathfrak{g}_m)$ -modules.
- (ii) $\dim \mathfrak{g}_m \dim \mathfrak{g}_m^{\chi} = \dim \mathfrak{g}_k \dim \mathfrak{g}_k^{\psi}$.

Proof. Let I_k be the ideal of $U_{\chi}(\mathfrak{g}_m)$ generated by $\mathfrak{g}_m^{(>k)}$. Using the PBW theorem for reduced enveloping algebras, we observe that the map $\mathfrak{g}_m \to U_{\chi}(\mathfrak{g}_m)/I_k$ factors through $\mathfrak{g}_m \to \mathfrak{g}_k$ and induces an isomorphism $U_{\psi}(\mathfrak{g}_k) \cong U_{\chi}(\mathfrak{g}_m)/I_k$. Furthermore, I_k is generated by nilpotent elements and so since $U_{\chi}(\mathfrak{g}_m)$ is Artinian I_k is contained in the Jacobson radical by [35, Theorem 0.1.12]. Hence every simple $U_{\chi}(\mathfrak{g}_m)$ -module is annihilated by I_k , completing the proof of part (i).

Now observe that $\mathfrak{g}_m^{(>k)}$ is an ideal of \mathfrak{g}_m on which χ is 0, so $\mathfrak{g}_m^{(>k)} \subseteq \mathfrak{g}_m^{\chi}$ and hence $\mathfrak{g}_m^{\chi} + \mathfrak{g}_m^{(\leq k)} = \mathfrak{g}_m$. Now $\dim(\mathfrak{g}_m) = \dim(\mathfrak{g}_m^{\chi} + \mathfrak{g}_m^{(\leq k)}) = \dim(\mathfrak{g}_m^{\chi}) + \dim(\mathfrak{g}_m^{(\leq k)}) - \dim(\mathfrak{g}_m^{\chi} \cap \mathfrak{g}_m^{(\leq k)})$, so it is enough to show that $\mathfrak{g}_m^{\chi} \cap \mathfrak{g}_m^{(\leq k)} = \mathfrak{g}_k^{\psi}$ (where we again identify \mathfrak{g}_k with $\mathfrak{g}_m^{(\leq k)}$). But this follows from the fact $\mathfrak{g}_m/\mathfrak{g}_m^{(>k)} \cong \mathfrak{g}_k$ and the definition of ψ , completing the proof. \square

We now prove a parabolic induction theorem for \mathfrak{g}_m analogous to Theorem 3.2.1, which allows us to reduce the problem of classifying $U_{\chi}(\mathfrak{g}_m)$ -modules to the case where χ is nilpotent. It is very similar in spirit and proof to a famous result of Friedlander and Parshall [17, Theorem 3.2] for reductive Lie algebras. We start with the following result, which is easy to prove but vital to our later reductions.

Lemma 4.2.2. Suppose $\chi \in \mathfrak{g}_m^*$ is such that $\chi([\mathfrak{g}_m, \mathfrak{g}_m]) = 0$. Then the categories $U_{\chi}(\mathfrak{g}_m)$ -mod and $U_0(\mathfrak{g}_m)$ -mod are equivalent.

Proof. Let $\lambda \in \mathfrak{g}_m^*$ be the unique linear function satisfying $\chi(x)^p - \chi(x^{[p]}) = \lambda(x)^p$ for all $x \in \mathfrak{g}_m$. This λ certainly exists; for all $x \in \mathfrak{g}_m$, set $\lambda(x)$ to be the unique p-th root of $\chi(x)^p - \chi(x^{[p]})$. Since $[\mathfrak{g}_m, \mathfrak{g}_m]$ is a restricted Lie subalgebra, it follows that $\lambda([\mathfrak{g}_m, \mathfrak{g}_m]) = 0$ and so λ defines a one dimensional $U(\mathfrak{g}_m)$ -module \mathbb{k}_{λ} with p-character χ . The functor $M \mapsto M \otimes_{U(\mathfrak{g}_m)} \mathbb{k}_{\lambda}$ is then an equivalence $U_0(\mathfrak{g}_m)$ -mod $\to U_{\chi}(\mathfrak{g}_m)$ -mod with quasi-inverse $M \mapsto M \otimes_{U(\mathfrak{g}_m)} \mathbb{k}_{-\lambda}$.

Suppose χ is a p-character with Jordan decomposition $\chi = \chi_s + \chi_n$. By Proposition 4.1.1, after conjugating by an element of G_m we may suppose that χ_s vanishes on $\mathfrak{h}_m^{(\leq m)} \oplus \mathfrak{n}_m^{\pm}$. The stabiliser \mathfrak{g}^{χ_s} is a Levi subalgebra of \mathfrak{g} and, since χ_s is supported on $\mathfrak{g}_m^{(m)}$, we have that $(\mathfrak{g}_m)^{\chi_s} = (\mathfrak{g}^{\chi_s})_m$ and hence we may write $\mathfrak{g}_m^{\chi_s}$ unambiguously.

Let $\mathfrak{p} = \mathfrak{g}^{\chi_s} \oplus \mathfrak{r}$ be a parabolic subalgebra of \mathfrak{g} with nilradical \mathfrak{r} . Note that $\chi(\mathfrak{r}_m) = 0$ by Lemma 2.4.4, and so $U_{\chi}(\mathfrak{g}_m^{\chi_s})$ -modules can be naturally inflated to $U_{\chi}(\mathfrak{p}_m)$ -modules by letting \mathfrak{r}_m act trivially.

Consider the functors

$$U_{\chi}(\mathfrak{g}_m) \otimes_{U_{\chi}(\mathfrak{p}_m)} (\bullet) : U_{\chi}(\mathfrak{g}_m^{\chi_s}) \operatorname{-mod} \longrightarrow U_{\chi}(\mathfrak{g}_m) \operatorname{-mod},$$

$$M \longmapsto U_{\chi}(\mathfrak{g}_m) \otimes_{U_{\chi}(\mathfrak{p}_m)} M.$$

$$(\bullet)^{\mathfrak{r}_m} : U_{\chi}(\mathfrak{g}_m) \operatorname{-mod} \longrightarrow U_{\chi}(\mathfrak{g}_m^{\chi_s}) \operatorname{-mod},$$

$$M \longmapsto M^{\mathfrak{r}_m}.$$

Theorem 4.2.3. The functors $U_{\chi}(\mathfrak{g}_m) \otimes_{U_{\chi}(\mathfrak{p}_m)} (\bullet)$ and $(\bullet)^{\mathfrak{r}_m}$ are quasi-inverse equivalences of categories.

Proof. Our approach is very similar to [24, §7.4], in which the m=0 case is proven.

We first show that each $U_{\chi}(\mathfrak{g})$ -module M is free as a $U_0(\mathfrak{r}_m)$ -module. By Proposition 2.5.14 we have $\mathcal{V}_{\mathfrak{r}_m}(M) \subseteq \mathcal{N}^{[p]}(\mathfrak{g}_m^{\chi}) \cap \mathfrak{r}_m = 0$, and hence M is projective as a $U_{\chi}(\mathfrak{r}_m)$ -

module. Furthermore since $\chi([\mathfrak{r}_m,\mathfrak{r}_m])=\chi(\mathfrak{r}_m^{[p]})=0$, the argument from [24, Corollary 7.2] shows that that in fact M is free as a $U_{\chi}(\mathfrak{r}_m)$ -module.

It now follows that both functors are exact. The algebra $U_0(\mathfrak{r}_m)$ is a Frobenius algebra by [17, Proposition 1.2] and hence has a simple socle, so $\dim U_0(\mathfrak{r}_m)^{\mathfrak{r}_m} = 1$. Since $\dim(\mathfrak{r}) = \dim(\mathfrak{g}/\mathfrak{p})$, we have that $\dim M = p^{\dim \mathfrak{r}_m} \dim M^{\mathfrak{r}_m}$ and $\dim(U_{\chi}(\mathfrak{g}_m) \otimes_{U_{\chi}(\mathfrak{p}_m)} N) = p^{\dim \mathfrak{r}_m} \dim N$. In particular $\dim M = \dim(U_{\chi}(\mathfrak{g}_m) \otimes_{U_{\chi}(\mathfrak{p}_m)} M^{\mathfrak{r}_m})$ and $\dim N = \dim(U_{\chi}(\mathfrak{g}_m) \otimes_{U_{\chi}(\mathfrak{p}_m)} N)^{\mathfrak{r}_m}$.

To complete the proof we must show that the unit and counit of this adjunction are natural isomorphisms. In particular, we wish to show that the maps $\varphi_M: U_\chi(\mathfrak{g}_m) \otimes_{U_\chi(\mathfrak{p}_m)} M^{\mathfrak{r}_m} \to M$ and $\psi_N: N \to (U_\chi(\mathfrak{g}_m) \otimes_{U_\chi(\mathfrak{p}_m)} N)^{\mathfrak{r}_m}$ given by $\varphi_M(u \otimes m) = u \cdot m$ and $\psi_N(n) = 1 \otimes n$ are isomorphisms. We first observe that ψ_N is certainly injective, and so by considering dimensions is an isomorphism. Now we show that φ_M is an isomorphism for M simple: it must be surjective as it has non-zero image, and so considering dimensions once again it is an isomorphism. Since our functors are exact and these categories are all Artinian, a standard argument using the short Five Lemma shows φ_M is an isomorphism for any M.

Corollary 4.2.4. If χ is a regular semisimple p-character then

$$U_{\chi}(\mathfrak{g}_m) \cong \operatorname{Mat}_{p^{\frac{(m+1)}{2}(\dim(\mathfrak{g})-\operatorname{ind}(\mathfrak{g}))}} U_0(\mathfrak{h}_m)$$

Proof. Using [42, Theorem 2.3(ii)] and Theorem 4.2.3 we have the isomorphism $U_{\chi}(\mathfrak{g}_m) \cong \operatorname{Mat}_{p^{\frac{(m+1)}{2}}(\dim(\mathfrak{g})-\operatorname{ind}(\mathfrak{g}))} U_{\chi}(\mathfrak{h}_m)$. This requires the notion of a χ -admissible subalgebra (which we discuss later in §4.3), but in the case χ is regular semisimple the conditions for a subalgebra to be χ -admissible reduce to the subalgebra consisting of nilpotent elements. In particular, \mathfrak{n}_m is a χ -admissible subalgebra of \mathfrak{g}_m of dimension $\frac{m+1}{2}(\dim(\mathfrak{g}) - \operatorname{rank}(\mathfrak{g}))$. The equivalence described in Lemma 4.2.2 then induces an algebra isomorphism $U_0(\mathfrak{h}_m) \cong U_{\chi}(\mathfrak{h}_m)$.

We now describe the simple and projective modules over $U_0(\mathfrak{h}_m)$. Since \mathfrak{h}_m is com-

mutative we have that the simple modules are the 1-dimensional modules \mathbb{k}_{λ} for $\lambda \in \mathfrak{h}^*$ such that $\lambda(h)^p = \lambda(h^{[p]})$, where $\mathfrak{h}_m^{(0)}$ acts by λ and $\mathfrak{h}_m^{(\geq 1)}$ acts by 0. Note that there are $p^{\dim(\mathfrak{h})}$ such λ . Each simple module \mathbb{k}_{λ} then has a unique projective cover in the category $U_0(\mathfrak{h}_m)$ -mod, which we denote $Q^{\mathfrak{h}_m}(\lambda)$, and these $Q^{\mathfrak{h}_m}(\lambda)$ form a complete set of irreducible projective modules for $U_0(\mathfrak{h}_m)$.

Proposition 4.2.5. The only composition factor of $Q^{\mathfrak{h}_m}(\lambda)$ is \mathbb{k}_{λ} , which occurs with multiplicity $p^{m\dim(\mathfrak{h})}$.

Proof. Consider the module $P(\lambda) := U_0(\mathfrak{h}_m) \otimes_{U_0(\mathfrak{h}_m^{(0)})} \mathbb{k}_{\lambda}$. As a $U_0(\mathfrak{h}_m^{(\geq 1)})$ -module this is isomorphic to $U_0(\mathfrak{h}_m^{(\geq 1)})$ and so, since $\mathfrak{h}_m^{(\geq 1)}$ is p-nilpotent, $P(\lambda)$ is indecomposable by [24, Corollary 3.4]. Hence $\dim(Q^{\mathfrak{h}_m}(\lambda)) \geq \dim(P(\lambda)) = p^{m\dim(\mathfrak{h})}$. But $p^{(m+1)\dim(\mathfrak{h})} = \dim(U_0(\mathfrak{h}_m)) \geq \sum_{\lambda} \dim(Q^{\mathfrak{h}_m}(\lambda)) \geq \sum_{\lambda} \dim(P(\lambda)) = p^{\dim(\mathfrak{h})} p^{m\dim(\mathfrak{h})}$, so in fact $Q^{\mathfrak{h}_m}(\lambda) \cong P(\lambda)$. Hence $\mathfrak{h}_m^{(0)}$ acts by λ on $Q^{\mathfrak{h}_m}(\lambda)$ so the only composition factor of $Q^{\mathfrak{h}_m}(\lambda)$ is \mathbb{k}_{λ} , and since $\dim(Q^{\mathfrak{h}_m}(\lambda)) = \dim(P(\lambda)) = p^{m\dim(\mathfrak{h})}$ it must occur with multiplicity $p^{m\dim(\mathfrak{h})}$ as required.

4.3 The Kac-Weisfeiler conjectures in the truncated current case

Recall from §2.4.3 the following conjectures, originally made in [55]:

Conjecture 4.3.1 (KW1). Let \mathfrak{g} be a restricted Lie algebra and let $M(\mathfrak{g})$ be the maximal dimension among simple $U(\mathfrak{g})$ -modules. Then $M(\mathfrak{g}) = p^{\frac{1}{2}(\dim(\mathfrak{g})-\inf(\mathfrak{g}))}$.

Theorem 4.3.2 ([40], KW2). Let \mathfrak{g} be the Lie algebra of a reductive group G under the standard hypotheses. Then for any p-character χ and any simple $U_{\chi}(\mathfrak{g})$ -module M, $p^{\frac{1}{2}(\dim \mathfrak{g} - \dim \mathfrak{g}^{\chi})}$ divides $\dim M$.

Most of the known results on (KW1) mentioned in §2.4.3 do not apply to truncated current Lie algebras. The major exception is the case $\mathfrak{g} = \mathfrak{gl}_n$; here (as observed in

[57]) if e is a nilpotent element in $\mathfrak{gl}_{(m+1)n}$ corresponding to the rectangular partition $(m+1,m+1,\ldots,m+1)$ then $(\mathfrak{gl}_{(m+1)n})^e \cong (\mathfrak{gl}_n)_m$. Hence by [53, Theorem 4] (KW1) holds for $(\mathfrak{gl}_n)_m$. The result [33, Theorem 1.1] also implies that if we fix \mathfrak{g} and m, then in sufficiently large characteristic (KW1) holds for \mathfrak{g}_m , without giving an explicit bound. We give a proof of (KW1) for \mathfrak{g}_m which is valid where \mathfrak{g} the Lie algebra of a reductive group G with no assumptions beyond the standard hypotheses.

As mentioned in §2.4.3, Premet proved (KW2) in [40], subsequently giving other proofs in [42, §2.6] and [43, Theorem 5.6]. In [28], Kac conjectured that the statement (KW2) holds for all Lie algebras of algebraic groups, but very little is known about this statement outside the standard reductive case. We show that (KW2) holds for certain classes of p-characters of the truncated current Lie algebra \mathfrak{g}_m , and also give a proof for all p-characters of (\mathfrak{sl}_2) $_m$ in characteristic greater than 2.

4.3.1 The first Kac–Weisfeiler conjecture

Recall that earlier in the proof of Theorem 2.5.10 we defined $\mathfrak{h}_m^{\text{reg}} := \{ \sum x_i t^i \in \mathfrak{h}_m \mid x_0 \in \mathfrak{h}^{\text{reg}} \}.$

Proposition 4.3.3. Let $\chi = \hat{\kappa}_m(h)$ for some $h \in \mathfrak{h}_m^{\text{reg}}$. Then the algebra $U_{\chi}(\mathfrak{g}_m)$ has (up to isomorphism) precisely $p^{\text{rank}(\mathfrak{g})}$ simple modules and the same number of projective indecomposable modules. The simple modules have dimension $p^{\frac{m+1}{2}(\dim(\mathfrak{g})-\operatorname{rank}(\mathfrak{g}))}$ while the projective indecomposable modules have dimension $p^{\frac{m+1}{2}(\dim(\mathfrak{g})+\operatorname{rank}(\mathfrak{g}))-\operatorname{rank}(\mathfrak{g})}$.

Proof. Let $h = \sum_{x=0}^m x_i t^i$. Observe that the Jordan decomposition of h has semisimple part x_0 and nilpotent part $\sum_{x=1}^m x_i t^i$, so by Theorem 4.2.3 we obtain an equivalence between $U_{\chi}(\mathfrak{g}_m)$ and $U_{\chi}(\mathfrak{h}_m)$. Furthermore, by Lemma 4.2.2 together with Proposition 4.2.5 and the preceding discussion, the simple modules for $U_{\chi}(\mathfrak{h}_m)$ are 1-dimensional, the projective modules are $p^{m \operatorname{rank}(\mathfrak{g})}$ -dimensional, and there are $p^{\operatorname{rank}(\mathfrak{g})}$ of each. Now, from the proof of Theorem 4.2.3, for any $U_{\chi}(\mathfrak{h}_m)$ -module N we have $\dim(U_{\chi}(\mathfrak{g}_m) \otimes_{U_{\chi}(\mathfrak{b}_m)} N) = p^{\dim \mathfrak{b}_m} \dim N = p^{\frac{m+1}{2}(\dim(\mathfrak{g})-\operatorname{rank}(\mathfrak{g}))} \dim N$ and the desired result follows.

Corollary 4.3.4. The first Kac-Weisfeiler conjecture holds for \mathfrak{g}_m where \mathfrak{g} is the Lie algebra of a standard reductive group.

Proof. By the proof of Theorem 2.5.10, the conjugates of $\mathfrak{h}_m^{\text{reg}}$ are dense in \mathfrak{g}_m . Recall that we defined $M(\mathfrak{g}_m)$ to be the maximal dimension of a simple $U(\mathfrak{g}_m)$ -module. By [43, Proposition 4.2(1)] the set of p-characters χ such that all simple $U_{\chi}(\mathfrak{g}_m)$ -modules have dimension $M(\mathfrak{g}_m)$ is non-empty and open, and so in particular intersects the conjugates of $\hat{\kappa}_m(\mathfrak{h}_m^{\text{reg}})$ non-trivially. Now by Proposition 4.3.3, if χ is conjugate to an element of $\hat{\kappa}_m(\mathfrak{h}_m^{\text{reg}})$, then all simple $U_{\chi}(\mathfrak{g}_m)$ -modules have dimension $p^{\frac{m+1}{2}(\dim(\mathfrak{g})-\operatorname{rank}(\mathfrak{g}))}$ which is equal to $p^{\frac{1}{2}(\dim(\mathfrak{g}_m)-\operatorname{ind}(\mathfrak{g}_m))}$ by Theorem 2.5.10. Hence $M(\mathfrak{g}_m)=p^{\frac{1}{2}(\dim(\mathfrak{g}_m)-\operatorname{ind}(\mathfrak{g}_m))}$.

4.3.2 The second Kac–Weisfeiler conjecture

We now investigate the second Kac-Weisfeiler conjecture for the case of truncated currents on the Lie algebra of a general standard reductive group. In particular, although we do not show the statement holds for all p-characters, we give several families for which it does hold. Following [42, §2.3], for $\chi \in (\mathfrak{g}_m)^*$ a nilpotent p-character, we define a χ -admissible subalgebra of \mathfrak{g}_m to be a subalgebra \mathfrak{m} satisfying:

- (i) m consists of nilpotent elements.
- (ii) $\chi(\overline{[\mathfrak{m},\mathfrak{m}]})=0$, where \mathfrak{a} denotes the [p]-closure of a subalgebra $\mathfrak{a}\subseteq\mathfrak{g}$.
- (iii) $\mathcal{V}_{\mathfrak{g}_m}(\chi) \cap \mathfrak{m} = 0.$

By [42, Theorem 2.3(ii)], if there exists a χ -admissible subalgebra \mathfrak{m} such that $\dim(\mathfrak{m}) = \frac{1}{2}(\dim \mathfrak{g}_m - \dim \mathfrak{g}_m^{\chi})$ then (KW2) holds for χ .

Proposition 4.3.5. Let \mathfrak{g} be the Lie algebra of a standard reductive group. Then for \mathfrak{g}_m , we have:

(i) (KW2) holds for all homogeneous p-characters.

(ii) (KW2) holds for all semisimple p-characters.

Proof. These two claims follow directly from Proposition 4.2.1 and Theorem 4.2.3 respectively, since these results allow to reduce to the $\chi = 0$ case and in this $\mathfrak{m} = 0$ is clearly a χ -admissible subalgebra of the correct dimension.

Proposition 4.3.6. The second Kac-Weisfeiler conjecture holds for $(\mathfrak{sl}_2)_m$ provided p > 2.

Proof. First observe that by Theorem 4.2.3 we may reduce to the reductive case. Let $x = \sum_{j=i}^m x_j t^j$ where $x_i \neq 0$ be a nilpotent element of $(\mathfrak{sl}_2)_m$, and let $\chi = \hat{\kappa}_m(x)$. Choose a nilpotent $y_0 \in \mathfrak{sl}_2$ such that $[x_i, y_0] \neq 0$; this is always possible since p > 2. We claim that $\mathfrak{n}_l = \operatorname{span}\{y_0 t^k : 0 \leq k \leq l\}$ is χ -admissible for any $l \leq m-i$ and that $\dim((\mathfrak{sl}_2)_m)^\chi) \geq 2i + m + 1$, which by our earlier discussion implies that (KW2) holds for $(\mathfrak{sl}_2)_m$. By Proposition 4.1.2 we have that condition (i) holds, and since \mathfrak{n} is commutative condition (ii) also holds. Now, by Proposition 2.5.14, we have $\mathcal{V}_{(\mathfrak{sl}_2)_m}(\chi) = \mathcal{N}^{[p]}(((\mathfrak{sl}_2)_m)^\chi)$ so to show condition (iii) it suffices to show that $\sum_{k=0}^{m-i} \lambda_k y_0 t^k \notin ((\mathfrak{sl}_2)_m)^\chi$ for any scalars $\lambda_k \in \mathbb{k}$. But $[x, y_0 t^k] = [x_i, y_0] t^{i+k} + \sum_{j=i+1}^m [x_j, y_0] t^{j+k} \neq 0$, so by considering the smallest k such that $\lambda_k \neq 0$ we see this holds. Hence \mathfrak{n}_l is indeed χ -admissible for any $l \leq m-i$. Finally, observe that

$$((\mathfrak{sl}_2)_m)^{(\geq m-i+1)} \oplus \operatorname{span}\{\sum_{j=i}^m x_j t^{j-i+k} : 0 \leq k \leq m-i\} \subseteq ((\mathfrak{sl}_2)_m)^{\chi}$$

and so dim $((\mathfrak{sl}_2)_m)^{\chi} \geq 2i + m + 1$ as required.

4.4 Classification of simple modules

4.4.1 Baby Verma modules and their simple quotients

The aim of this section is to classify the simple $U_{\chi}(\mathfrak{g})$ -modules for p-characters χ which are homogeneous and nilpotent of standard Levi type.

In order to justify our restrictions on p-characters we make some basic observations about the role of baby Verma modules. In the case m=0 they are one of the main tools for explicit construction of simple modules. Our approach here is to pick a Borel subalgebra $\mathfrak{b} \subseteq \mathfrak{g}$, inflate one dimensional $U_{\chi}(\mathfrak{h}_m)$ -modules to $U_{\chi}(\mathfrak{b}_m)$ -modules and then induce. Note that this is only possible if we can choose a G_m -conjugate of χ which vanishes on \mathfrak{n}_m where $\mathfrak{n} \subseteq \mathfrak{b}$ is the radical. When m>0 this is not always the case; consider, for example, the case where $\mathfrak{g}=\mathfrak{sl}_2$ and m=1, and let $\{e,h,f\}\subseteq\mathfrak{sl}_2$ be the standard basis. Then for $\chi=\hat{\kappa}_m(e+ft)$ there is no choice of \mathfrak{b} such that $\chi(\mathfrak{n}_m)=0$.

Now assume there exists a choice of Borel subalgebra \mathfrak{b} such that $\chi(\mathfrak{n}_m) = 0$ and fix such a choice. Then we can define the *baby Verma modules*. Let $\mathfrak{b} = \mathfrak{h} \oplus \mathfrak{n}$ where \mathfrak{h} a maximal torus of \mathfrak{g} . Define

$$\Lambda_{\chi} = \{\lambda \in \mathfrak{h}_m^* : \lambda(ht^i)^p - \lambda((ht^i)^{[p]}) - \chi(ht^i)^p = 0 : h \in \mathfrak{h}, 0 \le i \le m\} \subseteq \mathfrak{h}_m^*$$

Then for any $\lambda \in \Lambda_{\chi}$, the baby Verma module $M_{\chi}(\lambda)$ for $U_{\chi}(\mathfrak{g}_m)$ is given by

$$M_{\chi}(\lambda) := U_{\chi}(\mathfrak{g}_m) \otimes_{U_{\chi}(\mathfrak{b}_m)} \mathbb{k}_{\lambda}$$

where k_{λ} is the 1-dimensional module on which \mathfrak{h}_m acts by λ and \mathfrak{n}_m acts by 0.

Lemma 4.4.1. Let \mathfrak{b} be a Borel subalgebra of \mathfrak{g} with nilradical \mathfrak{n} , and let $\chi \in \mathfrak{g}^*$ be such that $\chi(\mathfrak{n}) = 0$. Then every simple $U_{\chi}(\mathfrak{g}_m)$ -module is isomorphic to a quotient of $M_{\chi}(\lambda)$ for some $\lambda \in \Lambda_{\chi}$.

Proof. The proof of the analogous statement [24, Proposition 6.7] is still valid here, replacing $\mathfrak{n}, \mathfrak{b}$ and \mathfrak{n}^- with $\mathfrak{n}_m, \mathfrak{b}_m$ and \mathfrak{n}_m^- respectively.

We consider the case where m > 0 and χ is a nilpotent homogeneous p-character of degree k, i.e. where $\chi(\bigoplus_{i \neq k} \mathfrak{g}(i)) = 0$. In this case, we can always find a Borel \mathfrak{b} such that $\chi(\mathfrak{b}_m) = 0$. In fact by Proposition 4.2.1, without loss of generality we can assume that

k = m so let $\chi = \hat{\kappa}_m(x)$ for some $x \in \mathfrak{g} \cong \mathfrak{g}_m^{(0)}$. Then by Theorem 4.2.3 and Proposition 4.1.2 we can also assume that x is a nilpotent element of \mathfrak{g} .

We focus in particular on the case where $\chi = \hat{\kappa}_m(e)$ for some $e \in \mathfrak{g}$ of standard Levi type. This means e is a regular nilpotent element of some Levi subalgebra of \mathfrak{g} , see [24, §10] for more detail.

For the rest of this section, we fix such an $e \in \mathfrak{g}$ and $\chi = \hat{\kappa}_m(e)$. We can choose a maximal torus $\mathfrak{h} = \text{Lie}(H) \subseteq \mathfrak{g}$, let Φ be the root system with respect to H, and choose a set of simple roots Δ such that the minimal Levi subalgebra containing e has root system generated by a subset of Δ .

Pick root vectors e_{α} for $\alpha \in \Phi$ such that $e = \sum_{\alpha \in I} e_{\alpha}$ for some $I \subseteq \Delta$. Let $\mathfrak{b}^+ \subseteq \mathfrak{g}$ be the Borel subalgebra corresponding to this choice of simple roots, \mathfrak{b}^- the opposite Borel corresponding to $-\Delta$, and \mathfrak{n}^- , \mathfrak{n} their respective nilradicals.

Finally, we let \mathfrak{g}_I be the Levi subalgebra $\mathfrak{h} \oplus \bigoplus_{\alpha \in \mathbb{Z}I} \mathfrak{g}_{\alpha}$ of \mathfrak{g} corresponding to the subset $I \subseteq \Delta$, and let \mathfrak{r} and \mathfrak{r}^- be the nilradicals of the parabolic subalgebras $\mathfrak{g}_I + \mathfrak{b}$ and $\mathfrak{g}_I + \mathfrak{b}^-$ respectively.

Proposition 4.4.2. Let $\chi = \hat{\kappa}_m(e)$ for some $e \in \mathfrak{g}$ in standard Levi type, and fix a toral basis $\{h_1, \ldots, h_r\}$ of \mathfrak{h} . Then $\Lambda_{\chi} = \{\lambda \in \mathfrak{h}^* : \lambda(\mathfrak{h}_m^{(\geq 1)}) = 0, \lambda(h_j) \in \mathbb{F}_p \subset \mathbb{k}\}.$

Proof. Consider first the equation $\lambda(ht^i)^p - \lambda((ht^i)^{[p]}) - \chi(ht^i)^p = 0$ in the case i = m. Here there is a unique solution $\lambda(ht^m) = \chi(ht^m) = 0$ since $(ht^i)^{[p]} = 0$. We then see inductively for $i = m - 1, m - 2, \ldots, 1$ that $\lambda((ht^i)^{[p]}) = 0$ and so again $\lambda(ht^i) = \chi(ht^i) = 0$. Now, if $h \in \mathfrak{h}$ is such that $h^{[p]} = h$ then since $\chi(h) = 0$ we have that $\lambda(h)^p - \lambda(h^{[p]}) = \lambda(h)^p - \lambda(h) = 0$. Hence $\lambda(h_j)$ can be chosen to be any value in \mathbb{F}_p for all $1 \leq j \leq r$. \square

In light of this, from here on we regard elements of Λ_{χ} as elements of \mathfrak{h}^* rather than of \mathfrak{h}_m^* . For χ in standard Levi form, we have the following result analogous to [24, Proposition 10.2] which gives a relationship between the baby Verma modules and the simple $U_{\chi}(\mathfrak{g}_m)$ -modules.

Lemma 4.4.3. Let $\chi \in \mathfrak{g}^*$ be nilpotent and in standard Levi form. Then each $M_{\chi}(\lambda)$ has a unique maximal submodule and hence a unique simple quotient $L_{\chi}(\lambda)$.

Proof. Just as for Lemma 4.4.1, the proof of the analogous statement [24, Proposition 10.2] is still valid here, replacing \mathfrak{n} , \mathfrak{b} and \mathfrak{n}^- with \mathfrak{n}_m , \mathfrak{b}_m and \mathfrak{n}^- respectively.

4.4.2 Simple modules

By Theorem 4.2.3, along with Lemmas 4.4.1 and 4.4.3, to classify the simple $U_{\chi}(\mathfrak{g}_m)$ modules for χ in standard Levi form supported in degree m it suffices to determine when $L_{\chi}(\lambda) \cong L_{\chi}(\mu)$ for $\lambda, \mu \in \Lambda_{\chi}$. We first consider the case $I = \Delta$, i.e. when e is a regular
nilpotent element of \mathfrak{g} and $\mathfrak{g}_I = \mathfrak{g}$.

Proposition 4.4.4. Let $\chi = \hat{\kappa}_m(e)$ for $e \in \mathfrak{g}$ regular nilpotent. Then $M_{\chi}(\lambda)$ is simple for all $\lambda \in \Lambda_{\chi}$.

Proof. Letting $\mathfrak{m} = \mathfrak{n}_m^-$ and observing that $\mathfrak{g}_m^{\chi} \cap \mathfrak{n}_m^- = 0$, we can apply Proposition 2.5.14 to see that \mathfrak{m} is a χ -admissible subalgebra (note that χ vanishes on $[\mathfrak{m}, \mathfrak{m}]$ since we have assumed that e is in standard Levi form). It follows by the same argument as used in the second paragraph of the proof of Theorem 4.2.3 that any $U_{\chi}(\mathfrak{g}_m)$ -module is free as a $U_{\chi}(\mathfrak{m}_m)$ -module. Now let N be a $U_{\chi}(\mathfrak{g}_m)$ -submodule of $M_{\chi}(\lambda)$. By dimensional considerations N is free of rank 0 or 1 over $U_{\chi}(\mathfrak{m}_m)$, and hence N=0 or $N=M_{\chi}(\lambda)$, i.e. $M_{\chi}(\lambda)$ is simple.

Proposition 4.4.5. Let $\chi = \hat{\kappa}_m(e)$ for $e \in \mathfrak{g}$ regular nilpotent. Then $L_{\chi}(\lambda) \cong L_{\chi}(\mu)$ if and only if $\lambda|_{\mathfrak{z}(\mathfrak{g})} = \mu|_{\mathfrak{z}(\mathfrak{g})}$.

Proof. Fix some $\lambda \in \Lambda_{\chi}$ and $\alpha \in \Delta$, and consider the element $f_{\alpha}t^{m} \otimes 1_{\lambda} \in M_{\chi}(\lambda)$ for some $f_{\alpha} \in \mathfrak{g}_{-\alpha}$. This element is highest weight since if $x \in \mathfrak{n}_{m}$ then $[x, f_{\alpha}t^{m}] \in (\mathfrak{b}_{m})^{(m)}$, so $x \cdot (f_{\alpha}t^{m} \otimes 1_{\lambda}) = f_{\alpha}t^{m} \otimes (x \cdot 1_{\lambda}) + 1 \otimes ([x, f_{\alpha}t^{m}] \cdot 1_{\lambda}) = 0$. Furthermore, it generates $M_{\chi}(\lambda)$ since $(f_{\alpha}t^{m})^{p-1} \cdot (f_{\alpha}t^{m} \otimes 1_{\lambda}) = \chi(f_{\alpha}t^{m})^{p} \otimes 1_{\lambda}$ and $\chi(f_{\alpha}t^{m}) \neq 0$. Hence we have a

surjective homomorphism from $M_{\chi}(\lambda - d\alpha)$ to $M_{\chi}(\lambda)$, which by considering dimensions must in fact be an isomorphism. But we also have that $M_{\chi}(\lambda)$ is simple for all $\lambda \in \Lambda_{\chi}$ by Corollary 4.4.4, so $L_{\chi}(\lambda) = M_{\chi}(\lambda)$. Hence it suffices to show that $\operatorname{span}_{\mathbb{F}_p} \{ d\alpha : \alpha \in \Delta \} = \{ \lambda \in \Lambda_{\chi} : \lambda|_{\mathfrak{z}(\mathfrak{g})} = 0 \}$, but this can be verified by a case-by-case computation on each irreducible component of the root system Φ .

On the other hand, if $\lambda|_{\mathfrak{z}(\mathfrak{g})} \neq \mu|_{\mathfrak{z}(\mathfrak{g})}$ then $\mathfrak{z}(\mathfrak{g}_m)$ acts on $L_{\chi}(\lambda)$ and $L_{\chi}(\mu)$ by different scalars, so they have different central characters and hence must be non-isomorphic. \square

The following Lemma, analogous to a result [24, Proposition 10.7] in the reductive case, allows us to extend this classification of simple modules to p-characters $\chi = \hat{\kappa}_m(e)$ for any $e \in \mathfrak{g}$ of standard Levi type.

Lemma 4.4.6. There is a bijection between simple $U_{\chi}(\mathfrak{g}_m)$ -modules and simple $U_{\chi}((\mathfrak{g}_I)_m)$ modules sending M to $M^{\mathfrak{r}_m}$. This bijection takes the simple $U_{\chi}(\mathfrak{g}_m)$ -module $L_{\chi}(\lambda)$ to the
simple $U_{\chi}((\mathfrak{g}_I)_m)$ -module $L_{\chi}(\lambda)$.

Proof. The first part follows from general results on graded modules, namely Corollary 1.4 and Theorems 1.1 and 1.2 in [48]. To apply these results, we require only the fact that \mathfrak{r}_m and \mathfrak{r}_m^- act nilpotently on the baby Verma modules (and hence on their simple quotients) which follows from the definition of $M_{\chi}(\lambda)$ and the fact that $\chi(\mathfrak{r}_m^-) = 0$. The second part follows easily once we observe that if we take a highest weight generator of weight λ for the $U_{\chi}(\mathfrak{g}_m)$ module $L_{\chi}(\lambda)$, then this element lies in $M^{\mathfrak{r}_m}$ and generates it as a $U_{\chi}((\mathfrak{g}_I)_m)$ -module.

Theorem 4.4.7. $L_{\chi}(\lambda) \cong L_{\chi}(\mu)$ if and only if $\lambda|_{\mathfrak{z}(\mathfrak{g}_I)} = \mu|_{\mathfrak{z}(\mathfrak{g}_I)}$.

Proof. By Lemma 4.4.6, $L_{\chi}(\lambda) \cong L_{\chi}(\mu)$ if and only if the corresponding simple $U_{\chi}((\mathfrak{g}_{I})_{m})$ modules are isomorphic. But by Proposition 4.4.5 this occurs precisely when $\lambda|_{\mathfrak{z}(\mathfrak{g}_{I})} = \mu|_{\mathfrak{z}(\mathfrak{g}_{I})}$.

4.4.3 The general linear algebra

Here we explain how the results of the previous section give a complete classification of simple $U(\mathfrak{g}_m)$ -modules with homogeneous p-characters when $\mathfrak{g} = \mathfrak{gl}_n$.

Suppose $\chi \in \mathfrak{g}^*$ is homogeneous. Then by Proposition 4.2.1(i) we may assume that χ is supported in degree m. Decomposing χ into semisimple and nilpotent parts as $\chi = \chi_s + \chi_n$, we consider the stabiliser $(\mathfrak{g}_m)^{\chi_s} = (\mathfrak{g}^{\chi_s})_m$. Note that \mathfrak{g}^{χ_s} is a Levi subalgebra of \mathfrak{gl}_n , and therefore it is isomorphic to $\mathfrak{l} = \mathfrak{gl}_{n_1} \times \cdots \times \mathfrak{gl}_{n_k}$ for some $\sum_{i=1}^k n_i$ and $1 \leq k \leq n$. Furthermore $\chi([\mathfrak{l}_m, \mathfrak{l}_m]) = 0$.

The classification of simple $U_{\chi}(\mathfrak{g})$ -modules follows easily from the classification of the simple $U_{\chi|\mathfrak{gl}_{n_i}}(\mathfrak{gl}_{n_i})$ modules for i=1,...,k and so we have now reduced the problem to classifying simple $U_{\chi}(\mathfrak{g}_m)$ -modules, where the semisimple part of χ is supported on the centre of \mathfrak{g}_m . Using Lemma 4.2.2 we may assume that χ is nilpotent and supported in degree m. But every nilpotent $\chi \in (\mathfrak{gl}_n)^*$ can be placed in standard Levi form, and so the classification of simple $U_{\chi}(\mathfrak{gl}_n)$ -modules can be deduced from our earlier results.

Remark 4.4.8. This classification can even be applied in the case where the semisimple and nilpotent parts are each homogeneous but are supported in different degrees.

Remark 4.4.9. Outside type A not all nilpotent elements have standard Levi type, so the above classification breaks down. If p > 2 and \mathfrak{g} is simple of type B, C or D then the nilpotent orbits are still classified by partitions (although in type D certain partitions correspond to two distinct orbits). The partitions corresponding to standard Levi nilpotent orbits are as follows; see for example [27, §1]:

- Type B: All parts of the partition occur with even multiplicity, with the exception of one odd part which can occur with odd multiplicity.
- Type C: All parts occur with even multiplicity, with the possible exception of one even part which can occur with odd multiplicity.

• Type D: There are 2k parts which occur with even multiplicity, say $\lambda_{j_1}, ..., \lambda_{j_{2k}}$. The final two parts of the partition have sizes $2n - \sum_{i=1}^{2k} \lambda_{j_i} - 1$ and 1.

If the nilpotent part of a homogeneous p-character χ corresponds to a partition satisfying these properties, we obtain a classification of simple $U_{\chi}(\mathfrak{g})$ -modules from the above results in the same manner as in the type A case.

4.5 Cartan invariants for the restricted enveloping algebras

Recall that for any finite dimensional algebra A with simple modules L_1, \ldots, L_s there is for each L_i a unique (up to isomorphism) projective cover P_i . These P_i form a complete set of indecomposable projective A-modules. The composition multiplication of the P_i are known as the Cartan invariants of A.

In Corollary 4.2.4 and Proposition 4.2.5 we calculated the Cartan invariants of $U_{\chi}(\mathfrak{g}_m)$ whene χ is regular semisimple. In this section we deal with the opposite extreme; the case where $\chi = 0$.

Recall from earlier that we have a triangular decomposition $\mathfrak{g} = \mathfrak{n}^- \oplus \mathfrak{h} \oplus \mathfrak{n}^+$, where $\mathfrak{h} = \mathrm{Lie}(H)$ for some choice of torus $H \subseteq G_m$. We consider baby Verma modules $Z_0(\lambda)$ for the restricted enveloping algebra $U_0(\mathfrak{g}_m)$. Since $\chi = 0$ is fixed, from now on we omit the subscript in the notation for baby Verma modules and instead write $Z(\lambda)$. We also define $Z^{\mathfrak{g}}(\lambda)$ to be the baby Verma module for $U_0(\mathfrak{g})$:

$$Z^{\mathfrak{g}}(\lambda) := U_0(\mathfrak{g}) \otimes_{U_0(\mathfrak{b})} \mathbb{k}_{\lambda}$$

where \mathbb{k}_{λ} is the 1-dimensional \mathfrak{b} module on which \mathfrak{h} acts by λ and \mathfrak{n} acts by 0.

We can inflate this to a $U_0(\mathfrak{g}_m)$ -module by letting $\mathfrak{g}_m^{(\geq 1)}$ act by 0; by abuse of notation we also label this module $Z^{\mathfrak{g}}(\lambda)$. Now observe that by Proposition 4.2.1, the simple $U_0(\mathfrak{g}_m)$ -modules are just the simple modules for $U_0(\mathfrak{g})$ with $\mathfrak{g}_m^{(\geq 1)}$ acting by 0. As with

the baby Verma modules, these modules are labelled by Λ_0 , so again abusing notation we write $L(\lambda)$ for both the simple $U_0(\mathfrak{g})$ -module of weight λ and the simple $U_0(\mathfrak{g}_m)$ -module of weight λ . Here as in the previous section that we may view Λ_0 as a subset of \mathfrak{h}^* using Proposition 4.4.2.

Recall that if N is a $U_0(\mathfrak{g}_m)$ -module, then we write $[N:L(\lambda)]$ for the composition multiplicity of $L(\lambda)$ in N. Similarly, if $\{M(\lambda):\lambda\in\Lambda_0\}$ is a family of $U_0(\mathfrak{g}_m)$ -modules and N admits a filtration such that each section is isomorphic to some $M(\lambda)$, then we write $(N:M(\lambda))$ to denote the number of times $M(\lambda)$ occurs as a subquotient in the filtration. Of course, the quantities $(N:M(\lambda))$ may not be well-defined in general; however, whenever we use this notation we shall show that they are.

The following proposition is not difficult to prove and is useful in the computation of composition multiplicities.

Proposition 4.5.1. If $(N:M(\mu))$ exists, then for any $\lambda \in \Lambda_0$ we have $[N:L(\lambda)] = \sum_{\mu \in \Lambda_0} (N:M(\mu))[M(\mu):L(\lambda)].$

Proof. Let $0 = N_0 \subset N_1 \subset \cdots \subset N_k = N$ be a filtration of N in which for all $\mu \in \Lambda_0$, $M(\mu)$ appears as a quotient N_i/N_{i-1} precisely $(N:M(\mu))$ times. Then refining this filtration to a composition series we see that $[N:L(\lambda)] = \sum_{i=1}^k [N_i/N_{i-1}:L(\lambda)]$ and the desired result follows.

4.5.1 Graded $U_0(\mathfrak{g}_m)$ -modules

We now wish to compute the composition multiplicities of projective $U_0(\mathfrak{g}_m)$ -modules. One tool we will use in doing so is a grading on $U_0(\mathfrak{g}_m)$ and a category of graded $U_0(\mathfrak{g}_m)$ -modules, generalising a well-known technique in the m=0 case, see for example [24, §11].

We grade $U_0(\mathfrak{g}_m)$ by $X^*(H)$, the character lattice of the maximal torus $H \subseteq G$, in the following way. First define a $X^*(H)$ -grading on $U(\mathfrak{g}_m)$ by letting $\mathfrak{g}_{\alpha}t^i$ lie in grading α , \mathfrak{h}_m lie in grading 0, and extending this to all of $U(\mathfrak{g}_m)$. Then we observe that the kernel

of the quotient map $U(\mathfrak{g}_m) \to U_0(\mathfrak{g}_m)$ is generated by homogeneous elements, so this grading descends to an $X^*(H)$ grading on $U_0(\mathfrak{g}_m)$. We consider a category \mathcal{C} of certain $X^*(H)$ -graded $U_0(\mathfrak{g}_m)$ -modules whose gradings are compatible with the action of action of \mathfrak{h} in the following sense. Let $M = \bigoplus_{\gamma \in X^*(H)} M_{\gamma}$ be an $X^*(H)$ -graded $U_0(\mathfrak{g}_m)$ -module. Then the objects of the category \mathcal{C} are all $X^*(H)$ -graded $U_0(\mathfrak{g}_m)$ -modules such that \mathfrak{h} acts on M_{γ} by $d\gamma$ for all $\gamma \in X^*(H)$. The observations of [24, §11.4] actually show that this special case allows one to understand the category of all $X^*(H)$ graded $U_0(\mathfrak{g}_m)$ -modules.

For $M \in \mathcal{C}$, we write ch M, the character of M, for the function $X^*(H) \to \mathbb{Z}_{\geq 0}$ which sends $\gamma \in X^*(H)$ to dim M_{γ} (analogously to the definition given in the characteristic 0 case given in §3.4.1). We use the same notation for composition multiplicities in \mathcal{C} as in the ungraded case. We emphasise that graded and ungraded composition multiplicities are not the same, since a shift by $p\gamma$ in the grading of any module in \mathcal{C} yields a non-isomorphic graded module with isomorphic underlying ungraded module.

Let $\gamma \in X^*(H)$. Then we define the graded baby Verma module $\widehat{Z}(\gamma)$ to be the module $Z(d\gamma)$ with grading given by letting $\prod (f_{\alpha}t^i)^{m_{\alpha,i}} \otimes 1_{d\gamma}$ lie in grading $\gamma - \sum \alpha m_{\alpha,i}$. As noted above, if $\beta \in X^*(H)$, then $\widehat{Z}(\gamma)$ and $\widehat{Z}(\gamma + p\beta)$ have isomorphic module structures but different gradings. We also define $\widehat{Z}^{\mathfrak{g}}(\gamma)$ in a similar way.

Lemma 4.5.2. For all $\gamma \in X^*(H)$, $\widehat{Z}(\gamma) \in \mathcal{C}$ and $\widehat{Z}^{\mathfrak{g}}(\gamma) \in \mathcal{C}$.

Proof. The first claim follows immediately from the observation in $\widehat{Z}(\gamma)$ that $\mathfrak{h}_m^{(0)}$ acts on $\prod (f_{\alpha}t^i)^{m_{\alpha,i}} \otimes 1_{d\gamma}$ by $d\gamma - \sum (d\alpha)m_{\alpha,i}$, and a similar argument applies for $\widehat{Z}^{\mathfrak{g}}(\gamma)$.

By Lemma 4.4.3 and standard results on graded modules, we also see that $\widehat{Z}(\gamma)$ has a unique simple quotient $\widehat{L}(\gamma)$ which is isomorphic to $L(d\gamma)$ as a $U_0(\mathfrak{g}_m)$ -module.

Proposition 4.5.3. The $\widehat{L}(\gamma)$ form a complete set of isomorphism classes of simple objects in \mathcal{C} , and the $\operatorname{ch} \widehat{L}(\gamma)$ form a basis for the additive group of linear functions $X^*(H) \to \mathbb{Z}$. Hence the graded composition multiplicities of any module $M \in \mathcal{C}$ are determined by $\operatorname{ch} M$.

Proof. By [25, §1.5], the argument from which still applies here, all simple objects in \mathcal{C} are simple as $U_0(\mathfrak{g}_m)$ -modules and by Theorem 4.4.7 we have a classification of simple $U_0(\mathfrak{g}_m)$ -modules. Let $M \in \mathcal{C}$ be a simple object isomorphic as a $U_0(\mathfrak{g}_m)$ -module to $L(\lambda)$ for some $\lambda \in \Lambda_0$. For any grading on $L(\lambda)$, the element $1 \otimes 1_{\lambda}$ must be homogeneous and so must lie in grading γ for some $\gamma \in X^*(H)$ such that $d\gamma = \lambda$. Then by considering the $X^*(H)$ grading on $U_0(\mathfrak{g}_m)$ and module structure of $L(\lambda)$, we must have $M \cong \widehat{L}(\gamma)$. The second assertion follows from the observation that $\operatorname{ch}(\widehat{L}(\gamma))(\gamma) = 1$ and $\operatorname{ch}(\widehat{L}(\gamma))(\beta) = 0$ unless $\beta \leq \gamma$.

4.5.2 Composition multiplicities of restricted baby Verma modules

We now use the graded versions of the simple modules just introduced to compute the composition multiplicities of the baby Verma modules, which is a vital intermediate step towards our goal of computing the Cartan invariants for $U_0(\mathfrak{g}_m)$. The following formula allows us to express these in terms of the composition multiplicities for the baby Verma modules for the original reductive Lie algebra \mathfrak{g} .

Theorem 4.5.4. For any $\lambda, \mu \in \Lambda_0$, we have:

$$[Z(\lambda):L(\mu)] = \begin{cases} l_{\mu} p^{\frac{m}{2}(\dim(\mathfrak{g}) - \operatorname{rank}(\mathfrak{g})) - \operatorname{rank}(\mathfrak{g})} & \text{if } \lambda|_{\mathfrak{z}(\mathfrak{g})} = \mu|_{\mathfrak{z}(\mathfrak{g})} \\ 0 & \text{otherwise} \end{cases}$$

where $l_{\mu} = \sum_{\nu \in \Lambda_0} [Z^{\mathfrak{g}}(\nu) : L(\mu)]$. In particular, these composition multiplicatives depend only on $\lambda|_{\mathfrak{z}(\mathfrak{g})}$.

Proof. We first prove a relationship between the characters of the families of graded modules $\widehat{Z}(\gamma)$ and $\widehat{Z}^{\mathfrak{g}}(\gamma)$. This will then allow us to deduce a relationship between their composition multiplicities in the category \mathcal{C} of graded modules introduced in the previous section, from which the theorem follows by forgetting the gradings everywhere.

Start by defining a generalisation of Kostant's partition function $p_m: X^*(H) \to \mathbb{Z}_{\geq 0}$ by

$$p_m(\gamma) = |\{(m_{i,\alpha})_{\alpha \in \Phi^+, 1 \le i \le m} : \sum \alpha m_{i,\alpha} = \gamma, 0 \le m_{i,\alpha} \le p - 1\}|$$

and observe in particular that $p_m(\gamma) = 0$ if $(d\gamma)|_{\mathfrak{z}(\mathfrak{g})} \neq 0$. Let $I = \{((m_{i,\alpha})_{\alpha \in \Phi^+, 1 \leq i \leq m} : 0 \leq m_{i,\alpha} \leq p-1\}$ and fix $\gamma \in X^*(H)$. For each $\mathbf{m} \in I$, we can define an $X^*(H)$ -graded subspace $\widehat{Z}_{\mathbf{m}} = \{\prod f_{\alpha}^{n_{\alpha}} \prod (f_{\alpha}t^i)^{m_{\alpha,i}} \otimes 1_{\gamma} : 0 \leq n_{\alpha} \leq p-1\}$ of $\widehat{Z}(\gamma)$, and furthermore we have a decomposition $\widehat{Z}(\gamma) = \bigoplus_{\mathbf{m} \in I} \widehat{Z}_{\mathbf{m}}$. Now observe that $\operatorname{ch} \widehat{Z}_{\mathbf{m}} = \operatorname{ch} \widehat{Z}^{\mathfrak{g}}(\gamma - \sum \alpha m_{i,\alpha})$. Hence we obtain the formula $\operatorname{ch} \widehat{Z}(\gamma) = \sum_{\beta \in X^*(H)} p_m(\gamma - \beta) \operatorname{ch} \widehat{Z}^{\mathfrak{g}}(\beta)$, and can then immediately deduce that $[\widehat{Z}(\gamma) : \widehat{L}(\delta)] = \sum_{\beta \in X^*(H)} p_m(\gamma - \beta)[\widehat{Z}^{\mathfrak{g}}(\beta) : \widehat{L}(\delta)]$. Forgetting the gradings on these modules, we then see that:

$$[Z(d\gamma):L(d\delta)] = \sum_{\beta \in X^*(H)} p_m(\gamma - \beta)[Z^{\mathfrak{g}}(d\beta):L(d\delta)].$$

Set $\lambda = d\gamma$, $\mu = d\delta$, and $\nu = d\beta$. If $\lambda|_{\mathfrak{z}(\mathfrak{g})} \neq \mu|_{\mathfrak{z}(\mathfrak{g})}$ then for all $\beta \in X^*(H)$ either $p_m(\gamma - \beta) = 0$ or $[Z^{\mathfrak{g}}(\nu) : L(\mu)] = 0$, so $[Z(\lambda) : L(\mu)] = 0$. On the other hand, if $\lambda|_{\mathfrak{z}(\mathfrak{g})} = \mu|_{\mathfrak{z}(\mathfrak{g})}$, then using the fact that $d\beta = d(\beta + p\xi)$ for any $\xi \in X^*(H)$ we have $[Z(\lambda) : L(\mu)] = \sum_{\delta \in X^*(H)} p_m(\gamma - p\delta) \sum_{\nu \in \Lambda_0} [Z^{\mathfrak{g}}(\nu) : L(\mu)]$. Hence it remains only to show that for any $\gamma \in X^*(H)$, we have $\sum_{\delta \in X^*(H)} p_m(\gamma - p\delta) = p^{\frac{m}{2}(\dim(\mathfrak{g}) - \operatorname{rank}(\mathfrak{g})) - \operatorname{rank}(\mathfrak{g})}$.

Let $S_{\gamma} = \bigcup_{\delta \in X^*(H)} \{ (m_{i,\alpha})_{\alpha \in \Phi^+, 1 \leq i \leq m} : \sum \alpha m_{i,\alpha} = \gamma - p\delta, 0 \leq m_{i,\alpha} \leq p-1 \}$. Then observe that for any $\beta \in \Phi^+$ there is a bijection $f: S_{\gamma} \to S_{\gamma+\beta}$ given by

$$f(\mathbf{m})_{i,\alpha} = \begin{cases} m_{i,\alpha} & \text{if } i > 0 \text{ or } \alpha \neq \beta \\ m_{i,\alpha} + 1 \mod p & \text{if } i = 0 \text{ and } \alpha = \beta \end{cases}$$

Hence $|S_{\gamma}| = |S_{\delta}|$ for all $\gamma, \delta \in X^*(H)$. Now, observe that $\sum_{\gamma \in X^*(H)} p_m(\gamma) = p^{\frac{m}{2}(\dim(\mathfrak{g})-\operatorname{rank}(\mathfrak{g}))}$ and we have $p^{\operatorname{rank}(\mathfrak{g})}$ distinct sets S_{γ} , so for any $\gamma \in X^*(H)$, $|S_{\gamma}| = \sum_{\delta \in X^*(H)} p_m(\gamma - p\delta) = p^{\frac{m}{2}(\dim(\mathfrak{g})-\operatorname{rank}(\mathfrak{g}))-\operatorname{rank}(\mathfrak{g})}$, completing the proof.

Corollary 4.5.5. If $\mathfrak{z}(\mathfrak{g}) = 0$ then the composition multiplicaties $[Z(\lambda) : M(\lambda)]$ are inde-

pendent of λ , so all baby Verma modules have the same composition factors and multiplicities.

The fact the composition multiplicities depend only on $\lambda|_{\mathfrak{z}(\mathfrak{g})}$ is analogous to a result that states that for a the Lie algebra of a reductive group, restricted baby Verma modules with linked weights all have the same composition factors. The difference in this case is that the only condition for our weights to be linked is that they take the same values on $\mathfrak{z}(\mathfrak{g})$. This bears some similarity to Theorem 3.2.2 in the characteristic 0 case. For the following corollary, we recall the definition of a block from §2.3.3.

Corollary 4.5.6. Let $\lambda, \mu \in \Lambda_0$. If $\lambda|_{\mathfrak{z}(\mathfrak{g})} = \mu|_{\mathfrak{z}(\mathfrak{g})}$, then $L(\lambda)$ and $L(\mu)$ lie in the same block of $U_0(\mathfrak{g}_m)$. In particular, if $\mathfrak{z}(\mathfrak{g}) = 0$ then $U_0(\mathfrak{g}_m)$ has only one block.

Proof. Suppose $\lambda|_{\mathfrak{z}(\mathfrak{g})} = \mu|_{\mathfrak{z}(\mathfrak{g})}$. Then by Corollary 4.5.5 we have that $[Z(\lambda):L(\lambda)]$ and $[Z(\lambda):L(\mu)]$ are both non-zero, and by Lemma 4.4.3 we have that $Z(\lambda)$ is indecomposable. Hence $L(\lambda)$ and $L(\mu)$ lie in the same block.

Remark 4.5.7. Theorem 4.5.9 will imply that the condition $\lambda|_{\mathfrak{z}(\mathfrak{g})} = \mu|_{\mathfrak{z}(\mathfrak{g})}$ is also necessary for $L(\lambda)$ and $L(\mu)$ to lie in the same block. More generally, it is easy to see that the number of blocks of $U_0(\mathfrak{g}_m)$ is $p^{\dim \mathfrak{z}(\mathfrak{g}_m)} = p^{(m+1)\dim \mathfrak{z}(\mathfrak{g})}$.

4.5.3 Cartan invariants for $U_0(\mathfrak{g}_m)$

Recall the notation $Q^{\mathfrak{h}_m}(\lambda)$ from Proposition 4.2.5. We define the following families of $U_0(\mathfrak{g}_m)$ -modules:

$$Z_{proj}(\lambda) = U_0(\mathfrak{g}_m) \otimes_{U_0(\mathfrak{b}_m^+)} Q^{\mathfrak{h}_m}(\lambda)$$

$$DZ(\lambda) = (U_0(\mathfrak{g}_m) \otimes_{U_0(\mathfrak{b}_m^-)} (\mathbb{k}_{\lambda})^*)^*$$

where for $Z_{proj}(\lambda)$ we inflate $Q^{\mathfrak{h}_m}(\lambda)$ to a $U_0(\mathfrak{b}_m^+)$ -module by letting \mathfrak{n}_m^+ act by 0, and for $DZ(\lambda)$ we inflate $(\mathbb{k}_{\lambda})^*$ to a $U_0(\mathfrak{b}_m^-)$ -module by letting \mathfrak{n}_m^- act by 0. Here the module structure on the dual M^* of a $U_0(\mathfrak{l})$ -module M for some Lie algebra \mathfrak{l} is given by $(x \cdot f)(m) =$

 $-f(x \cdot m)$ for $x \in \mathfrak{l}$, $f \in M^*$, and $m \in M$. Finally, we write $Q(\lambda)$ for the projective cover of $L(\lambda)$.

These definitions allow us to state the following theorem, which is a special case of [39, Theorem 1.3.6].

Theorem 4.5.8. For all $\lambda, \mu \in \Lambda_0$, we have $(Q(\lambda) : Z_{\text{proj}}(\mu)) = [DZ(\mu) : L(\lambda)].$

This theorem is the key ingredient in the proof of the following result, which gives a formula for the Cartan invariants of $U_0(\mathfrak{g}_m)$.

Theorem 4.5.9.

$$[Q(\lambda): L(\mu)] = \begin{cases} l_{\lambda} l_{\mu} p^{m \operatorname{dim}(\mathfrak{g}) - \operatorname{rank}(\mathfrak{g})} & \text{if } \lambda|_{\mathfrak{z}(\mathfrak{g})} = \mu|_{\mathfrak{z}(\mathfrak{g})} \\ 0 & \text{otherwise} \end{cases}$$

where again $l_{\lambda} = \sum_{\nu \in \Lambda_0} [Z^{\mathfrak{g}}(\nu) : L(\lambda)].$

Proof. We aim to find formulae for $[DZ(\mu):L(\lambda)]$ and $(Z_{\text{proj}}(\mu):Z(\lambda))$; the result will then follow by Theorem 4.5.8 and several applications of Proposition 4.5.1. First, we claim that given $\gamma \in X^*(H)$ such that $\mu = d\gamma$ we can construct an $X^*(H)$ -grading on $DZ(\mu)$ such that the graded module $\widehat{DZ}(\gamma) \in \mathcal{C}$ has the same character as $\widehat{Z}(\gamma)$; it is then immediate that $[DZ(\mu):L(\lambda)]=[Z(\mu):L(\lambda)]$ for all $\lambda,\mu \in \Lambda_0$. Observe that $(\mathbb{k}_{\mu})^*\cong \mathbb{k}_{-\mu}$, so we can define an $X^*(H)$ -grading on $U_0(\mathfrak{g}_m)\otimes_{U_0(\mathfrak{b}_m^-)}(\mathbb{k}_{\mu})^*$ by letting $\prod (e_{\alpha}t^i)^{m_{\alpha,i}}\otimes 1_{-\mu}$ lie in grading $-\gamma + \sum \alpha m_{\alpha,i}$.

Now, for any finite-dimensional $M \in \mathcal{C}$ we can choose a basis $\{x_1, \ldots, x_k\}$ such that x_i lies in grading δ_i , and then define a grading on the dual module M^* by choosing a dual basis $\{f_1, \ldots, f_k\}$ and letting f_i lie in grading $-\delta_i$. The resulting graded module M^* also lies in \mathcal{C} , and we see that $\operatorname{ch} M^*(\delta) = \operatorname{ch} M(-\delta)$. Applying this to $M = U_0(\mathfrak{g}_m) \otimes_{U_0(\mathfrak{b}_m^-)} (\mathbb{k}_{\mu})^*$ equipped with the grading described earlier, we obtain a graded module $\widehat{DZ}(\gamma)$ such that $\operatorname{ch} \widehat{DZ}(\gamma) = \operatorname{ch} \widehat{Z}(\gamma)$ as required.

Next, we observe that $U_0(\mathfrak{g}_m)$ is free as a $U_0(\mathfrak{b}_m^+)$ -module and hence the functor

 $U_0(\mathfrak{g}_m) \otimes_{U_0(\mathfrak{b}_m^+)} (-)$ is exact. Hence by Proposition 4.2.5 we see that $(Z_{proj}(\lambda) : Z(\mu)) = \delta_{\lambda,\mu} p^{m \operatorname{rank}(\mathfrak{g})}$. We can then apply Proposition 4.5.1 to compute:

$$\begin{split} [Q(\lambda):L(\mu)] &= \sum_{\nu_1 \in \Lambda_0} (Q(\lambda):Z_{proj}(\nu_1))[Z_{proj}(\nu_1):L(\mu)] \\ &= \sum_{\nu_1,\nu_2 \in \Lambda_0} (Q(\lambda):Z_{proj}(\nu_1))(Z_{proj}(\nu_1):Z(\nu_2))[Z(\nu_2):L(\mu)] \\ &= \sum_{\nu_1,\nu_2 \in \Lambda_0} [Z(\nu_1):L(\lambda)](Z_{proj}(\nu_1):Z(\nu_2))[Z(\nu_2):L(\mu)] \\ &= \sum_{\nu \in \Lambda_0} [Z(\nu):L(\lambda)]p^{m\operatorname{rank}(\mathfrak{g})}[Z(\nu):L(\mu)] \\ &= \begin{cases} l_{\lambda}l_{\mu}p^{m\operatorname{dim}(\mathfrak{g})-\operatorname{rank}(\mathfrak{g})} \text{ if } \lambda|_{\mathfrak{z}(\mathfrak{g})} = \mu|_{\mathfrak{z}(\mathfrak{g})} \\ 0 \text{ otherwise} \end{cases} \end{split}$$

Remark 4.5.10. It is possible to the use results of [47] along with the theory of translation functors to deduce that for p > 2h - 1 these Cartan invariants can actually be calculated in terms of the machinery of p-Kazhdan–Lusztig polynomials.

LIST OF REFERENCES

- [1] Henning Haahr Andersen and Catharina Stroppel. Twisting functors on \mathcal{O} . Represent. Theory, 7:681–699, 2003.
- [2] T. Arakawa and A. Premet. Quantizing Mishchenko-Fomenko subalgebras for centralizers via affine W-algebras. Trans. Moscow Math. Soc., 78:217–234, 2017.
- [3] Tomoyuki Arakawa, Lewis Topley, and Juan J. Villarreal. The centre of the modular affine vertex algebra at the critical level, 2023. arXiv:2305.17765.
- [4] Sergey Arkhipov. Algebraic construction of contragradient quasi-Verma modules in positive characteristic. In *Representation theory of algebraic groups and quantum groups*, volume 40 of *Adv. Stud. Pure Math.*, pages 27–68. Math. Soc. Japan, Tokyo, 2004.
- [5] Alexandre Beĭlinson and Joseph Bernstein. Localisation de g-modules. C. R. Acad. Sci. Paris Sér. I Math., 292(1):15–18, 1981.
- [6] J.-L. Brylinski and M. Kashiwara. Kazhdan-Lusztig conjecture and holonomic systems. *Invent. Math.*, 64(3):387–410, 1981.
- [7] Matthew Chaffe. Category \mathcal{O} for Takiff Lie algebras. *Math. Z.*, 304(1):Paper No. 14, 35, 2023.
- [8] Matthew Chaffe and Lewis Topley. Category \mathcal{O} for truncated current lie algebras. Canadian Journal of Mathematics, page 1–27, 2023.
- [9] Matthew Chaffe and Lewis Topley. Modular Representations of Truncated current Lie algebras, 2023. arXiv:2311.08208.
- [10] Shun-Jen Cheng, Volodymyr Mazorchuk, and Weiqiang Wang. Equivalence of blocks for the general linear Lie superalgebra. *Lett. Math. Phys.*, 103(12):1313–1327, 2013.

- [11] Neil Chriss and Victor Ginzburg. Representation theory and complex geometry. Birkhäuser Boston, Inc., Boston, MA, 1997.
- [12] David H. Collingwood and William M. McGovern. Nilpotent orbits in semisimple Lie algebras. Van Nostrand Reinhold Mathematics Series. Van Nostrand Reinhold Co., New York, 1993.
- [13] Charles W. Curtis. Noncommutative extensions of Hilbert rings. *Proc. Amer. Math. Soc.*, 4:945–955, 1953.
- [14] Jacques Dixmier. *Enveloping algebras*, volume 11 of *Graduate Studies in Mathematics*. American Mathematical Society, Providence, RI, 1996. Revised reprint of the 1977 translation.
- [15] Edward Frenkel. Langlands correspondence for loop groups, volume 103 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 2007.
- [16] Eric M. Friedlander and Brian J. Parshall. Rational actions associated to the adjoint representation. Ann. Sci. École Norm. Sup. (4), 20(2):215–226, 1987.
- [17] Eric M. Friedlander and Brian J. Parshall. Modular representation theory of Lie algebras. *Amer. J. Math.*, 110(6):1055–1093, 1988.
- [18] Skip Garibaldi. Vanishing of trace forms in low characteristics. *Algebra Number Theory*, 3(5):543–566, 2009. With an appendix by Alexander Premet.
- [19] François Geoffriau. Homomorphisme de Harish-Chandra pour les algèbres de Takiff généralisées. *J. Algebra*, 171(2):444–456, 1995.
- [20] P. J. Hilton and U. Stammbach. A Course in Homological Algebra, volume 4 of Graduate Texts in Mathematics. Springer-Verlag, New York, second edition, 1997.
- [21] James E. Humphreys. Algebraic groups and modular Lie algebras, volume No. 71 of Memoirs of the American Mathematical Society. American Mathematical Society, Providence, RI, 1967.
- [22] James E. Humphreys. Introduction to Lie algebras and representation theory, volume Vol. 9 of Graduate Texts in Mathematics. Springer-Verlag, New York-Berlin, 1972.

- [23] James E. Humphreys. Representations of semisimple Lie algebras in the BGG category \mathcal{O} , volume 94 of Graduate Studies in Mathematics. American Mathematical Society, Providence, RI, 2008.
- [24] Jens Carsten Jantzen. Representations of Lie algebras in prime characteristic. In Representation theories and algebraic geometry (Montreal, PQ, 1997), volume 514 of NATO Adv. Sci. Inst. Ser. C: Math. Phys. Sci., pages 185–235. Kluwer Acad. Publ., Dordrecht, 1998. Notes by Iain Gordon.
- [25] Jens Carsten Jantzen. Modular representations of reductive Lie algebras. *J. Pure Appl. Algebra*, 152(1-3):133–185, 2000. Commutative algebra, homological algebra and representation theory (Catania/Genoa/Rome, 1998).
- [26] Jens Carsten Jantzen. Representations of algebraic groups, volume 107 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, second edition, 2003.
- [27] Jens Carsten Jantzen. Nilpotent orbits in representation theory. In *Lie theory*, volume 228 of *Progr. Math.*, pages 1–211. Birkhäuser Boston, Boston, MA, 2004.
- [28] Victor G. Kac. Review of the article "Irreducible representations of Lie algebras of Reductive Groups and the Kac-Weisfeiler conjecture" by A. Premet. MathSciNet, 1995.
- [29] Masoud Kamgarpour. Compatibility of the Feigin-Frenkel isomorphism and the Harish-Chandra isomorphism for jet algebras. *Trans. Amer. Math. Soc.*, 368(3):2019–2038, 2016.
- [30] David Kazhdan and George Lusztig. Representations of Coxeter groups and Hecke algebras. *Invent. Math.*, 53(2):165–184, 1979.
- [31] Tiago Macedo and Alistair Savage. Invariant polynomials on truncated multicurrent algebras. J. Pure Appl. Algebra, 223(1):349–368, 2019.
- [32] Gunter Malle and Donna Testerman. Linear algebraic groups and finite groups of Lie type, volume 133 of Cambridge Studies in Advanced Mathematics. Cambridge University Press, Cambridge, 2011.
- [33] Benjamin Martin, David Stewart, and Lewis Topley. A proof of the first Kac-Weisfeiler conjecture in large characteristics. *Represent. Theory*, 23:278–293, 2019. With an appendix by Akaki Tikaradze.

- [34] Volodymyr Mazorchuk and Christoffer Söderberg. Category \mathcal{O} for Takiff \mathfrak{sl}_2 . J. Math. Phys., 60(11):111702, 15, 2019.
- [35] J. C. McConnell and J. C. Robson. *Noncommutative Noetherian rings*. Pure and Applied Mathematics (New York). John Wiley & Sons, Ltd., Chichester, 1987. With the cooperation of L. W. Small, A Wiley-Interscience Publication.
- [36] George J. McNinch and Donna M. Testerman. Central subalgebras of the centralizer of a nilpotent element. Proc. Amer. Math. Soc., 144(6):2383–2397, 2016.
- [37] Alexander Molev. Sugawara operators for classical Lie algebras, volume 229 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2018.
- [38] Mircea Mustață. Jet schemes of locally complete intersection canonical singularities. *Invent. Math.*, 145(3):397–424, 2001. With an appendix by David Eisenbud and Edward Frenkel.
- [39] Daniel K. Nakano. Projective modules over Lie algebras of Cartan type. Mem. Amer. Math. Soc., 98(470):vi+84, 1992.
- [40] Alexander Premet. Irreducible representations of Lie algebras of reductive groups and the Kac-Weisfeiler conjecture. *Invent. Math.*, 121(1):79–117, 1995.
- [41] Alexander Premet. Complexity of Lie algebra representations and nilpotent elements of the stabilizers of linear forms. *Math. Z.*, 228(2):255–282, 1998.
- [42] Alexander Premet. Special transverse slices and their enveloping algebras. Adv. Math., 170(1):1–55, 2002. With an appendix by Serge Skryabin.
- [43] Alexander Premet and Serge Skryabin. Representations of restricted Lie algebras and families of associative l-algebras. *J. Reine Angew. Math.*, 507:189–218, 1999.
- [44] Alexander Premet and David I. Stewart. Rigid orbits and sheets in reductive Lie algebras over fields of prime characteristic. *J. Inst. Math. Jussieu*, 17(3):583–613, 2018.
- [45] Mustapha Raïs and Patrice Tauvel. Indice et polynômes invariants pour certaines algèbres de Lie. J. Reine Angew. Math., 425:123–140, 1992.

- [46] R. W. Richardson. Derivatives of invariant polynomials on a semisimple Lie algebra. In Miniconference on harmonic analysis and operator algebras (Canberra, 1987), volume 15 of Proc. Centre Math. Anal. Austral. Nat. Univ., pages 228–241. Austral. Nat. Univ., Canberra, 1987.
- [47] Simon Riche and Geordie Williamson. A simple character formula. *Ann. H. Lebesgue*, 4:503–535, 2021.
- [48] Guang Yu Shen. Graded modules of graded Lie algebras of Cartan type. II. Positive and negative graded modules. Sci. Sinica Ser. A, 29(10):1009–1019, 1986.
- [49] T. A. Springer. Linear Algebraic Groups. In *Perspectives in mathematics*, pages 455–495. Birkhäuser, Basel, 1984.
- [50] Helmut Strade and Rolf Farnsteiner. Modular Lie algebras and their representations, volume 116 of Monographs and Textbooks in Pure and Applied Mathematics. Marcel Dekker, Inc., New York, 1988.
- [51] S. J. Takiff. Rings of invariant polynomials for a class of Lie algebras. Trans. Amer. Math. Soc., 160:249–262, 1971.
- [52] Patrice Tauvel and Rupert W. T. Yu. Indice et formes linéaires stables dans les algèbres de Lie. J. Algebra, 273(2):507–516, 2004.
- [53] Lewis Topley. Invariants of centralisers in positive characteristic. *J. Algebra*, 399:1021–1050, 2014.
- [54] Lewis Topley. A non-restricted counterexample to the first Kac-Weisfeiler conjecture. *Proc. Amer. Math. Soc.*, 145(5):1937–1942, 2017.
- [55] B. Ju. Veĭsfeĭler and V. G. Kac. The irreducible representations of Lie *p*-algebras. Funkcional. Anal. i Priložen., 5(2):28–36, 1971.
- [56] Benjamin J. Wilson. Highest-weight theory for truncated current Lie algebras. *J. Algebra*, 336:1–27, 2011.
- [57] Oksana Yakimova. Surprising properties of centralisers in classical Lie algebras. Ann. Inst. Fourier (Grenoble), 59(3):903–935, 2009.

- [58] Xiaoyu Zhu. Simple modules over the Takiff Lie algebra for \mathfrak{sl}_2 , 2022. arXiv:2211.07261.
- [59] Xiaoyu Zhu. Induced modules and central character quotients for Takiff \mathfrak{sl}_2 , 2024. arXiv:2401.17627.